

PHD THESIS



Design for Perception

A human-centric approach to the design of driving automation systems based on the driver's perception

FJOLLË NOVAKAZI

DEPARTMENT OF INDUSTRIAL AND MATERIALS SCIENCE CHALMERS UNIVERSITY OF TECHNOLOGY Gothenburg, Sweden 2023 www.chalmers.se

Design for Perception

A human-centric approach to the design of driving automation systems based on the driver's perception

FJOLLË NOVAKAZI

Department of Industrial and Materials Science Division Design & Human Factors

CHALMERS UNIVERSITY OF TECHNOLOGY

Gothenburg, Sweden 2023

Design for Perception A human-centric approach to the design of driving automation systems based on the driver's perception

Fjollë Novakazi

ISBN: 978-91-7905-947-7

© Fjollë Novakazi, 2023.

Doktorsavhandlingar vid Chalmers tekniska högskola Ny serie nr 5413 ISSN 0346-718X

Published and distributed by Department of Industrial and Materials Science Division Design & Human Factors Chalmers University of Technology SE-412 96 Gothenburg, Sweden Telephone + 46 (0)31-772 1000

Cover by Johannes Nilsson.

Printed by Chalmers Reproservice Gothenburg, Sweden 2023

Abstract

The automotive industry is rapidly developing driving automation systems (DAS) with the aim of supporting drivers by means of the automation of longitudinal and lateral vehicle control. As vehicle complexity increases and update-over-the-air features are enabling continuous development of vehicle software and functionality, the driver's understanding of their responsibility and their vehicle's capabilities and limitations is becoming significantly more important. In order to motivate manufacturers to adopt a human-centric perspective for the development of driving automation systems, the factors influencing the driver's perception of these systems during their usage needs to be understood. Therefore, the aim of this thesis is to contribute towards the systematic development of DAS from a human-centric perspective.

The core research for this thesis is organised into four empirical studies, embedding a mixed methods research design. Study I aimed to investigate the usage of DAS in different driving contexts by facilitating an online survey to drivers in Germany, Spain, China, and the US. Study II aimed to explore the driver's contextual usage of DAS and which factors affect their understanding and employed an explanatory sequential mixed methods approach consisting of a Naturalistic Driving Study (NDS) in the greater Gothenburg area over a 7-month period. This was followed up by in-depth interviews to elicit knowledge about how drivers understand the DAS, and which factors influence their usage. Study III and Study IV aimed to gain further insights into which factors in the driver's perception of the DAS affect their understanding and consequent usage of DAS. Thus, Study III applied a Wizard-of-Oz on-road driving study, simulating a vehicle offering a Level 2 and a Level 4 DAS in the San Francisco Bay Area paired with pre- and post-driving in-depth interviews. Finally, Study IV applied a Wizard-of-Oz on-road driving study, simulating a vehicle offering a Level 2 and a Level 3 DAS, and contrasting two different human machine interfaces in Gothenburg, paired with post-driving in-depth interviews.

The results from these studies allowed a contribution to the body of research in a theoretical and practical form. The theoretical contribution is the unification of aspects that shape a driver's understanding of a DAS into a conceptual model. The unified model describes the process of how this understanding is shaped through the driver's perception of the DAS. The developed model further facilitated the development of a design toolkit by applying a participatory design approach (Study V) that facilitated co-creation sessions with domain experts (designers of DAS) in an industrial setting, which is considered a practical contribution to the field. The toolkit serves as a common foundation for aligning the motivations and goals of developers, designers, and strategists with regulators. Consequently, it can support practitioners to: 1. explore possible solutions driven by a systematic approach; 2. identify areas of improvement by applying the lens of the user; and 3. ideate and evaluate design decisions through a structured process. Thus, it facilitates the identification of design, evaluation, and training approaches that promote appropriate usage strategies for drivers and the building of a sufficient understanding of a DAS.

Keywords: Driving automation, levels of automation, perception, understanding, mental model, human-centric design, design tool, design method, empirical research, mixed-methods research, cognitive engineering, human factors.

Acknowledgments

Going into this, and throughout, I was reminded that getting a PhD would be challenging. While I won't dispute that statement, I will add that these were some of my favourite years. It was some of the best years of my life because I got to learn and grow alongside and from a diverse group of people I would never have met otherwise, and I am grateful for that. However, W.A. Ward remarked that *"feeling gratitude and not expressing it is like wrapping a present and not giving it"*, therefore, I want to express my thankfulness to everyone making this work possible and a difference during this journey.

First and foremost, my deepest gratitude goes to MariAnne Karlsson, my principal supervisor who supported me throughout the process and beyond. Not only, was I lucky to receive the best scientific education and guidance over the past years, but also more support and encouragement than I could have ever wished for. Thank You from the bottom of my heart. Many thanks also go to Lars-Ola Bligård, my co-supervisor. I always will remember our discussions and your jokes as one of the keys for having lots of fun, while approaching complex questions.

As an industrial PhD candidate, you get twice the manpower, and I was especially lucky, that I got to work with a number of great people at Volvo Cars over the years. First, I want to thank Pernilla Krook for your insight and friendship over the years, and Mikael Gordh who believed in me and gifted me his support. Further, I want to thank Christoffer Kopp, Annie Rydström and especially Alexander Eriksson who acted in the role of my industrial supervisors, confidents, and guides during different stages of this work. You all brought your unique perspective and knowledge to our discussions. Especially you, Alex, without you, your support and believe in my work, the last bit of this thesis would likely not have happened the way it did. Casper Wickman, I cannot begin to express how grateful I am to have had you by my side throughout this journey. You made it, unlike many others, from the beginning to the end, and you have been an inspiration, mentor, champion, and true friend. For that, I am indebted to you.

Special appreciation goes also to my collaborators and co-authors. Thank You, Julia Orlovksa, for inviting me to collaborate on Study II, which ended up giving us both a lot more to learn then we set out to do. The most fun and learning, however, I had with you Mikael Johansson. Thank You, for joining forces during Study III, always being up for another idea or discussion and letting me in on your perspective – I hope that our paths continue to cross. I am also grateful for Soyeon Kim, who with her creativity and abundance of energy brought about another spur of motivation during the last sprint of this thesis. I am so happy we got to collaborate on your endeavours and on Study IV and to have made a friend in you.

Not to forget, all the research conducted was enabled by my colleagues at Volvo Cars, who supported by providing their expertise and time in the preparation and conduction of the studies, and most importantly their interest in what I do. Thank you, Tor Guttormsen, Hanna Gerdeskans, Alvaro Alonso Bivou, Alfons Massey, Claas Wedin and especially Kaj Luukkonen[†] for making Study III possible and so much fun. Thanks also to Erick Hernandez, Patrik Castenbladh, Niclas Lind, Alice Nilsson and Pauline Lorin that supported Study IV in the same extent. And a very special thank you to Linnea Olsson – I really enjoyed working with you. A big thank you also to Fernando Sánchez-Ulloa, who made me an unofficial member in his team allowing me to understand the needs of my colleagues, and ultimately providing the space and time to conduct Study V. However, without your time and enthusiasm, Kevin Orende, Tatiana Choeuiri and Manu Revi, I would have never learned as much as I did. Yet my warmest appreciation goes to you, Manu, you did not only believe in the usefulness of the toolkit, and applied it, but also supported me during the year, incorporating all the feedback and improving it. Thank you, it was the most fun working with you. Finally, a special thank you to Amanda Jonell, who was not only my biggest cheerleader during the last study, but also supported with her skills as a researcher during the interviews. And last but not least, appreciation goes to Kaspar Raats, who jumped in when I needed a moderator and has been my brother in mind during these past few years, walking the same path and always ready for a good conversation and a laugh. However, none of my studies would have been possible without all the participants. Thank you, for dedicating your time, driving around with us, answering so many questions, and describing your experiences to me. You contributed to fun and insightful sessions that formed this work.

A lot of people involved, but this kind of work requires not only expertise and lots of work, but also idle time. One of the things I learned from MariAnne is that good ideas need incubation time. And the motivation and ideas that enabled me to go the extra mile came to me in the days between major milestones. For that, I first want to thank each of my colleagues at the division Design & Human Factors at Chalmers. Not only have you been a second home, providing so much encouragement and support, but also endless fikas, craft meets, rock parties and skitgubbe tournaments. You are truly amazing and fun people and I count myself lucky to have had the pleasure being part of it. I am especially grateful for the exchanges with you Antonio Cobaleda Cordero, the abundance in laughter and positivity you bring, Göran Smith, for good advice and better music, and Fredrick Ekman, for always making time for a meaningful conversation. My deepest gratitude, however, goes to the wonderful women I've had the pleasure of meeting here. Siw Eriksson, Cecilia Berlin, Eva Simonsen, and Anneli Selvefors - you are all amazing and have taught me more than you might know. And most significantly, Maral Babapour and Helena Strömberg, I truly admire you, appreciate your insightful advice, and friendship - you are an inspiration I am grateful to have you in my life.

And last, but not least, I want to thank my family. I am profoundly thankful for my parents who have been the greatest role models and supporters throughout my entire life. You proved to me the value of hard work and perseverance. Faleminderit për dashninë dhe mbështetjen. Besimi i patundshëm në mua është dhurata më e mire që do të kisha kërkuar. Finally, Johannes, I am grateful beyond words for you. You have been the greatest support anyone could ever ask for, cooking the best meals, making sure I relax and get a laugh in at least once a day, listening to me rambling about my theories and models, making my sketches into beautiful illustrations, joining me on travels, and entertaining unsolicited educational monologues about driving automation and human cognition with patience, all the while renovating our home with me. Without you, the last years would have been only half as fun. Thank You.

Appended Publications

Paper A

Karlsson, I.C.M., & Novakazi, F. (2023). Drivers' usage of driving automation systems in different contexts: A survey in China, Germany, Spain and USA. *IET Intelligent Transport Systems*, 17(10), 2004–2019. https://doi.org/10.1049/itr2.12392

Contribution: Karlsson and Novakazi designed the survey and collected the data. Analysis, writing, and editing of the paper were carried out in collaboration between the authors.

Paper B

Orlovska, J., Novakazi, F., Lars-Ola, B., Karlsson, I.C.M., Wickman, C., & Söderberg, R. (2020). Effects of the driving context on the usage of Automated Driver Assistance Systems (ADAS) - Naturalistic Driving Study for ADAS evaluation. *Transportation Research Interdisciplinary Perspectives*, 4, 100093. https://doi.org/10.1016/j.trip.2020.100093

Contribution: Novakazi planned and conducted the qualitative study phase. Analyses, writing, and editing of the paper were carried out in collaboration between the authors.

Paper C

Novakazi, F., Orlovska, J., Bligård, L. O., & Wickman, C. (2020). Stepping over the threshold linking understanding and usage of Automated Driver Assistance Systems (ADAS). *Transportation Research Interdisciplinary Perspectives*, 8, 100252. https://doi.org/10.1016/j.trip.2020.100252

Contribution: Novakazi planned and conducted the qualitative study phase, with feedback from Bligård. The analysis of the data was carried out in collaboration between the authors. Novakazi wrote the manuscript with feedback from the other authors.

Paper D

Johansson, M., & Novakazi, F. (2020). To Drive or Not to Drive – When Users Prefer to Use Automated Driving Functions. *Proceedings of the 7th Humanist Conference.* Rhodes Island, Greece, September 2020.

Contribution: Novakazi and Johansson planned, conducted, and analyzed the study in collaboration. Johannson wrote the paper with feedback from the other authors.

Paper E

Novakazi, F., Johansson, M., Strömberg, H., & Karlsson, I.C.M. (2021). Levels of What? Investigating Drivers' Understanding of Different Levels of Automation in Vehicles. *Journal of Cognitive Engineering and Decision Making*, 15(2–3), 116–132. https://doi.org/10.1177/15553434211009024

Contribution: Novakazi and Johansson planned, conducted, and analyzed the study in collaboration. Novakazi and Johansson wrote the paper with feedback from the other authors.

Paper F

Johansson, M., Novakazi, F., Strömberg, H. & Karlsson, I.C.M. (2023, accepted with revision). Piecing together the puzzle – exploring how different information sources influence users' understanding of automated vehicles. *Behaviour & Information Technology.*

Contribution: Novakazi and Johansson planned, conducted the study, and analyzed the data. Johansson and Karlsson wrote the paper with feedback from the other authors.

Paper G

Kim, S., Novakazi, F. and Karlsson, I.C.M. (2023, in submission). Interaction Challenges in Automated Vehicles with Multiple Levels of Driving Automation - An On-Road Study. *Applied Ergonomics.*

Contribution: Novakazi planned and conducted the study, with feedback from Karlsson. Analysis, writing and editing of the paper were carried out in collaboration between the authors.

Paper H

Novakazi, F. & Bligård, L.O. (2023, in submission). Design for Perception – Co-Creation and Validation of a Design-Tool for Driving Automation Systems. *Conference of Intelligent User Interfaces, 2024.*

Contribution: Novakazi planned, conducted the study, analyzed the data, wrote, and edited the manuscript with feedback from Bligård.

Additional Publications

Johansson, M., & Novakazi, F. (2023). Action-Meaning Networks - A Novel Methodology to Identify Unsafe Usage of Driving Automation. *Proceedings of the 8th Humanist Conference.* September 21-22 Berlin, 2023, Germany.

Kim, S., Shi, E., Novakazi, F., & Oviedo-Trespalacios, O. (2023 Stakeholder-Centred Taxonomy Design for Automated Vehicles. Adjunct Proceedings of the 15th International Conference on Automotive User Interfaces and Interactive Vehicular Applications. https://doi.org/10.1145/3581961.3609829

Lidander, L., Novakazi, F., & Erhardsson, G. (2022). Building Blocks of Responsibility - A Conceptual Model Illustrating the Factors Influencing Perceived Responsibility Over the Driving Task when Interacting with Driving Automation Systems. Adjunct Proceedings of the 14th International Conference on Automotive User Interfaces and Interactive Vehicular Applications. https://doi.org/10.1145/3544999.3552524

Novakazi, F., Eriksson, A., & Bligård, L. O. (2022). Design for Perception - A Systematic Approach for the Design of Driving Automation Systems based on the Users' Perception. Adjunct Proceedings of the 14th International Conference on Automotive User Interfaces and Interactive Vehicular Applications. https://doi.org/10.1145/3544999.3552525

Rydström, A., Mullaart, M. S., Novakazi, F., Johansson, M., & Eriksson, A. (2022). Drivers' Performance in Non-critical Take-Overs from an Automated Driving System—An On-Road Study. *Human Factors*. https://doi.org/10.1177/00187208211053460

Johansson, M., Söderholm, M., Novakazi, F., & Rydström, A. (2021). The decline of user experience in transition from automated driving to manual driving. *Information*, 12(3). https://doi.org/10.3390/info12030126

Novakazi, F., Johansson, M., Erhardsson, G., & Lidander, L. (2021). Who's in charge? The influence of perceived control on responsibility and mode awareness in driving automation. *IT - Information Technology*, 62(2). https://doi.org/10.1515/itit-2020-0020 Orlovska, J., Novakazi, F., Wickman, C., & Söderberg, R. (2019). Mixed-method design for user behavior evaluation of automated driver assistance systems: An automotive industry case. 22nd International Conference on Engineering Design, ICED 2019. Delft, Netherlands. https://doi.org/10.1017/dsi.2019.186

List of Abbreviations

ACC:	Adaptive Cruise Control
AV:	Automated Vehicle
BA:	Behavioural Adaptation
BASt:	Bundesanstalt für Straßenwesen
CC:	Conventional Cruise Control
CW:	Cognitive Walkthrough
ECW:	Enhanced Cognitive Walkthrough
DAS:	Driving Automation Systems
DC:	Drive Cycle
DDT:	Dynamic Driving Task
DFMEA:	Design Failure Modes and Effects Analysis
FMEA:	Failure Modes and Effects Analysis
GDPR:	General Data Protection Regulations
GPS:	Global Positioning System
HCI:	Human-Computer Interaction
HMI:	Human Machine Interface
HOR:	Hand-Over Request
JCS:	Joint Cognitive Systems
GUI:	Graphical User Interface
LoA:	Level of Automation
LCA:	Lance Centring Assist

LKA:	Lane Keeping Aid
NDS:	Naturalistic Driving Study
NHTSA:	National Highway Traffic Safety Administration
ODD:	Operational Design Domain
OEM:	Original Equipment Manufacturer
PA:	Pilot Assist (Level 2 DAS in Volvo vehicles)
PR:	Participatory Research
SAE:	Society of Automotive Engineers
SUV:	Sport Utility Vehicle
TOR:	Take-Over Request
UI:	User Interface
WOz:	Wizard-of-Oz

Contents

Ab	strac	xt	i
Ac	know	ledgements	iv
Ар	penc	led Publications	vii
Ad	ditio	nal Publications	ix
Ac	rony	ms	xii
Lis	t of	Figures x	viii
Lis	t of	Tables	xx
1	Intro	oduction	1
	1.1	Background	1
	1.2	Aim and Research Question	6
	1.3	Thesis Outline	7
2	Frar	ne of Reference	13
	2.1	Automation	13
		Perspectives and Challenges	14

		Levels of Automation
		Levels of Driving Automation and Human Factor Issues 19
	2.2	Understanding
		Perception
		Mental Models
	2.3	From Perception to Understanding: A Unified Model 29
3	Res	earch Approach 3
Ū	3.1	Personal Context 3
	3.2	Methodological Approach
	0.2	Organisation of Research 30
	33	Overview of Studies
	0.0	Ethical Considerations
	34	Analysis and Evaluation
	0.4	Cross-Study Analysis
		Development and Evaluation of the Design Toolkit
		Development and Evaluation of the Design Tookit 4
4	Sun	nmary of Studies 57
	4.1	Study I: Context of Use
		Aim
		Method
		Findings
	4.2	Study II: Context of Use and Understanding 65
		Aim
		Method
		Findings
	4.3	Study III: Perception and Understanding
		Aim
		Method
		Findings
	4.4	Study IV: Perception and Understanding
		Aim
		Method
		Findings
	4.5	Study V: Co-Creation of a Design-Tool
	-	Aim
		Method

		Findings	93
5	Syn	thesis	99
	5.1	Factors Impacting the Driver's Perception and Consequent Un-	
		derstanding	99
		The Driver's Understanding	102
		The Driver's Perception	107
	5.2	The Process of how Perception Shapes Understanding $\ . \ . \ .$	116
	5.3	Implications for the Design of Driving Automation	119
	5.4	The Design for Perception Toolkit	121
		The Toolkit's Variants	122
6	Disc	cussion	131
	6.1	Contributions	131
		Theoretical Contributions	132
		Practical Contributions	135
		Design for Perception	138
	6.2	Reflections on the Approach	141
		Study I – International Online Survey	141
		Study II – Naturalistic Driving Study with Subsequent In-	
		Depth Interviews	142
		Study III and Study IV – Woz Driving Study with Observations	
		and In-Depth Interviews	143
		Study V – Co-Creation	144
		Concluding Remarks on the Research Approach	145
7	Con	clusions	151
Re	ferer	ices	155

List of Figures

2.1	Taxonomy and Definitions for Terms Related to Driving Au- tomation Systems for On-Road Motor Vehicles. (Adapted after SAE International [7])	22
2.2	Conceptual model of how apperception and understanding in- teract	30
3.1	Organisation of Research.	40
4.1	Explanatory Sequential Mixed-Methods Design	62
4.2	The Wizard-of-Oz vehicle, with video cameras facing the UI, and (1) driving wizard, (2) HMI wizard, (3) test leader and (4) participant.	68
4.3	Visual and Auditory description of GUI sequence and states for each of the driving modes	70
4.4	Route for the observation study in the San Francisco Bay area.	72
4.5	The Wizard-of-Oz vehicle, with video cameras facing the UI, and (1) driving wizard, (2) HMI wizard, (3) test leader and (4) participant.	78
4.6	Visual and auditory description of GUI sequence and states for each of the DAS and comparison of concepts GUI A and GUI B.	81

4.7	Route for driving session highlighting stretches for Level 3 DAS
	availability and exposure time
4.8	Participatory design research process indicating activities and
	participants during the project
4.9	Illustration of the toolkit's application during different activi-
	ties during the design process
5.1	Descriptive model of the process of how the driver's perception
	shapes their understanding of DAS
5.2	QR Code to the Design for Perception repository 121
5.3	Table 01 Understanding. Section illustrating the structure of
	the checklist with example contents for a Level 2 DAS 123
5.4	Table 02 Perception. Section illustrating the structure of the
	checklist with example contents for a Level 2 DAS 124
5.5	Deck 01 Understanding. Selection of cards from the deck 125
5.6	Deck 02 Perception. Selection of cards from the deck 126
5.7	Deck 03 Impact. Selection of cards from the deck

List of Tables

2.1	Endsley and Kaber's [58] LoA taxonomy for human-computer performance
3.1	Overview of the Studies, describing data collection method, assessed level of automation, number of participants, and context. 42
4.1	Description of DAS capabilities and limitations and information participants received prior to the driving session in Study III 71
4.2	Description of DAS capabilities and limitations and information participants received prior to the driving session in Study IV 82
4.3	Overview of application areas for the different variants (Excel Sheet vs Card Deck) of the tool
5.1	Context. Description of aspects and sub-aspects 103
5.2	Vehicle. Description of aspects and sub-aspects 105
5.3	Driver. Description of aspects and sub-aspects 106
5.4	Perceptual Sets (Top-Down Processing Factors): Preconceptions.109
5.5	Perceptual Sets (Top-Down Processing Factors): Previous Ex- periences 110
56	Parcentual Sats (Top Down Processing Factors): Parceived Safety 111
5.0 F 77	D = (10 + 00 + 10 + 10 + 10 + 10 + 10 + 10
5.7	Perceptual Sets (10p-Down Processing Factors): Trust 112

Sensory Information (Bottom-Up Processing Factors): Vehicle
Behaviour
Sensory Information (Bottom-Up Processing Factors): Infor-
mation Sources

Chapter 1 Introduction



CHAPTER 1

Introduction

This chapter introduces the area of research by providing a short description of the background and problem area. This is followed by a presentation of the aim and the research questions based on this description. The chapter concludes with an outline of this thesis.

1.1 Background

The development of automated vehicles (AVs) has garnered significant attention in various sectors, including industry, academia, and the general public, because of its argued potential to revolutionise transportation. However, the widespread availability of fully automated vehicles capable of operating under all situations is unlikely to occur over the next few decades, despite the rapid evolution of technology. This has resulted in a transitional phase for drivers, where the vehicle and the driver share the responsibility and control over the driving task. During this transitional phase, there is a need for clear guidelines and regulations to ensure the safe integration of driving automation systems into vehicles. Driving automation systems (DAS) can be conceptualised as a compilation of active safety technologies designed to assist the driver. Therefore, DAS are widely recognised as effective measures for tackling issues related to traffic safety [1][2][3]. Nevertheless, while it is acknowledged that these systems are not capable of entirely eliminating accidents, research conducted by Cicchino [4][5][6] has demonstrated that they do play a significant role in reducing the occurrence of fatalities and severe injuries.

However, drivers find themselves in the situation where the capabilities and limitations of such systems vary greatly, as they become increasingly complex with the introduction of vehicles offering several levels of automation (LoA). According to the definition provided by SAE International [7] the Levels of Driving Automation range from Level 0 (No Driving Automation) to Level 5 (Full Driving Automation). Vehicles classified as Level 0 to Level 2 automation are fitted with driver assistance systems that possess varying capabilities and functions. For example, Level 1 vehicles are equipped with adaptive cruise control (ACC), and Level 2 vehicles have both lane-keeping and adaptive cruise control functionalities [8]. Vehicles equipped with Level 3 and Level 4 DAS can operate without any input or intervention from the driver in specified driving contexts. These are termed operational design domains (ODD), i.e., specific environment, scenarios, and situations in which the automated system is intended to function without human intervention. Level 5 DAS can fully automate the vehicle in all ODDs and does not require any driver input.

A number of studies have investigated different factors that impact drivers' safe utilisation of DAS. Many conclude that the utilisation of automated systems is heavily influenced by the driver's understanding of the capabilities and limitations of these systems [9][10][11][12]. However, it has been shown manifold that a significant number of drivers lack awareness or possess incomplete comprehension about the constraints associated with the automated systems in their personal vehicles [13][14][15][16][17]. Therefore, for these technologies to improve traffic safety as well as enhance the driving experience and comfort [18], DAS must be designed so that drivers accept them, understand their capabilities and limitations, and use them appropriately, without abusing them or becoming overly reliant on them.

However, the implementation of several levels of automation (i.e., driving

modes) in vehicles has brought about a level of intricacy that poses considerable difficulties for drivers in understanding the capabilities of a DAS and their own role when interacting with such systems [19]. The availability of several driving modes, which are offered depending on different ODDs, can result in a state of confusion about what the system's and what the driver's responsibility is at a given time. The state where a driver is not aware of what driving mode or automation level is currently active, is also known as mode confusion [20].

A simulator study by Feldhütter and colleagues [21] on the topic of mode confusion in the interaction with automated vehicles, found that mode confusion between two driving modes is especially an issue in transitions. Some work contends that the user interface (UI) is essential in addressing this issue. For example, Banks and colleagues [22] found during an on-road study with Level 2 vehicles that a poor design of the UI and physical controls led to a number of the observed mode confusion instances. Based on similar observations, Carsten and Martens [23], argue for the importance of the UI design when developing automated systems in order to increase the driver's awareness about the driving mode, especially in the interaction with a vehicle offering several levels of automation.

A wide range of studies has investigated how transitions effects in automated driving (cf. [24]). A literature study conducted by Kim and colleagues [25] reviewed the effects of the UI in transitions and found that there are many conflicting results as to if the UI can have a positive effect on the driver's interaction with DAS. These findings highlight that there are other factors that need to be investigated.

Therefore, it can be deduced that one of the elements crucial for the safe utilization of DAS, is the driver's mental model. It has been found that drivers' understanding of how automated systems work and how they can utilize such systems often does not match the mental model of the designers intended use. For example, Beggatio and Krems [26] conducted a longitudinal driving simulator study to examine the potential changes in a driver's mental model of their interaction with Level 1 DAS, specifically ACC, when it is consistently employed in the same situation. While the precise effects remain unknown, this investigation indicates that drivers may develop an inaccurate or incomplete mental model of the systems in use, even after prolonged use, particularly if they are not exposed to specific situations that require them to adjust their mental model. Additional research endeavours have provided evidence of the intrinsic constraints associated with trial-and-error learning [27][28], coming to similar conclusion that driver's understanding of a DAS will not evolve, if it is sufficiently developed to allow for a perceived safe usage [29]. Consequently, the experience of individuals with the systems is affected. This indicates that the driver's understanding of the systems impacts their interaction and therefore safe utilization of such systems [30][13][14][31][15]. Thus, in order for drivers to utilise the systems in a safe manner, it is essential that they have a thorough understanding of the various modes of operation and limitations associated with these systems. Failure to acquire this knowledge can result in an inability to formulate effective usage strategies and lead to hazardous situations [32][26][22][33].

There are various methods available for disseminating information to drivers on vehicle functionalities, with one such method being the utilisation of car manuals, which aim to offer insights into the underlying principles of vehicle functions. Yet, other investigations have shown that car manuals, which include information about the operation of the systems, are not successful in training drivers due to the provided information being too technical and abstract and therefore go underused [34][35][36]. As a result, drivers often encounter challenges when attempting to put the information into practical application [15][37]. Moreover, Abraham and colleagues [38] drew attention to the lack of research on dealerships' provision of DAS-related information and training to consumers. Based on the findings of their exploratory research conducted across multiple dealerships, it was determined that there is a significant lack of educational initiatives. Consequently, the study concluded that consumers may continue to have insufficient or incorrect knowledge, i.e., mismatch in mental model, regarding the automated systems installed in their personal vehicles [38].

While many streams of research have identified issues with existing taxonomies, some have proposed improvements, and attempts have been made to reduce driver confusion, including investigating how the names of DAS are associated with different levels of automation [38], rephrasing the descriptions of responsibilities (e.g., [39]), and framing the driver's responsibility in terms of "driving" and "riding" [40]. In light of this realisation, SAE has added a visual chart to the initial taxonomy to simplify and clarify the "Levels of Driving Automation" for consumers [7]. The addition describes the levels of automation through simpler language and graphical representation, illustrating what the driver must do and what the system does. However, despite the fact that the human role has been clarified, the taxonomy continues to illustrate which actor is responsible for which task and does not account for human perception of the system use and purpose. Critics continue to argue that misconceptions and lack of understanding can be traced to the technological focus during the development of these systems, based on current taxonomies, which take a task allocation approach and do not regard the driver's perception of the system's capabilities and their own responsibilities [39][41].

As DAS become increasingly complex the need for a human-centric perspective increases, relying on new and human factor driven approaches to address the increasing challenges. However, while there is a number of human factors methods applied in the automotive industry, the chosen methods are usually driven by a human error paradigm, which aims to identify and quantify human error in the interaction with a system through different Human Reliability Analysis (HRA) techniques (cf. [42]). This paradigm is driven by the perspective that humans are the root cause in the failure of a technical system, focusing on errors rather than the effects of all forms of human behaviour, and thus, not considering the complex interaction that takes place in the human-automation interaction. Therefore, currently, a notable gap exists in the automotive industry, where designers lack a human-centric approach to effectively address the development of driving automation from a humancentric perspective which clarifies cognitive processes and drivers' needs in the interaction with a DAS. Recognizing the vital role of design methods in guiding design and engineering practices, a human-centric perspective is instrumental in assisting designers to accurately identify driver needs and validate their design choices [43], especially for the development of complex driver-automation interaction.

The highlighted factors only represent a subset of the complexities involved in the complex interaction between the driver and the vehicle, and still has not fully answered the question on how this knowledge can be applied to the design of driving automation. To the best of the author's knowledge, no prior research has endeavoured to establish a unified description of or has provided an in-depth overview of the factors that influence the driver's perception and consequent understanding of driving automation systems. Therefore, the presented work aimed to investigate the various factors that influence drivers' perception and consequently shape their understanding of DAS, with the goal of informing the humancentric design of driving automation from a human-centric perspective.

1.2 Aim and Research Question

This thesis has examined the interaction between humans and automation in the context of driving, specifically focusing on end-users of Driving Automation Systems. The primary objective was to contribute towards the systematic development of DAS from a human-centric standpoint. Given the problem description and thesis aim, the following research questions are posed:

RQ1: What are the factors that impact the driver's perception and subsequent understanding of driving automation systems?

RQ2: How can this knowledge be applied to guide design decisions for practitioners involved in the development of driving automation systems?

In order to support a problem-solving approach and motivate manufacturers to adopt a human-centric perspective for the design of DAS, it is imperative to comprehend the various factors that influence the drivers while utilising such systems. These research questions guide the generation of knowledge that can inform future design decisions in the development of automated driving systems from a human-centric standpoint.

1.3 Thesis Outline

The present thesis is structured into seven distinct chapters, the contents of which are briefly described below.

Chapter 1 introduces the area of research by providing a short description of the background and problem area. Based on that, the aim and the research questions are presented. The chapter ends with a brief outline of the thesis structure.

Chapter 2 details the theories through which the empirical data extracted from the four empirical studies were viewed. First, the chapter provides the reader with an overview of relevant background and where the theories originated, and it then concludes by discussing the author's perspectives and observations regarding the subject matter. The purpose is to provide a condensed version of the corpus of knowledge to the reader.

Chapter 3 provides an overview of the methodological strategy employed in the research. The paper examines the author's philosophical standpoint regarding scientific theories and explores its implications for their methodological choices in addressing the research questions. The connection between the attached papers and the research questions is illustrated, and afterwards, the methodology for conducting the cross-study analysis is explicated.

Chapter 4 provides a comprehensive overview of the synthesised findings derived from the cross-study analysis. It presents the findings in the form of a descriptive model that explains the interplay between perception and understanding. Additionally, it provides an overview of the developed design toolkit and its variants.

Chapter 5 establishes the foundation for addressing the research questions. The responses are based on a summary of the investigations that were carried out and details a summary of the results from the cross-study analysis for each of the conducted studies independently.

Chapter 6 concludes the thesis by presenting a comparison between the contribution made through this work and the current body of research, followed
by a discussion of the implications for design. The chapter concludes with reflections on the research approach.

Chapter 7 summarizes the conclusions of this thesis.

Chapter 2 Frame of Reference



CHAPTER 2

Frame of Reference

At the core of this thesis lie the concepts of automation and understanding, and therefore this chapter in divided into these two sections. The purpose is to provide a condensed version of the theoretical foundations to the reader. Firstly, the chapter provides the reader with an overview of the relevant background and thus, provides the frame of reference within which the data extracted from the four empirical studies was examined. It concludes by discussing the author's perspectives and observations regarding the subject matter.

2.1 Automation

This chapter presents an overview of several theories related to automation and Levels of Automation, as well as the existing taxonomies. In addition, it examines the human factors challenges that arise when implementing a system that provides several levels of automation, particularly in the automotive domain, and how this pertains to the design of DAS.

Perspectives and Challenges

Humans continuously strive to make work easier. Automation can be defined as the process by which mechanical or electronic devices autonomously operate an apparatus, process, or system, thereby replacing human labour [44]. A household thermostat, GPS route planners, parking assistance, and automatic windscreen wipers in vehicles, as well as sophisticated industrial control room systems, are all examples of applications that fall under the umbrella of automation. Control complexity can range from straightforward on-off control to intricate multivariable algorithms. Automation is used for information gathering and analysis, decision-making, carrying out an action, system monitoring, and aiding humans in executing a range of tasks. This human-machine interaction and cooperation is described by various models. One influential attempt aiming to describe the human role in the interaction with automated systems is Fitts' list [45]. Fitts allocated different tasks or functions to either the human (detection, perception, judgment, induction, and improvisation) or the machine (speed, power, computation, replication, and simultaneous operations), based on which could perform them in the best way.

The outsourcing of demanding tasks to an automated system has been developed for functions that humans do not wish to perform or cannot perform as accurately or reliably as machines, resulting in a transition from manual to supervisory control by the operator [46]. Although Fitt's [45] goal was to adopt a compensatory strategy, seeking to determine the optimal utilisation of automation to improve human abilities, the primary driving force behind this approach is the desire for efficiency, safety, quality, and productivity. However, it is crucial to acknowledge the intricate nature and difficulties associated with human-automation interaction. To that regard, Bainbridge [47] pointed out the paradox of automation, arguing that it may expand rather than eliminate problems with human operators. This irony derives from the deterioration of manual control skills, cognitive abilities, and the inability to stay vigilant during monitoring tasks. Consequently, human operators are often seen as a significant source of unpredictability and unreliability in system performance. This perception led many system designers to advocate for minimising human involvement in the execution of tasks. Notably, the need for a human operator stems from the inability to fully automate certain tasks due to their technical complexity, leaving operators with systems and inherent

processes that are difficult to fully comprehend and thus prone to use errors [47][48].

Additionally, Dekker and Woods [49], posited that placing emphasis on the allocation of tasks may create the misconception that automation can be effectively implemented by merely transferring human responsibilities to automated systems. This view emphasizes that a pure task allocation approach (cf. Fitt's list [45]) fails to acknowledge that the implementation of automation does not merely replace human labour but rather introduces novel responsibilities for individuals tasked with overseeing the automated processes.

Thus, for complex and unexpected systems that heavily rely on the adaptive capacity of individuals to handle unforeseen variability, it is imperative to shift the focus from a task allocation approach towards an understanding of the collaboration between humans and machines. In an attempt to address these challenges, different research streams have endeavoured to describe the human role in automated systems. For example, Kaber and Endsley [50] pointed out that the goal of human-centred automation is to create systems that retain the human operator in control loops with meaningful and well-designed tasks. One key issue they have identified in human-automation interaction is a lack of situation awareness [51], i.e., operators struggle to attend to monitoring and supervisory tasks over prolonged periods of time, loosing oversight of the system state and task progress for example. To overcome this problem, Kaber and Endsley's [50] approach on automation, focused on adaptive automation to manage operator workload through dynamic control allocation between the human and machine in collaboration. Other research findings on the search of a human-centred approach support achieving situation awareness in the interaction with complex systems, by distributing the tasks between the human and automation in a way that a team effort is achieved, which enhances the overall performance and decision-making capabilities of the operators. This approach recognizes the importance of maintaining a balance between human control and automation, allowing operators to maintain a high level of situational awareness while also benefiting from the capabilities of automation. By effectively distributing tasks and responsibilities, operators can focus on higher-level cognitive processes, such as problem-solving and decision-making, while automation handles routine or repetitive tasks [52][53].

Hollnagel and Woods [54] proposed a another perspective, known as the Joint

Cognitive Systems (JCS) theory, for characterising human involvement with an automated system. The JCS theory is an approach in cognitive science and human-computer interaction that emphasizes the interaction and collaboration between humans and their tools, technologies, or environments. It focuses on understanding cognition not just as an individual process but as a distributed phenomenon between the human and the machine, in order to achieve a common objective. JCS does not aim to diminish the importance of the operator or eliminate the involvement of a human operator but examines how humans and technology collaborate and complement each other's abilities, leading to effective problem-solving and decision-making, and thus, entails the collaborative interaction between humans and machines, rather than solely focusing on its individual constituents. This perspective highlights the necessity to consider the driver's understanding of DAS, as it assumes that the human operator and the machine have shared knowledge about situations encountered. This leads to the consideration of the ongoing communication and collaboration between driver and vehicle and spotlights the perception process as a crucial component in understanding the collaborative interaction.

Levels of Automation

Discussions regarding a cooperative relationship between human and machine in the control of complex systems have been conducted under the overarching concept of Levels of Automation. The following subsections will discuss this concept and its various taxonomies.

The term Level of Automation (LoA) refers to the level of task planning and performance interaction maintained between a human operator and a complex system [55][56]. Over the years, a number of different taxonomies describing LoA have been developed (cf. [57][58][59]). However, all of these taxonomies have in common that they, similar to Fitt's List, solely determine decisionmaking and control authority, for a wide range of tasks, which enables an assessment of whether human or machine should be carrying out certain tasks in a system.

In an early work, Sheridan and Verplank [60] developed a hierarchy which included the allocation of decision making and action selection of various degrees. The objective of Sheridan and Verplank's LoAs was to define if the machine or the human should have the control, rather than deciding how they share the control. Endsley [57] developed a LoA to extend the description of human decision making in the context of expert systems. The hierarchy constituted five levels ranging from 1. Manual Control, 2. Decision Support, 3. Consensual Artificial Intelligence, 4 Monitored Artificial Intelligence, and 5. Full Automation. This list concentrates on the cognitive tasks expert operators have to conduct, based on the information the system provides. Building on foregoing efforts, Endsley and Kaber [58] developed a taxonomy for LoA encompassing wider array of aspects, including a range of cognitive and psychomotor tasks in different domains, e.g., air traffic control, aircraft

piloting, and advanced manufacturing.

Throughout all the analysed domains, four generic functions were identified as common: 1. Monitoring: Taking in all information relevant to perceiving system status; 2. Generating: Formulating options or task strategies for achieving goals; 3. Selecting: Deciding on particular options or strategies; and 4. Implementing: Carrying out the chosen options. Based on these generic functions ten levels of automation were formulated, by assigning each function to either Human, Computer (i.e., automated system), or both, as can be seen in **Table 2.1**. The taxonomy serves as a means to identify decision-making and control authority and is applicable to a variety of tasks, which enables the assessment of whether a human or a computer should be responsible for certain tasks within a system, based on their performance. Parasuraman and

Levels of Automation		Monitoring	Generating	Selecting	Implementing
1	Manual Control	Human	Human	Human	Human
2	Action Support	Human-Computer	Human	Human	Human-Computer
3	Batch Processing	Human-Computer	Human	Human	Computer
4	Shared Control	Human-Computer	Human-Computer	Human	Human-Computer
5	Decision Support	Human-Computer	Human-Computer	Human	Computer
6	Blended Decision Making	Human-Computer	Human-Computer	Human-Computer	Computer
7	Rigid System	Human-Computer	Human-Computer	Human	Computer
8	Automated Decision Making	Human-Computer	Human-Computer	Computer	Computer
9	Supervisory Control	Human-Computer	Computer	Computer	Computer
10	Full Automation	Computer	Computer	Computer	Computer

Table 2.1: Endsley and Kaber's [58] LoA taxonomy for human-computer performance.

colleagues [46] provided a similar approach to LoA. However they do not provide a taxonomy, but rather a framework that aims to offer insights into the potential automation of various systems across different automation levels. The framework considers whether 1. Information Acquisition, 2. Information Analysis, 3. Decision Selection and 4. Action implementation are functions to be automated. The information acquisition, decision selection and action implementation for a function, are identical to Endsley and Kabers' [58] monitoring, selection, and implementation features, but, this framework does not include providing information as a function. Instead, it views information analysis as a separate step that needs to be automated.

The concept of LoA are still heavily discussed, and it has been suggested that the different taxonomies overlook certain key issues, or that they do not satisfactorily project a human-centred view on automation. Therefore, several studies have suggested alternative ways of categorizing automated system functions. One example is Spath and colleagues [61] who suggested a more generalized view of types of automation, and simply differentiated between a 'semi-automated' and 'completely automated' system. A system is categorized as semi-automated when the machine needs some degree of support from the human in performing the tasks at hand. For example, the start, end and succession of the functions are controlled by the human. 'Complete automation' is achieved when the machine completely relieves the human of their task to achieve higher precision, efficiency, or safety.

In a similar vein, Onken and Schulte [62] put forth a conceptual framework that delineates two distinct categories of automation: operator-controlled automation, which is typically regarded as a form of supervisory control, and built-in automation, which is seen as an inherent component of the system. In the context of supervisory control, the operator activates, monitors, and deactivates the automated function, which bears resemblance to the concept of semi-automation. The authors claim that, by engaging with the automated function, users develop a mental representation of the interaction, which then influences their use of the system.

However, an integrated automation, when built into a system, operates without direct user control. This lack of control can lead to users being unaware of the presence of automated functions or the current status of the system. The presence of concealed operations in built-in automation runs the risk of elicit frustration among human operators, may lead to a lack of awareness about the inherent system states [63] and in certain instances, may even result in hazardous situations. This was exemplified in the case of the Boeing 737 MAX incident, wherein a malfunctioning sensor initiated a sequence of events that led to a state of perplexity and ultimately culminated in a tragic accident [64].

The airline industry, where most of this research has gathered its findings, has tried to address the aforementioned challenges since the 1970s and introduced additional training, educating pilots about the changes to their role when flying with an automation [65][66], as well as educating them about the technical and functional aspects of the automation system [67]. Through these efforts, the airline industry has managed to provide pilots with a more complete understanding of the systems and significantly reduced the number of such events [68].

The various taxonomies and different approaches proposed over the years often serve a specific domain, and sometimes one domain applies different taxonomies [69]. For instance, the taxonomies created by Parasuraman [46] and Endsley and Kaber [58] were based on the knowledge gathered in the aviation sector but were considered as a basis for the introduction of automation into the automotive sector. However, there is a disparity between the levels provided in each of the automation descriptions, showing that a unification of approaches is not an easy task and different stakeholders regard different aspects as suitable for their respective approaches and domains.

Within the context of the automotive domain, the introduction of automated technologies into vehicles appears to reintroduce similar challenges. However, one significant obstacle is the fact that the general population possesses a lesser understanding of driving automation and its many levels compared to proficient operators, such as pilots, and training possibilities for DAS are lacking [70]. Thus, the introduction of levels of driving automation presents a new set of challenges for designers, developers, and regulators of such systems.

Levels of Driving Automation and Human Factor Issues

Driving Automation Systems are designed to assist the driver and facilitate the use of numerous comfort and safety functions. These systems feature a variety of assistive functions including anti-locking systems, blind spot warning, collision aid, and emergency brakes [71][72][73]. The adoption of electronic brake and drive control technologies facilitated the introduction of partially automated driving. Partially automated driving features provide assistance to the driver by utilizing longitudinal and/or lateral control of the vehicle through the process of scanning and analysing the surrounding environment [74][75]. To capture their environment, DAS utilize radar and video technologies to enable longitudinal vehicle motion (i.e., direction of travel) and lateral vehicle motion (i.e., position of the vehicle in the lane). Cruise Control was the initial implementation of longitudinal control, which aimed to maintain a predetermined speed during long travel. The term "Conventional Cruise Control (CCC)" has been widely adopted to differentiate it from Adaptive Cruise Control (ACC) systems. ACC systems additionally incorporate forward-looking sensors that enable the system to detect and regulate the distance and closing speed of a lead vehicle [73][76]. Active lateral vehicle control is an enhancement that builds upon lane departure warning technologies, which are designed to identify events when a vehicle deviates from its designated lane. Thus, active lane positioning control systems have the capability to steer the vehicle back to the lane's centre, rather than simply alerting the driver, provided that distinct lane markings are visible. The present generations of lateral vehicle control systems are commonly referred to as Lane Keeping Assist (LKA), Lane Centring Assist (LCA), or similar distinctions, which may vary based on the specific manufacturer.

Throughout this work 'system' is defined as a collection of various components or clusters of functions that support longitudinal (direction of travel) and lateral (lane position) vehicle motion. In contrast, a 'mode' or 'driving mode' refers to the specific level of automation executed by the system.

Vehicle automation is a prominent example of a domain where LoA taxonomies are continuously discussed. Several organisations and consortiums, including the German Federal Highway Research Institute (BASt), the National Highway Traffic Safety Administration (NHTSA), and SAE International, have devised taxonomies describing levels of driving automation. The most prominent, SAE's taxonomy J3016 [7], was created with the intention of establishing a common understanding of the various levels of automation and providing a definition for the classification of driving automation in order to support regulations. See **Figure 2.1** for an overview of the taxonomy and definitions related to DAS according to SAE International.

The taxonomy describes variations in driver and driving automation task allocations. The classification ranges from level 0 (No Driving Automation) to level 5 (Full Driving Automation), with each level having distinct functional specifications for the vehicle and, consequently, a distinct task for the driver. For instance, in Level 0 solely offers alerts and temporary support through vehicle safety subsystems, such as antilock braking and stability control. Level 1 (Driver Assistance) performs the longitudinal or lateral driving task but requires the presence of a driver who supervises the system and handles the remaining dynamic driving tasks (DDT). Level 2 (Partial Driving Automation) the DAS performs the longitudinal and lateral driving tasks simultaneously, under supervision of the driver, who also performs the remaining DDT. Throughout all of these driving modes, the vehicle is regarded as an assistive system, with the driver retaining full control and responsibility for the driving task at all times.

In Level 3 (Conditional Driving Automation) and Level 4 (High Driving Automation), the driver is relieved of the driving task under certain driving conditions and for a certain amount of time but retains supervisory control. A significant differentiation between Level 3 and Level 4 or 5 DAS is in the requirement for the human driver in Level 3 vehicles to be fallback-ready to resume control of the vehicle whenever prompted by the system. In contrast, at Level 4, the human driver does not need to be alert or prepared to take over the DDT at any time. In Level 5 (Full Automation), the driver is relieved completely from the driving task and becomes merely a passenger.

The existing frameworks for describing different levels of automated driving have advantages, such as describing the task of the automation and/or human [77][19], but they have been criticised for being based on detailed technical and functional taxonomies (e.g., [39][40] or a narrow a function allocation approach (e.g., [78]. However, the more driving modes are introduced into the vehicle, the less likely is it for the driver to understand the intricate system states fully, as the increased complexity makes it difficult for drivers to understand the various driving modes the vehicle offers and, thus, it becomes more challenging to grasp how the vehicle will behave in different situations. This



Figure 2.1: Taxonomy and Definitions for Terms Related to Driving Automation Systems for On-Road Motor Vehicles. (Adapted after SAE International [7])

contributes to ambiguity surrounding automation modes in vehicles, reducing the driver's overall understanding of the vehicle's capabilities and their own responsibilities (cf. [20][79][69][19].

Hence, it is argued that control allocation becomes a significant issue as the driving task (e.g., steering, accelerating, braking, keeping distance to other road users, navigating, etc.) shifts from the driver to the vehicle, especially as the driving task becomes increasingly shared between the driver and the vehicle. The transfer of control authority [80] during interaction with DAS is recognised to have both direct and indirect impacts on the driver and the system's utilisation. Direct effects refer to specific system function specifications that are designed to improve performance on one or more control levels, such as improved lane keeping performance. Indirect effects refer to unintended consequences that are not intended or inferred by the designers' functional specifications.

It has been shown that the use of these technologies can lead to the driver making behavioural adaptations (BA) to the system, potentially reducing safety and comfort benefits the use of these technologies promises [76]. The phenomenon was first recognised in the 1990s and defined as 'those behaviours which may occur following the introduction of changes to the road-vehicleuser system and which were not intended by the initiators of the change' [81]. Manser and colleagues [82] categorised BA process into three distinct time periods: immediate, which refers to the initial encounter; short-term, which spans days or weeks; and long-term, which encompasses months or years. The majority of studies focus on the immediate or short-term impacts and examine the early deviations from the mental model of interacting with the DAS. As a result, the long-term effects of these systems remain mostly uncertain [83][84]. Another stream of research on behavioural adaptation has developed a considerable body of literature on the identification of 'adverse behavioural consequences' [85] or negative behavioural adaptations. For instance, different research endeavours have long pointed out that drivers tend to have over-trust, which promotes over reliance, and misuse [86][87][88][22] [33], the effects of automation on situation awareness and the connected 'out-of-the-loop'-problem [89][90][91][92][93], fatigue through the monotony of supervising the vehicle [94][84] or that drivers simply use the systems in ways and in ODDs that were not intended by designers [95], leading to unsafe usage practices.

Multiple studies indicate that the driver's understanding of the capabilities and limitations of driving DAS, their awareness of the available and currently active driving modes, and ability to maintain the appropriate level of engagement and intervention in crucial scenarios is a matter of worry [32][13][14]. These concerns are widely based on evidence about driver misconception of the relevant functions, or over-confidence in the systems' capabilities and limitations.

Some research streams attempted to tackle the problem of informing the driver about the intention of the automation through the application of user interface design guidelines for automated vehicles (e.g., [23]). Nonetheless, the irony of automation lies in the perception of increased capability of such systems, which leads to a range of issues, extending beyond the user interface. Other research streams, in search of alternative explanations for the observed human factor issues, have indicated that the existing design of DAS lacks adequate attention to the driver's mental model of interacting with a DAS (e.g., [26][15]. The introduction of several driving modes into one vehicle only exacerbates the existing problems (e.g., [21][19]. Driver's struggle to understand feedback received from the vehicle about the different mode changes or are confused about information received from the vehicle, and what they are supposed to do [22][96]. A survey conducted by Seppelt and colleagues [40] revealed that the taxonomy proposed by SAE International [7] led to confusion regarding the roles and responsibilities of drivers, for example, that drivers could distinguish between driver assistance and full automation but had difficulty distinguishing between the levels in between. Similarly, Abraham et al. [38] demonstrated that the design of DAS affects the driver's perception of the system and the expected levels of accountability. Furthermore, Homans and colleagues [97], concluded based on a survey, that drivers' understanding of automation levels does not align with the taxonomies. This ambiguity may result in misuse or even non-use of the system, thereby diminishing the potential benefits that automation support can provide [46].

Furthermore, Yang and colleagues [39] argue that the extant levels of driving automation taxonomies are technology-centred and presented from the perspective of vehicle technology development or policy development, which is problematic as this approach may impact the design of the driver-vehicle interaction and might not address the ways drivers comprehend such systems. Seppelt and colleagues [40] noted that classifications are written by engineers for engineers. Smith [98] further argues that the established taxonomies influence the way designers think about the system, resulting in design decisions driven by technological instead of driver-centric viewpoints. Consequently, a technology-centred taxonomy may result in design decisions that disregard crucial driver perspectives, such as how drivers perceive and comprehend the systems.

Overall, research indicates that there are numerous factors that impact the driver's interaction with a DAS. Although DAS have the potential to greatly improve safety and comfort, the existing challenges highlight the importance of designing these systems to assist drivers in developing an accurate understanding of what driving modes the vehicle offers, what the currently active driving mode is and what the driver's role in the interaction with and use of an automated system is. Thus, in order to create a positive driving experience and prevent the misuse of DAS due to overreliance and misaligned mental models, designers must carefully consider the implications of various variables, such as cognitive, behavioural, and contextual aspects, and how they influence the driver's perception of and consequent interaction with DAS.

2.2 Understanding

This section presents an overview of the principles governing human perception of information and the cognitive process involved in understanding driving automation. It concludes with an explanation of the author's perspectives and definitions of these key concepts.

Perception

Our comprehension of the world is based on the information we perceive through our senses from the environment. In other words, perception is the process by which objects, events and relationships become phenomenally here, now, and real [99]. The five senses that comprise perception are visual, auditory, gustatory, olfactory, and tactile. Beyond this, perception includes proprioception, which is the ability to determine one's body's position in space, and vestibular senses, i.e., balance and motion.

In contemporary research, there is a prevailing consensus that the existing theories pertaining to perception do not offer comprehensive explanations for the intricate processes involved. Consequently, it can be inferred that the field of study pertaining to perception is an ongoing endeavour. To date, there are opposing views concerning the concept.

One of those, is the 'bottom-up'-theory of perception, which was coined by James Gibson [100]. He stated that the sensation of stimuli is perception and that there is no need to process incoming information because it is analysed in a single direction; in other words, the unprocessed data from our senses enables a direct interpretation of the environment. Thus, bottom-up processing of information is frequently referred to as data-driven processing because it is assumed that perception is directly influenced by stimuli generated by the sensory system. The contrasting approach, is the 'top-down' theory of perception, described by Richard Gregory [101]. He described perception as a process that is based on integrating higher cognitive information, like previous experiences, and knowledge. Thus, top-down processing of information is a goal-driven process, supported by recognizing patterns and categorizing them in the context in which they are perceived.

This means that, the 'bottom-up' approach requires no learning, as a percept¹ is solely based on stimuli that are currently being experienced from the envi-

 $^{^1{\}rm the}$ mental representations formed as a result of perception

ronment, whereas the top-down approach recognizes that previous knowledge, experiences, and expectations are essential in order to recognize a percept.

Other work suggests that perception is determined by the experiences and previous knowledge we have. Jerome Bruner [102] stated that, "all perceptual experience is necessarily the end product of a categorization process". and that perception, therefore, is a process of categorization of stimuli from merely cues to meaning. Further, he believed that the values we hold, and our needs and goals determine how we perceive the world, down to the lowest levels of the visual system. From this, one can infer that perception is the result of a combined bottom-up and top-down process, and hence, that we perceive the world in categories and patterns, guided by our needs, expectations and beliefs. This mental predisposition or readiness to perceive sensory stimuli in a particular way based on previous experiences, expectations, beliefs, and context, is also called 'perceptual sets' [103]. Perceptual sets regulate choices between competing alternative activities and, therefore, influence the outcomes of the perception process. This means, at a fundamental level, that perceptual sets influence what in the available sensory data we perceive and what we ignore. This is also in line with expectation theory [104], which asserts that a person will act in a particular manner based on their personal frame of reference, motivations, and interests, resulting in that decisions are based on the individual's estimation of how closely the anticipated outcomes of a given behaviour correspond to the desired outcomes. Similarly, an individual's interpretation of an event or adjacent stimuli is influenced by subconscious blinders such as lack of awareness and similarity of stimuli. Consequently, ambiguous information is interpreted based on a person's perceptual sets, which influence how they process presented information [104].

Nevertheless, in the literature there is ongoing controversy regarding this matter. However, according to a widely accepted definition, perception is defined as "the process or result of becoming aware of objects, relationships, and events through the senses, which includes activities such as recognising, observing, and discriminating" [105]. In other words, perception is the process of receiving sensory information, and then organising it into patterns, and recognizing or interpreting its meaning, in order to plan and execute an action. According to this interpretation, one can infer that top-down and

bottom-up processing occur simultaneously during the perception process and that attention is guided by either of them at any given moment, depending on the strength of the stimulus or the individual's objectives. For instance, when interacting with a machine interface, changes in the focus of attention will be influenced by the individual's knowledge or expectations regarding the location of the desired information (top-down process) and/or the inherent characteristics of the stimuli (bottom-up process). In the case of DAS, visual signals regarding the status of the DAS are sought in the instrument cluster, while an audible cue associated with the system status can direct or draw attention when necessary.

Thus, throughout this work, the operational definition of perception pertains to the cognitive process by which individuals evaluate and derive meaning from sensory stimuli in their surrounding environment. This process encompasses the identification of sensory stimuli, which will, throughout this work, be referred to as *sensory information*, as well as the readiness to receive stimuli and interpret information, relying on previous experiences or expectations, which will be referred to as *perceptual sets*.

Mental Models

The role of prior knowledge and cognitive processes (top-down perception) in the interpretation of sensory information (bottom-up perception) is also referred to as 'apperception' [106]. The term, more specifically, refers to the process by which the content of information obtained from sensory experiences is synthesised with the knowledge obtained from previous experiences and brought into consciousness. Thus, apperception is the process that assigns meanings to experiences and phenomena and integrates them into concepts and categories (i.e., mental representations) that assist us in navigating and understanding the world, e.g., understanding how to interact with a machine or product.

However, the concept of understanding is complex and without a unified definition. Gijsbers [107] defines understanding as the knowledge of which connections exist between various phenomena. Further, he argues that understanding enables us to successfully classify events, and thus comprehend and apply that knowledge.

Within the area of system design, this knowledge is commonly referred to as a 'mental model' (cf. [108]). In this context, the mental model of a system represents the user's understanding of the system's contents, operations, and underlying logic [109]. Thus, the mental model contains sufficient knowledge of the system's purpose, explanations of how it functions and the system states, as well as predictions on future states [110]. Mental models, although subject to development and evolution through experience, primarily consist of static knowledge related to the product or system. This includes its prominent features, operational mechanisms, interrelationships among different components, and the anticipated behaviour of its components when exposed to various external conditions [51].

Within the research community, mental models have been recognised as a significant element in ensuring the safe utilisation of DAS. Many investigations address different aspects of the topic. For instance, how do drivers construct a mental representation of a DAS operation? What is an adequate mental representation that facilitates safe utilisation? In what ways might Human-Machine Interface design contribute to the formation and development of a mental model? How do faulty mental models emerge, and how can they be avoided?

While ongoing attempts are being made to address these enquiries and others of a similar nature, it is well recognised that addressing the mental model during the design of a DAS can aid drivers in their understanding the vehicle's capabilities and limitations [111][14]; information and feedback received from the vehicle is relevant for drivers to be able to develop a mental model [112][113][16]; and that the ability to anticipate potential hazards and make effective decisions and responses on the road is impacted by the ability to forma a sufficient mental model [26][114][115]. Other work has investigated how exposure and use of a system impacts the driver's mental model of it [27][28], or how training can aid the development of a sufficient mental model for better utilization of a DAS [116].

The existing research recognises that mental models are constructed based on diverse ideas and perceptions originating from several sources. However, there is a dearth of knowledge regarding the specific components that comprise a driver's mental model of a DAS, as well as the sources that shape these components.

By investigating these various components, this thesis aims to shed light on the concepts of perception and understanding in the context of driving automation.

2.3 From Perception to Understanding: A Unified Model

While there are opposing views on the concept of perception, it is the author's belief that a unified view of the perception process is the answer to understanding the user's perception of a system. Therefore, the author refers to the concept of perception as an integrated top-down and bottom-up process, which is the synthesis of sensory information and perceptual sets leading to the development of a 'mental model' of a product or a system. This mental model is referred to throughout this thesis as 'mental representation' or 'understanding'. Therefore, the process of understanding incorporates the process of apperception – a combined top-down and bottom-up process of perception – as a crucial component in answering the question of how users understand their interaction with a product or system.

Figure 2.2 exemplifies the author's position regarding how the mental representation is formed through apperception, i.e., perceptual sets, and sensory input, and how this affects the user's overall understanding of a product (or system) in a conceptual illustration. The depicted feedback loops emphasize the distributed and adaptive joint cognition at play in the interaction between driver and vehicle.

During the use of a DAS the driver will recall the knowledge that they have about the system, i.e., a mirror of the mental representation of the system that helps them navigate their interaction with it. During use, the driver's apperception will synthesise retrieved stimuli into information that informs their next action, based on the knowledge (mental representation) they have of the system. This means that, during use, the driver will – subconsciously continuously assess the sensory information retrieved and weigh it against the perceptual sets, which will influence how they will perceive the feedback from and interaction with the vehicle, and consequently how they will interact with it. While that mental representation is largely static, new information that



Figure 2.2: Conceptual model of how apperception and understanding interact.

is encountered, for example during an unfamiliar situation, will be fed back through a continuous loop and can alter the driver's understanding of the system, or simply reinforce already known information, and thus the current mental model. This will also cause the reassessment of the perceptual sets and may cause the driver to update these with regard to the newly encountered knowledge.

All in all, this conceptual model represents how the process of apperception and the building of a mental representation of a product occur – the process of understanding. This conceptual model is the foundation of the work presented here, which seeks to shed light on how drivers perceive and consequently understand DAS.

Chapter 3 Research Approach



CHAPTER 3

Research Approach

This chapter outlines the methodological approach to the research. The author's personal context and philosophical perspective is discussed, as well as how this affects their methodological decisions when approaching the research questions. The relationship between the attached papers and the research questions is demonstrated, and finally the procedure for the cross-study analysis and the development of a design tool are subsequently explained.

3.1 Personal Context

The research was conducted during the author's tenure in the automobile industry, balancing academic and industrial commitments. The positioning of this thesis within an industrial setting has significant implications not only for its foundational epistemology but also for the choice of design examples. Rooted in the author's extensive industry experience, the work is grounded in a pragmatic approach. This approach involved selecting methods that were considered to be most effective for addressing the research questions. Further, a pragmatic approach prioritises real-world implications in the design process, emphasising the practical applicability of knowledge. The author's background has fostered a spirit of curiosity, driving the exploration of innovative techniques that extend beyond the conventional design tools commonly employed in daily industrial practice.

The aim of this research was to generate knowledge that enables automotive developers to make more informed design decisions when creating DAS. Laozi formulated the dictum: "Those who have knowledge, don't predict. Those who predict, don't have knowledge", that describes the author's aspiration to approach their research and development practices in a methodological and structured way. This distinctive perspective enriches the research, infusing it with practical relevance and a forward-looking ethos.

With a background in Information Science and Information Technology, coupled with a specialised focus on Human-Machine Interaction, the author inherently recognises the profound interconnections between human users and machines as pivotal elements in technical system design. The author's educational journey has nurtured a perspective where knowledge is gleaned through the scrutiny of artefacts and their practical application in real-world contexts, underpinned by a Human Factors Engineering approach. Moreover, the author brings substantial expertise as a practitioner in the realm of human factors engineering, specifically in designing and evaluating automobile user interfaces.

The primary objective of the work was to investigate and comprehend the driver's understanding when engaging with a vehicle offering several levels of automation. Therefore, a research approach that is both applicable and focused on people should be at the centre of the development process. Further, the author's practical experience highlighted the significance of approaching the development of DAS from a human-centric perspective, as opposed to a purely technological or safety-driven one.

Thus, the author's reflections and interpretations have been shaped by the acknowledgement that the human and the machine are two equal partners involved in a collaborative accomplishment of a shared task, and thus strive to highlight the human operator as a central factor for the design of technical systems. This viewpoint, shaped by Joint Cognitive Systems Theory [54], delves into the intricacies of human information processes, exploring how the

user's understanding and perception are influenced or enhanced by technical solutions. Ultimately, this approach merges the comprehension of technological systems with human perception, enabling the identification of cognitive states, processes, and the intricate interplay between humans and their environment. This perspective is the main contributor to the answering of the research questions that have guided the development of this body of work.

3.2 Methodological Approach

This work is consistent with a pragmatic worldview, and it makes use of methods that are both practical and outcome oriented. These approaches are founded on action, and they lead iteratively to additional action, which supported the answering of the research questions at hand. To put it another way, the pragmatism that was used took an openly value-oriented approach to the research that was conducted (cf. [117]). The pragmatist viewpoint acknowledges the existence of meaning not just within a person but also within an object. This is to say that our reality is continuously being renegotiated, disputed, and renegotiated in the light of its usefulness, with reference to the breadth of the issue. Because of this, information is seen to be both generalizable and contextually specific, and the nature of newly acquired information is thought to be best understood in terms of how it may be applied in real-world situations [118].

Thus, a combination of qualitative and quantitative research methods have been applied because pragmatism holds the belief that the approach to learning that is most effective is the one that results in the solution to the problem. Creswell and Clark [119] define mixed methods research as a paradigm in which the investigators collect, analyse, and integrate qualitative and quantitative approaches within a study or programme of inquiry, with the objective of gaining both a depth and breadth of understanding of the topic under investigation.

Therefore, a sequential, mixed methods [120] research design has been adopted and integrated throughout the course of this work. The sequential and embedded use of qualitative and quantitative methods aims to facilitate an integrated interpretation of the collected data by applying a hybrid analysis approach (inductive and deductive). The primary characteristic of sequential mixed methods design is the sequential execution of two phases, wherein the outcomes of the qualitative phase guide the structure of the quantitative phase (or vice versa). This iterative method enabled the author to leverage the insights acquired and triangulate the data throughout the design of the research approach to generate knowledge and develop theories that provide answers to the research questions.

An inductive approach, when combined with quantitative and qualitative re-

search approaches, paves the way for the generalisation of specific phenomena into plausible theories, the grounds for which led to the comprehension of the researched scope and provided support for verification through the triangulation of data. Subsequently, the present research has been based on the utilisation of inductive and exploratory methodologies in combination during the initial stages, specifically employing online surveys and naturalistic driving studies to gather empirical observations. These observations have been instrumental in identifying recurring patterns and developing a comprehensive understanding of drivers' behaviours and potential challenges for drivers when utilising DAS. Further, the author's objective has consistently been to incorporate in-depth interviews as a means of augmenting the scope and depth of the findings, thereby achieving a deeper understanding of observed patterns and phenomena. From there, the knowledge corpus was systematically expanded through the utilisation of deductive analysis approaches, enabled by on-road driving studies, in conjunction with qualitative data collection techniques, e.g., observations paired with think-aloud methods and in-depth interviews.

These combined approaches facilitated the contextualization of the research findings and the identification of underlying factors and rationales behind specific observations. This led to enhanced interpretation of the collected data and enhanced validity of the identified patterns and observed phenomena.

Organisation of Research

This work relies primarily on the author's applied and empirical research which aimed to identify factors influencing the driver's perception and their consequent understanding of DAS by utilizing a series of five studies. The following **Figure 3.1** illustrates the process of the different studies, their goals, their associated publications, and the synthesised output representing the theoretical and practical contributions made by this work.

Studies I, II and III built the foundation for the theoretical contribution presented in this work, which was a tentative model describing the different aspects constituting the driver's perception and consequent understanding of DAS. The associated papers all addressed different aspects of the underlying goal, which was to shed a light on how driver's utilise DAS, and from there to gain deeper knowledge about how they perceive and consequently understand such systems.



Figure 3.1: Organisation of Research.

The knowledge acquired in Studies I-III was built on to address RQ2, "In what ways may this knowledge be utilised as a means of providing design guidance for practitioners?". The aim was to propose a tool intended to support a more user or driver-centred design of DAS. Subsequently, a co-creation study (Study V) was undertaken to go deeper into the second research question. Study V employed a use case within the co-creation study to develop a user interface. This interface was then used in Study IV to acquire a more thorough understanding of driver's perception and understanding of DAS.

Finally, all four empirical studies (Studies I-IV) were analysed in a crossstudy analysis by utilizing a framework analysis approach to finalise the tentative model, which consequently addressed RQ 1, "Which factors influence the driver's perception and consequent understanding of DAS?". The synthesis of the results from these four empirical studies constitutes the theoretical contribution – a conceptual model describing how perception shapes understanding (presented in 5.1). The development of the design tool and the outcome of Study V represent the practical contribution provided by this work – a design toolkit called the Design for Perception toolkit for designers and developers of DAS (presented in 5.3).

3.3 Overview of Studies

To ensure consistency across datasets, each of the selected studies investigated driver understanding of DAS. The section that follows provides a summary of the four study designs, including a description of the data collection method(s), number of participants, context and assessed level of automation (according to SAE [7]). An overview of the conducted studies is presented in **Table 3.1**.

	Method of Data Collection	Level of Automation	Number of Participants	Context	Paper
Study I	Online Survey	Level 1 Level 2	Completed Questionnaires: 568 Germany; 532 Spain; 516 USA; 504 China	China, Germany, Spain, USA, 2018	A
Study II	Naturalistic driving study, 7-month data collection; consecutive semi- structured interviews	Level 1 Level 2	NDS: 132 vehicles, 7 months of data collection Interviews: 12 participants from identified usage groups	Gothenburg, Sweden, 2018- 2019	В, С
Study III	Wizard-of-Oz, on- road driving study; observations, and semi-structured interviews	Level 2 Level 4	20 participants, novice users	San Francisco, USA, 2019	D, E, F
Study IV	Wizard-of-Oz, on- road driving study; A/B GUI test; observations, and semi-structured interviews	Level 2 Level 3	16 participants, novice users	Gothenburg, Sweden, 2022	G
Study V	Co-Creation; Workshops and Feedback Sessions, and semi-structured interviews	n/a	6 AD/ADAS Designers and Developers; Author as facilitator, 1 User Researcher supporting facilitation	Gothenburg, Sweden, 2022	Н

 Table 3.1: Overview of the Studies, describing data collection method, assessed level of automation, number of participants, and context.

Study I (Paper A): Context of Use. The aim of this study was to investigate the application of Level 1 and Level 2 DAS, in a variety of driving contexts. In the course of the study, an online questionnaire was administered to individuals in the United States of America, China, Germany, and Spain. The survey aimed to compare usage Level 1 and Level 2 systems in personal

vehicles, in nine distinct driving contexts. The objective of conducting this study through an online questionnaire was to achieve a wide-reaching sample of drivers from various nations who possess access to the system in question, and thus, allowed for the collection of self-reported data regarding real-world user experiences with the systems in various driving contexts, enhancing the validity of the collected data.

Study II (Papers B and C): Context of Use and Understanding. The aim of this study was to acquire a more in-depth understanding of the drivers' reasons for using Level 1 and Level 2 DAS, as well as the drivers' understanding of the capabilities and limitations of these systems. Building upon Study I the contextual effects on the utilisation of the systems were also investigated.

This study was broken up into two distinct phases, a quantitative data collection and a qualitative data collection phase. A Naturalistic Driving $(ND)^1$ study was employed to acquire the quantitative data. In this analysis, the ND study provided a time-efficient and trustworthy method for conducting quantitative evaluations of system and driver performance, in conjunction with sensor-based data providing information such as weather conditions, road conditions, and data indicating traffic conditions on the roads. In order to shed a light on the sensor-based findings, in-depth interviews were conducted with a subset of the study participants in order to identify and explain the effect that driver's perception has on the use of DAS.

Study III (Papers D, E and F): Perception and Understanding. The aim of this study was to gather insights into first-time users' perception and understanding of a vehicle that offers several levels of automation, specifically Level 2 and Level 4. This aim was accomplished through conducting a semi-controlled on-road study in the San Francisco Bay Area, USA, where the participants experience two different DAS, in a Wizard-of- Oz^2 vehicle. The

¹ND study typically refers to a study in which data collection is not constrained by a strict experimental design [121]) This enables data collection in a natural driving context and under a variety of driving conditions, in real-world driving situations, and over prolonged periods of time.

²Wizard of Oz - The WOz technique, as described by [122]), entails the use of a human who assumes the role of a machine, all while keeping the participant unaware of this fact. In the application domain of driving automation, a WOz vehicle enables the simulation of different automation levels and conceptual user interfaces, thus enabling testing in
study facilitated the elicitation of insights by means of observation during the driving sessions, collection of think-aloud data, as well as in-depth semistructured interviews before and after the driving sessions.

Study IV (Paper G): Perception and Understanding. The aim of this study was to analyse the dynamics of human-machine interaction in a vehicle that incorporated several levels of automation, specifically Level 2 and Level 3. This was accomplished by making use of a semi-controlled Wizard-of-Oz (WOz) driving study that was conducted on-road, in Gothenburg, Sweden. During the driving sessions, the participants experienced two alternative graphical user interfaces (GUIs) designed to enhance drivers' mode awareness in a vehicle offering several levels of automation. Data was collected through recordings of participants' behaviour and think aloud comments. Additionally, in-depth semi-structured interviews were conducted after the driving sessions to further enhance the insights gathered. The results were then contrasted and compared, with the end goal of gaining insights into the factors that influence drivers' perception of a vehicle offering several levels of automation.

STUDY V (Paper H): Co-Creation of a Design Tool. The goal of this study was to utilize an initial version of the design tool and continue developing it further throughout a participatory design study, and by means of co-creation³ and other collaborative feedback methods. This study was structured around a series of workshops which enabled the domain experts, i.e., designers and developers who were also the target users, to use the tool within a defined use case. Thus, the workshops provided a setting for collaboration in which the participants could get to know the tool and share their experiences with it and requirements on it. Subsequently, the participants were interviewed, using a semi-structured approach to gain insights into the toolkit's efficacy and limitations, as well as utilisation. The result of the co-creation study was the Design for Perception toolkit, which has of two variants: A Heuristic Checklist in Excel, and a Card Deck.

early development phases.

 $^{^{3}}$ Co-design is a method, that is part of the Participatory Research (PR) approach, and is characterized by an iterative design process that includes consumers in the design of products intended for them [123]

Ethical Considerations

In the realm of research involving human participants, a paramount ethical consideration revolves around ensuring the well-being, confidentiality, and informed consent of the individuals involved. Online surveys, interviews, naturalistic driving studies, and on-road driving studies are valuable methods employed in various disciplines, but they demand meticulous ethical scrutiny. In all these methodologies, researchers bear the responsibility of upholding the dignity and rights of participants, ensuring their voluntary participation, and guaranteeing that the data collected is utilised responsibly and ethically. Thus, efforts were made for each of the conducted studies to adhere to the guidelines provided by the Swedish Research Council [124].

In the digital age, online surveys are convenient but necessitate stringent data protection measures to safeguard participants' privacy. Thus, the retrieval, storage, and processing of the collected survey responses for **Study I** rigorously adhered to the General Data Protection Regulations (GDPR) of the European Union. The participants were reimbursed for their time through a third-party recruitment agency that provided access to the different markets.

Naturalistic Driving Studies, capturing real-world driving behaviour, require transparent consent processes and anonymization techniques to protect participants' identities. Therefore, prior to the collection of the data for **Study II**, an internal ethical review board reviewed the data collection plan and assessed the potential risks and benefits for participants. This ensured that the study complied with ethical guidelines and protected the privacy of the individuals involved. Even though all participants were part of a co-development fleet that already had mechanisms for data collection in place, they were provided with clear information about the purpose of the study, their rights, and how their data would be handled to ensure informed consent and their agreement to participate in this particular study. Further, the retrieval, storage, and processing of the collected vehicle data and interview responses rigorously adhered to the General Data Protection Regulations (GDPR) of the European Union. The participants who joined the interviews were reimbursed by a cinema voucher for two tickets.

Studies III and IV applied Wizard-of-Oz driving observations in a real-

traffic environment. This demanded careful supervision and adherence to safety protocols to prevent any harm to the participants and the test crew. Thus, the preparation for the studies included meticulous training of the WOzdrivers and HMI-wizards (see Chapters 4.3 and 4.4 for details), as well as an assessment of the test procedure using a risk catalogue. This risk assessment was conducted prior to each study and assessed potential risks that could occur on the designated test route. Each anticipated or potential risk had to be addressed, solutions found, or the approach adjusted in order to ensure the safety of all participants. Further, local authorities and traffic management organisations evaluated and approved the routes, and regular communication within the research team took place in order to address identified issues. This comprehensive approach allowed for a smooth and safe execution of the experiments on public roads. Additionally, all participants were informed prior to the study about the test setup and the data collected during the session in order to ensure informed consent. The data collection, which included the collection of data by means of vehicle-signals, video- and audio-recordings, was conducted in accordance with the General Data Protection Regulations (GDPR) of the European Union. Finally, since both studies utilised a WOzapproach enabling the simulation of technical solutions that were not yet developed, all participants were informed after their driving session about the details of the set-up and had the chance to ask questions and share their experience with the test leaders. Local recruitment agencies paid all participants for their time.

During **Study V**, colleagues of the author were recruited to participate in the co-creation study. All participants participated on a voluntary basis, out of interest and support for the topic. Due to the study taking place during working hours, the participants were reimbursed through their salaries. The acquisition and analysis of data during the co-creation stage and subsequent interviews were conducted with the explicit consent of all participants, adhering to the guidelines outlined in the EU General Data Protection Regulation (GDPR).

3.4 Analysis and Evaluation

The following sections describes the steps taken during the cross-study analysis in detail, and how the collected knowledge was turned into actionable insights in form of a design toolkit.

Cross-Study Analysis

The empirical studies were followed by a cross-study analysis that synthesised the results of the four empirical studies into a summary of the process by which the driver's perception influences their understanding of DAS. The cross-study analysis primarily depends on the driver's verbal accounts and observed behaviours while using a driving automation system, as well as their descriptions of the systems and their usage gathered during the conducted interviews.

The 'Framework Approach' [125] was utilised to approach the large and complex dataset. The framework approach, also referred to as framework analysis, is an inherently comparative form of thematic analysis that uses an organised structure of inductively and deductively derived themes (i.e., thematic framework) to conduct cross-study analyses by combining data description and abstraction [125][126]. These methods facilitate the identification of similarities and differences in qualitative data prior to focusing on the relationships between the various data components and attempting to draw descriptive and/or explanatory conclusions. The defining characteristic of framework analysis is that it follows explicit steps to generate a structured output in a matrix or table (cases, codes, and summarised data), which is then used to systematically reduce the data [125]. Since the overall objective of framework analysis is to identify, describe, and interpret critical patterns across themes within the phenomenon of interest, it was chosen as the method to address RQ1: "What are the factors that impact the driver's perception and subsequent understanding of driving automation systems?".

The data analysis was divided into five stages in accordance with the framework methodology:

Data Familiarisation: Immersion into the collected data and becoming familiar with the raw information. The purpose of this step was to

gain a deeper understanding of the data's contents and context.

- **Framework Identification:** Identifying an appropriate analytical framework. A first step in the framework identification was the mapping of data into concept maps, allowing to organise the data. The purpose of this step was to establish a structured approach for categorising and understanding the large amount of collected data and its nuances.
- **Indexing:** Systematic coding and labelling of segments in the data set. The codes represent themes, which were later categorised into different aspects and factors, enabling to identify recurring patterns across all the four studies and data types, allowing to triangulate the data.

- **Charting:** In this step, the indexed data was condensed into matrixes, summarising and facilitating an overview of the identified categories, which aided in the identification of relationships and hierarchies.
- Mapping and Interpretation: Finally, the matrixes were analysed deeper, and its contents compared in order to identify overarching themes with regard to the driver's perception or understanding. The purpose of this step was to interpret and synthesise the data into tables, organised by the different categories and typologies.

The final stage was conducted iteratively and yielded a combination of key learnings that summarise the factors describing the driver's perception and the aspects comprising the driver's understanding of DAS. In addition, the structured approach generated typologies and enabled the mapping of connections between and within the identified aspects.

This, ultimately led to the development of a model illustrating the process of how perception shapes understanding, including the identified factors of perception, and aspects of understanding associated with DAS. This model developed through an iterative process, and by conducting a workshop with colleagues who are experts in the area of automation and cognition and scrutinizing the iterated ideas in discussions and thought experiments.

The result was a unified descriptive model, incorporating all identified factors, describing how user's perception shapes, through a top-down and bottom-up process, the understanding of a DAS. The unified descriptive model is presented in detail in **Chapter 5**, and constitutes the theoretical contribution made in this work.

Development and Evaluation of the Design Toolkit

Since the results of the cross-study analysis were already structured into tables, these tables were transferred into Excel and organised into two different sheets: (i) aspects of understanding and (ii) factors of perception. The table containing the aspects describing the driver's understanding was further provided with supporting questions and columns allowing the user (e.g., the designer assessing a design solution) to add information about the technical specifications of the desired solution, as well as a column where usage scenarios could be described. The table containing the aspects describing the driver's perceptions was similarly structured and provided with supporting questions. Additional columns aimed to foster a deeper discussion around possible failures and effects of the provided design solutions.

This initial version was used as a starting point for the co-creation study, which was organised around a use case study and a series of workshops. The purpose of the workshops was to build a new user interface for DAS applying the Design for Perception toolkit. When creating the interface, the toolkit provided a systematic framework that led the participants to address crucial aspects and factors. The workshops offered a cooperative environment in which participants could share their experiences with and requirements on the design tool, with the purpose of contributing to the design process for the design toolkit. The following sections briefly describe the development activities and the evaluation of the toolkit.

Developing the Toolkit

Throughout the development of the toolkit, the author implemented multiple iterations and feedback cycles to ensure that the toolkit aligned with the feedback and requirements of the team. This approach continued throughout the duration of the co-creation project, running in parallel with the workshops. The aim of this was to gather input from the workshops and engage the designers during the prototyping stage to elicit their feedback and foster their active involvement in designing the toolkit. Further, the author analysed the outcomes of the workshops, discussed them with colleagues, and formulated solutions throughout the project with the aim of identifying and implementing solutions that could effectively tackle the requirements and obstacles mentioned by the designers. In order to ensure the effectiveness of this collaboration, a system of regular meetings and communication channels was implemented. This facilitated continuous collaboration and the exchange of ideas between the author and the designers, thereby promoting a sense of ownership and shared responsibility in the design process.

The collaboration was shaped by a range of activities, such as designers offering input on the author's proposed solutions and actively engaging in the conceptualization and creation of solutions. This collaboration typically centred on the various variants of the toolkit and the structuring of information and contributed to the development and design of a card deck variant of the toolkit, with the goal of making the use of the toolkit more accessible and easier to incorporate into the designers' daily tasks. Additionally, there were frequent discussions pertaining to the integration of the toolkit within existing processes and its potential to improve or optimize such processes. The objective was to guarantee the seamless integration of the toolkit into the current workflow without introducing any new stages or complexities.

After the conclusion of the workshop series, the prototyping co-creation was continued. As a group representative, one of the designers worked closely with the author over an extended period of time, concentrating predominantly on the card deck (physical and digital) variant of the toolkit. Evaluating the Toolkit

To gather thoughts and comments on the efficacy, limitations, and possible enhancements of the design toolkit, a series of semi-structured interviews were conducted with the participants in Study V. The purpose of the interviews was to assess the results from the workshops and evaluate the perceived value of the toolkit among the designers, as well as identify areas for potential further enhancements.

The interviews were held after a period of three months had passed. During these months, the participants had been granted unrestricted access to the toolkit and its variants, enabling them to apply it in a manner they deemed suitable. Hence, it was considered an opportune moment to assess the utilization and application of the toolkit.

The author created a topic guide for the interviews, which was organized into three main areas: (i) the daily activities and challenges faced by designers, (ii) the toolkit's utilization and perceived usefulness, and (iii) the positive and negative aspects of the toolkit. The interview guide was not exhaustive and allowed for additional questions. The interviews were carried out by a colleague acting as an impartial third party in order to mitigate bias and guarantee that the respondent's answers were not affected by the author's affiliation with the team. The conducted interviews yielded significant insights into the potential challenges and benefits associated with the integration of the toolkit into established design workflows. Further, the panel of experts offered recommendations for further additions and optimizations to optimize the toolkit's efficacy in practical design situations.

The interviews were transcribed verbatim. Subsequently, the transcriptions were subjected to coding and analysis. Due to the fact that the structure and content of the interviews were based on a topic guide, a deductive thematic coding strategy was used in the first phase (cf. [127]).

The codes underwent a process of revision, refinement, and validation after the first coding phase. During the thematic organization of the codes, statements pertaining to the usage scenarios, strengths and limitations, potential enhancements, and additional value of the toolkit were discerned. From the findings, it became evident that the designers held the belief that the various versions of the toolkit could be employed in distinct ways. Consequently, significant emphasis was placed on the particular representation and its usefulness when consolidating the gathered insights, despite the fact that the contents of the toolkit remained unchanged.

Chapter 4 Summary of Studies



CHAPTER 4

Summary of Studies

This chapter lays the groundwork for solving the research questions. The responses are based on a summary of the investigations that were carried out in Studies I-IV, and details the results from the cross-study analysis for each of the conducted studies independently. Only relevant results are reported in the dedicated sections. It concludes with presenting the co-creation approach applied during Study V that enabled the development of a design tool, based on the key findings.

4.1 Study I: Context of Use

The study's objective was to gather insights into drivers' usage of DAS in different driving contexts, by means of an online survey. The survey was exploratory in nature and relied on the driver's self-assessment of their use of Level 1 and Level 2 driving automation system in their personal vehicles. A detailed description of this study can be found in Paper A.

Aim

The aim of this study was to understand the extent to which drivers use DAS. Most studies have addressed Level 1 DAS, particularly Adaptive Cruise Control. To the author's knowledge, no study has compared the use of Level 1 and Level 2 DAS. Additionally, few studies have studied how context, specifically the traffic environment and traffic circumstances, affect DAS use. Thus, this study compares driver's use of these systems in different driving contexts by means of an international online survey.

Method

The survey was administered to 2,000 drivers in Germany (DE), Spain (ES), the United States (US), and China (CHN) and covered their usage and experience with Level 1 and Level 2 DAS. The ages of the respondents ranged from 18 to over 65. The majority of respondents reported annual mileage between 5,001 and 20,000 kilometres, a smaller proportion between 20,0001 and 30,000 kilometres, and a very few between 5,000 and 30,000 kilometres. The driving contexts in all countries were mainly urban areas, followed by motorways or highways, and then the countryside, with a fairly even distribution between these, with the exception of China, where the majority of participants drove primarily in urban areas, with close to no driving in the other driving contexts.

The responses were collected via an online survey which was sent out via email by a third party with access to the various markets. There were a total of 36 questions, and it took participants between 10 and 15 minutes to complete the survey. The majority of the questions were Likert-type [128], scenario-based statements querying the participants on their use of the DAS in their personal vehicles in different driving contexts, e.g., road types, weather and light conditions, road conditions and the drivers' physical and mental condition.

Findings

The key findings of this study concentrate on the different driving contexts drivers report when utilizing DAS.

In summary, the results indicate that system usage was context-dependent and that the systems were not utilised in the same manner across all contexts. Overall, respondents utilized Level 2 systems slightly more than Level 1 systems. In variable traffic conditions, the Level 2 system was used in both less congested and heavy traffic situations. This result may be explained by the fact that the vehicles were equipped with both systems and users had the option to use the Level 2 system, which incorporates all the capabilities of the Level 1 system plus a Lane Keeping Aid (LKA). As the system relieves the driver of a portion of the driving task, the additional support function may also explain why drivers reported a greater reliance on the Level 2 system when they were fatigued or when the driving was monotonous. The respondents utilized the Level 2 system over the Level 1 system when driving in urban areas. On the highway, however, both systems were utilised frequently, which may be attributed to the overall assistance they provide, such as sustaining a speed and maintaining a safe distance from other vehicles.

However, the results also indicate that in adverse weather conditions or at night, when visibility is poor, a greater proportion of drivers preferred to drive themselves rather than rely on automated systems. Under certain weather conditions, this decision can be attributed to the limitations of the Level 2 system, which relies on visible lane markings and unobstructed lines of sight for sensors and cameras, as well as the driver's lack of trust in the system's ability to handle adverse weather conditions in daytime driving conditions. Finally, there were preferences for using both systems, but especially the Level 2 systems, when the driving was deemed monotonous, and generally a tendency to use the systems when the drivers were tired, which can be assumed to be due to the assistive features of the systems.

Summary of Insights

- ◇ The driving context and thus the question "When can I use the system?" appears essential for the driver's understanding of the DAS's capabilities and limitations – irrespective of the automation level.
- ◊ The context includes both driving situations and a driver's physical or mental condition.
- ♦ Aspects falling under driving context are road types, traffic conditions, weather conditions and time of day.
- ♦ Aspects falling under personal the driver's physical or mental condition are boredom or monotony, and tiredness.

♦ While DAS are not designed to handle all traffic, weather and road conditions, the results from this study showed that participants did not always use the systems as the manufacturer intended. On the contrary, frequent usage occurred outside of the intended operational design domain (ODD).

4.2 Study II: Context of Use and Understanding

The second study was exploratory in nature and employed a sequential mixedmethods research design to investigate the use of Level 1 and Level 2 DAS. This study's aim was to investigate the factors influencing the drivers' utilisation and understanding of DAS by triangulating data from a Naturalistic Driving (ND) study with clarifications and reflections from in-depth interviews with selected participants. Additional comprehensive findings are documented in the appended Papers B and C.

Aim

Drivers must have a solid understanding of the various modes of operation and inherent limitations of DAS in order to utilise them safely and effectively, as well as gain the benefits they provide. Failure to acquire this knowledge will hinder the development of appropriate trust and the formulation of effective usage strategies. Hence, the aim of this investigation was to examine the motivations behind drivers' usage strategies when utilizing Level 1 and Level 2 DAS and determine the factors that impact their understanding of the investigated systems.

Method

This was accomplished by adopting an Explanatory Sequential Mixed Methods [120] methodology, combining data from a naturalistic driving (ND) study with insights and reflections obtained through in-depth interviews with a select group of participants. The sequential application of quantitative and qualitative approaches (see **Figure 4.1**) was intended to facilitate an integrated interpretation of the effect of driving context on DAS utilisation. The ND study was conducted over a period of seven months, collecting data from 132 vehicles. Driver behaviour and system performance were categorised and measured in order to evaluate them independently and investigate their relationship. To evaluate driving contexts, the evaluation incorporated vehicle speed, driving distance, time of day, GPS data, wiper sensor status, road conditions data, and other sensor data. This analysis allowed for the identification of the driving conditions (e.g., road, traffic, and weather) in which the driver utilised the systems or decided to disable them.



Figure 4.1: Explanatory Sequential Mixed-Methods Design.

Subsequently, using in-depth, semi-structured interviews, an investigation and validation of the quantitative data were conducted to investigate the individual driver experience and understanding of the systems. The interview consisted of four sections: Contextual Information, System Usage and Scenarios, Perception and System Experience, and Information Display and Controls. Accordingly, a series of open-ended questions were formulated based on the four major themes. The structure of and questions in the interview were based on the preliminary findings of Study I and the learnings from the ND study.

Findings

The key findings of this study concentrate on insights as to when drivers utilize DAS, and the factors affecting the usage and the driver's perception of the systems.

According to the interviews, the majority of interviewees concurred that using the systems on long trips (over 50 kilometres) was preferable to short trips, where the majority of drivers did not use them. Adaptive Cruise Control (ACC) was highlighted throughout the interviews as a significant aid in managing speed, safety, and comfort on long trips. This demonstrates that the length of the journey was an important factor in system utilization and provides an explanation for the preference for use on highways over city driving (something that was also found in Study I). In urban environments, the infrastructure or other traffic required frequent deceleration and driver engagement, causing the driver to need to activate and deactivate the systems with excessive frequency. Several participants preferred the Level 2 DAS in slow-moving or congested traffic due to its more stable steering assistance. Rural roads were also deemed as inappropriate for system activation. The interviewees reported that rural roads frequently lack visible or distinct lane markings, causing the system to switch between active and standby, particularly in poor weather or low light. In extreme conditions, the systems may not function at all; therefore, participants preferred to drive manually rather than engage the DAS.

From the narratives of the drivers, one can conclude that perceived system performance affected whether and when the DAS was utilised. The Level 2 system's steering support was unstable, especially on country roads, so the majority of drivers were hesitant to use it. In variable traffic conditions with a large number of vehicles, participants chose to drive themselves, indicating a preference for control over the driving task. Others felt relieved when DAS assumed some driving responsibilities. These contradictory statements demonstrate that perceived control influences the driver's understanding of the system's authority over the driving task. Some believe advanced support means they can let go of the driving task, whereas others recognise they must always be involved and ready to intervene, but still appreciate the additional support. Overall, the system performance seemed to affect how much control drivers perceive they have over the system, which reflects their understanding of their responsibility for the driving task and their willingness to use the systems. Nevertheless, the majority of drivers found both automated driving systems to be useful. The systems were easy to use and supported them during the drive and were particularly useful for primary driving tasks such as accelerating, decelerating, and maintaining a safe distance or speed limit. This was frequently mentioned in connection with long-distance travel and highway monotony. All drivers who were regular DAS users reported feeling more physical and mental relief and at ease as a result of the enhanced comfort. When utilising the system, all drivers reported following the traffic flow rather than overtaking and setting their own tempo, which was perceived as safer and less aggressive driving behaviour. Numerous interviewees valued the enhanced safety they associated with the systems. Participants viewed PA as an additional pair of eves or quick reflexes that assisted them in focusing on a phone call or driving while fatigued.

Further, the interviews revealed that the driver's preconceptions about the systems influenced their learning and their mental representations of the system's capabilities and limitations. Previous usage appears to generate assumptions when encountering unfamiliar systems. When asked what was anticipated from the system, responses ranged from convenience and safety features to more advanced systems that took over driving for a certain amount of time. Depending on the driver's expectations and prior usage experiences, the outcome was either disappointing or beneficial. Thus, these expectations shaped the driver's perception of the system. However, the drivers' statements demonstrate that the system's capabilities differ from their expectations. The technology lacks situational awareness, but drivers appeared to expect a more intelligent system that would read traffic conditions such as traffic signals and other road users, as well as predict their behaviour. Some participants were more hesitant, stating that they did not know what to expect: consequently, their experience was positive. Those whose expectations were met or exceeded during the learning period had a lower threshold during testing and learning how to use the system, resulting in greater acceptance. Users with high hopes were dissatisfied with the systems and ceased using them. The causes of dissatisfaction were often linked to feelings of uncertainty and inadequate feedback provided by the system.

In numerous instances, drivers expressed their lack of awareness regarding system status changes, specifically noting that lateral control was unavailable while longitudinal control remained active. This phenomenon resulted in confusion and, in certain instances, hazardous situations, as the drivers failed to notice the change in the automation status and continued to rely on the lane keeping aid. Their failure to notice these inherent mode changes was ascribed to a lack of auditory feedback. A status indication solely on the displays failed to notify drivers, showing that drivers expected to have clearer indications and feedback about the system status.

Another commonly noted occurrence was the issuance of a hands-off alert by the system while drivers were not actively engaged in steering. The comments revealed that the perceived purpose of the driving automation differed significantly from the actual purpose of the system. Numerous drivers expressed their annoyance over the persistent prompts from the system, which consistently reminded them to keep their hands on the steering wheel. This frustration stemmed from the assumption, derived from the system's performance, that it was a hands-free system.

This shows that the drivers made assumptions about the system's capabilities that ascribed to them capabilities beyond what was intended by the manufacturer.

However, the feedback they got from the systems did not succeed in clarifying their responsibility for the driving task, causing frustration and in some cases the drivers ceased using the systems.

Lastly, trust was discussed as a result of the learning experiences. One of the most significant aspects of interviewees' trust in the system was its consistency. The drivers provided multiple clarifications, including about system performance. There were examples of functions that did not perform as anticipated, placing drivers in uncomfortable situations. The majority of drivers characterised the system as a support function, but they wanted it to mimic human driving behaviour and adapt to changing conditions. This suggests that some drivers have exaggerated expectations of the capabilities of the systems. The drivers also indicated that they had to develop trust in the system over time. There appears to be a learning phase in which the driver either gains sufficient trust to accept what the system can do and use it or encounters too many negative situations or inconsistencies that they cannot explain and therefore never accepts or uses the system. Further, it was indicated that their trust in the system varied depending on the driving context. They indicated that they chose a function or drove themselves based on the road type, the weather, and the amount of light. This implies that the greater the user's understanding of the system's capabilities and limitations, the more effectively expectations can be met, and trust can be established.

Summary of Insights

- ♦ The driving context and thus the question "When can I use the system?" appears fundamental for the driver's understanding of the DAS's capabilities and limitations independent of the automation level.
 - $\diamond\,$ The driving context includes aspects like road types, traffic condi-

tions, weather conditions, time of day and trip types.

- ◊ Aspects that fall under personal conditions are monotony, tiredness, physical and mental stress from the driving task.
- ◇ The vehicle's capabilities and limitations were crucial in understanding many aspects of the interaction, thus the question "What does the vehicle do?" emerged as another central guide for drivers.
 - ◇ Vehicle behaviour and how drivers perceived the system's performance influenced how capable they perceived the systems to be and their inclination to use them – this too was context-dependent.
- ◇ The question of what the driver's tasks were when using the systems emerged as "What do I do (now)?" and appeared to guide the driver's understanding of how much attention they needed to pay to the DAS and who was responsible and in control of the driving task at what time.
- Preconceptions seemed to play a significant role with regard to accepting and using the systems.
 - ♦ The system's capabilities, its usefulness and consequently the purpose of the system guided the drivers in their assessment.
- Previous experiences with other or similar systems set the driver up for how they would perceive a new system when they encountered it.
- ◇ Feedback with regard to the automation status was found to be relevant to understanding the system's purpose. However, their perceptions of the system performance could contradict and misguide drivers in their interpretation of feedback.
- ◇ Trust was closely connected to the driving context and the learning experience. The level of trust varied as well as the driver's perceptions of in which situations the systems could be trusted.

4.3 Study III: Perception and Understanding

The third study utilized a Wizard-of-Oz (WOz) driving study on actual roads and was exploratory in nature. This study's aim was to assess the driver's perceptions, and subsequent understanding, of the interaction with a driving automation system that offers multiple modes of operation, with the focus being on a level 2 partial automation function and a level 4 high automation function. For a greater level of detail, please refer to the findings presented in Papers D, E and F.

Aim

As a division of responsibilities between the driver and the vehicle in the driving task becomes more prevalent, there is a possibility that the driver will encounter instances where the assumed system mode differs from the actual driving mode, resulting in incorrect actions and uncertainty regarding the driving mode and the driver's responsibility. One potential factor contributing to this lack of understanding regarding automation levels and forms of engagement could be the driver's perception of the relationship and division of responsibilities between themselves and the automated system. Therefore, the aim of this study was to investigate the driver's understanding of vehicles equipped with multiple levels of automation in a real-world setting.

Method

This was achieved by gathering insights from an empirical road study conducted in the San Francisco Bay Area, California, United States, in June 2019, with 20 participants: 11 females and 9 males ranging in age from 22 to 62 years (Mean=42, SD=14). During the study, participants encountered two different driving modes (levels of automation) in a Wizard-of-Oz (WOz) vehicle: a Level 2 partial automation system and a Level 4 high automation system [7].

Equipment

Vehicle. The WOz vehicle was a prototype based on a Volvo XC90 platform that was modified to facilitate testing of the two levels of automation. The modified platform suggests to the driver that the vehicle is performing some or all of the driving task, simulating an authentic user experience for the participants. The simulation was made feasible by installing a steering wheel, instrument cluster, and pedals in the rear seat, allowing the vehicle to be driven from this position by the driving wizard. The human machine interface (HMI) prompts were orchestrated by an HMI wizard via a tablet according to a set of pre-defined rules. The installation was concealed and could be seen by those seated in the front seats.

The vehicle was modified in accordance with all applicable road permission standards and audited and approved for road testing by the local authorities, allowing for the investigation of drivers' experiences with automated driving in a real-world driving context. **Figure 4.2** exemplifies the setup in the vehicle, including all wizards, the test leader, and the participant.



Figure 4.2: The Wizard-of-Oz vehicle, with video cameras facing the UI, and (1) driving wizard, (2) HMI wizard, (3) test leader and (4) participant.

Automated Driving Systems. The Level 2 system was capable of automatically adjusting the vehicle's speed in relation to other moving objects in front, or sustaining a set speed, as well as providing lane-keeping assistance, but lacked advanced steering capability. This driving mode could be activated at any time by the driver and was considered a supervised driving system. This is because, even though the system provided both lateral and longitudinal support, the driver remained fully in charge and responsible for the driving task.

The Level 4 system was an unsupervised driving mode that operated within a specified operational design domain (ODD), i.e., under certain conditions. Since traffic congestion was a daily occurrence in the region where the study was conducted, it was decided to limit the context of this system to a congested traffic scenario. It was necessary to have dense traffic with other vehicles in front of the vehicle for the mode to be available. Upon encountering such a circumstance, the driver could activate the system, and the vehicle would assume full control of the vehicle as long as the conditions were met. If this was no longer the case, for example, if the congestion cleared, the driver was prompted to retake control with a visual and auditory cue and a one-minute preparation time. However, while the Level 4 system was engaged, the driver was free to engage in other activities than the driving task.

Human Machine Interface (HMI). For this study, a graphical user interface (GUI) concept was developed and embedded into a prototype that offered various means of feedback to the driver, i.e., visual, auditory, and haptic feedback. In the context of this study, the term HMI pertains to a multi-modal user interface, while GUI specifically pertains to the visual concept and associated auditory elements. The different sequences and states of each of the developed GUI are presented in comparison in Figure 4.3.

The HMI wizard controlled the HMI from the rear seat via a tablet, and various graphics were displayed based on the vehicle's status. When the availability conditions were met, an offer to be driven by the system was displayed in the visual displays of the vehicle (i.e., instrument cluster and centre display) along with an auditory alert. To activate or deactivate the system, the driver was required to press two buttons on the left and right sides of the steering wheel for a prolonged time until the system activated or deactivated. During automated driving, the instrument cluster displayed graphics depicting the vehicle's current intent. When the system's availability conditions were no longer met, participants were instructed to resume manual driving, by first receiving a tug from the seatbelt, and then auditory and visual cues about what they were supposed to do.

Procedure

All twenty participants experienced both modes during the driving session. Participants could use the Level 2 system whenever they wished, but they



Figure 4.3: Visual and Auditory description of GUI sequence and states for each of the driving modes.

could only use the Level 4 system when they received a prompt indicating that the necessary contextual conditions had been met and the system was available. Before each session, participants received information in written form during the test's introduction, as well as in the car prior to the driving sessions, and they were given the opportunity to ask questions. The information the participants received about the two systems is summarised in **Table 4.1**.

The introduction of the systems was comparable to what a new car customer would receive when purchasing a vehicle from a dealership. This decision was made in order to provide a high level of realism, as the purpose of the study was to investigate the driver's understanding of a vehicle with multiple levels of automation and intensive driver training is typically not provided when a vehicle is delivered.

Driving Observations. To satisfy the two distinct conditions for the operational design domain of the system, a route with multiple instances of slow or halted traffic and a section with free-flowing traffic was required. Based on

 Table 4.1: Description of DAS capabilities and limitations and information participants received prior to the driving session in Study III.

System	Description	ODD	Limitations	Interaction
Level 2	-Maintains speed	-Always available	-Clear view of lane	- Driver in control of steering
	-Adjusts speed to		markings	- Driver responsible at all times
	vehicle in front			- Activation/deactivation via
	-Lane keeping support			steering wheel button
Level 4	-Takes over the driving	- "Bumper to	- System prompts	- Vehicle in full control
	task completely	bumper"/slow	when available	- Driver does not need to
	under congested	moving traffic	- Clear view of lane	supervise and can engage in
	traffic conditions		markings	secondary tasks while system
				is active
				- Activation/deactivation via
				long press of steering wheel
				button

the collected traffic data and the results of several test drives, a round-trip route including various road types was selected.

Thus, the driving sessions were conducted on highways (US highway 101: 6-lane partially controlled access road, speed limit = 70mph, LOS E-F; Interstate 280: 8-lane full access road, speed limit = 70mph, LOS C-D) and urban areas (State Route-84: 4-lane, speed limit 25-50mph, LOS B-C) in the San Francisco Bay area (see **Figure 4.4**). All trips were conducted during rush hour, in the morning and in the evening, with the intention of observing the interaction between the two modes the participants experienced. During the test drive, the participants were encouraged to test and experiment with the systems as much as they wanted, and asked to verbalise their thoughts so that their thought processes during interaction with the system could be documented.

Post-Interviews. After the drive, the session concluded with an interview focusing on the participant's understanding of the two systems they had experienced and their comparison. The participants were asked to describe and explain the various systems they encountered during the driving session. In particular, they were asked to describe the distinctions between the two



Figure 4.4: Route for the observation study in the San Francisco Bay area.

modes, and their perception of the systems during the different phases of the trip.

Findings

The key findings of this study concentrate on identifying how drivers perceive and consequently understand DAS, specifically in the interaction with a system offering several levels of automation. The following section summarizes the findings from the study.

Prior to the driving sessions, participants were questioned regarding their expectations of a highly automated system. All participants agreed that it would have a positive effect on their stress levels because they would be able to delegate driving tasks to the vehicle when they did not wish to drive. Thus, commuting to and from work, particularly during rush hour and congested traffic, was one of the most frequently mentioned situations. Similarly, participants anticipated being able to use the system on lengthy highway journeys, such as when travelling in the vehicle for extended periods. While both systems were believed to be useful on long journeys, it was determined that the Level 2 system was superior in free-flowing traffic, whereas the Level 4 system was superior on highways and in congested traffic.

After the driving session, in addition to what they believed prior to the driving session, participants indicated whether the Level 2 system would be a beneficial to use when they were fatigued or unable to focus due to personal conditions, such as being tired. The participants further stated that the system lowered their perceived stress levels and helped them to feel more relaxed. Moreover, some participants indicated that the system would enable them to be more productive or socialize with passengers, as they would be liberated from the driving task and could engage in other activities, such as day planning, email writing, and phone calls. In addition, it appeared to the participants that the system made driving simpler by relieving them of physical and mental workloads such as accelerating, braking, and maintaining a predetermined distance between vehicles. Furthermore, it was believed that having a system that takes over part or all of the driving is particularly useful when commuting to and from work, but also on longer trips, such as travel with the family (see also Paper D). However, the participants primarily emphasized that having a DAS would contribute to safer driving and, as a result, reduce the number of traffic accidents. The two systems experienced were referred to as an additional pair of eyes and were valued for their assistance in maintaining speed limits. Nevertheless, some participants were unsure of what the Level 2 system provided and were dissatisfied with the system's performance because they felt that too much interaction was required (for instance, irritation that they had to be constantly engaged in monitoring and steering) and therefore did not see how they benefited from using it, indicating that they misunderstood the system's purpose.

When discussing the purpose of the system, control was a frequent theme. Participants perceived the driver to be in control of the vehicle or to be receiving assistance from or sharing control with the vehicle when using the Level 2 system. As a result of the presumed shared control, several participants expressed confusion regarding control allocation when the system was operational, resulting in ambiguity regarding their actual responsibilities. Some participants believed they could disengage even though the Level 2 system does not perform the driving task for the driver but merely assists them. The Level 4 system was characterized as completely taking over control of the driving task, requiring minimal human input, and making the driver feel like a passenger in the vehicle. However, with the ability to retake control of the vehicle at any moment by deactivating the system, the majority of participants did not feel out of control, with only a few participants feeling uncertain about how much responsibility they could delegate to the vehicle. As they had activated the system and were seated in the driver's position however, many participants still felt responsible for monitoring the vehicle. Ultimately, control assignment during the various operating modes led to confusion, particularly when using the Level 2 system. Such confusion is risky because it can lead to difficulties for drivers to understand how much responsibility they must assume for the driving task and when they can disengage.

Beyond the aspects of perceived control, which seem partly related to the driver's expectations of the systems and partly to the haptic feedback they perceived from the vehicle, other information sources were also considered relevant when trying to understand the vehicle's behaviour. Obvious sources are the visual and auditory cues the vehicle gave the participants, such as indications in the vehicle's instrument cluster and infotainment system, as well as haptic feedback received from the seat belt retraction of the vehicle. However, more factors could be observed influencing the driver's perception of the vehicle.

During the interview, participants were asked if they trusted the systems they had encountered. All participants noted that, over time, they developed a measure of trust in the system and a desire to continue using it. Initial apprehensions were due to a lack of familiarity with the capabilities of the systems, as all participants had only ever used a Level 1 system in their personal vehicles. As a result, one of the reasons they gained trust in the system was because they witnessed the vehicle managing a variety of situations, such as matching their speed to that of the vehicle ahead or reacting to merging traffic. Thus, vehicle behaviour was interpreted as indicating that the vehicle had situational awareness comparable to that of a human driver, being able to anticipate the behaviour of other road users, and therefore was viewed as more intelligent and trustworthy than a system that only reacts to what it encounters. Comments like this were also frequently noted during the driving sessions.

Finally, the vehicle brand was cited multiple times as a reliable brand with a reputation for producing safe vehicles, prompting the participants to place trust in the unknown systems. However, different brands and their reputations were also discussed in light of recent technological developments, and information sources identified were various media outlets and social circles. In addition, the current legislative status of automated vehicles was considered and debated when discussing the Level 4 system, and whether drivers should relinquish control to engage in other tasks. Consequently, these secondary sources (i.e., non-experienced but otherwise learned information) influenced the drivers' expectations and the perceived safety of the systems experienced, as well as their willingness to use them.

Summary of Insights

- ◊ Also in Study III, the driving context and thus the question "When can I use the system?" appears essential for the driver's understanding.
 - $\diamond\,$ The driving context included aspects like road types, traffic conditions, weather conditions, time of day and trip types.
 - ◊ Aspects that fall under personal conditions are monotony, tiredness, physical and mental stress through the driving task.
- ◇ The vehicle's abilities were prominent in understanding many aspects of the interaction, thus the question "What does the vehicle do?" emerged here too as another central guide for drivers.
 - ◊ Vehicle behaviour and how drivers perceived the system's performance influenced how capable they perceived the systems to be.
 - $\diamond\,$ The less human involvement was required, the smarter and more capable the system was perceived to be.
- ◇ Trust was closely connected to the learning experience and seemed to be affected (either positively, or negatively) by observing how the systems

handled different driving situations.

- ◇ Previous experiences with other or similar systems were named as a benchmark and used to compare capabilities, in order to gain an understanding of the systems experienced.
- ◊ The question "What do I do (now)?" emerged as well, especially when trying to make sense of the different driving modes encountered.
 - ◊ The system performance and haptic feedback seemed to influence the driver's perception of who was in control in which driving mode.
- ◊ Preconceptions were relevant with regard to using an unknown system.
 - ◇ Information received through other sources than actual usage, including the media, social circles, and legislation, influences the driver's perception of the systems and their willingness to use them.
- ◊ Information about the vehicle's capabilities and intentions was received through multiple sensory channels.

4.4 Study IV: Perception and Understanding

The fourth study was exploratory-comparative in nature and utilised the Wizard-of-Oz (WOz) method as well as an A/B HMI test. This study's objective was to investigate the driver's understanding and interaction with a driving automation system that offers multiple modes of operation, with a focus on the Level 2 partial automation function and the Level 3 high automation function. For more insights, please refer to the findings presented in Paper G.

Aim

With the introduction of several driving modes into vehicles, the significance of developing efficient ways of communicating a vehicles capabilities and limitations with the driver is growing. The driver's perception of the division of responsibilities between themselves and the DAS may influence their understanding and cause uncertainties about the currently active and available driving modes in a vehicle. One potential way of addressing uncertainties is through the HMI. Therefore, the aim of this study was to investigate what influences the driver's perception and consequent understanding of vehicles equipped with multiple levels of automation (i.e., Level 2 and Level 3 DAS), by contrasting two different graphical user interfaces (GUI) concepts, in a real-world setting.

Method

This study was an empirical road study conducted in Gothenburg, Sweden, in September 2022, with 16 participants (7 female and 9 male), ranging in age from 23 to 70 years old (M = 44, SD = 13.48). During the study, participants encountered two different modes (levels of automation) in a Wizard-of-Oz (WOz) vehicle: a Level 2 partial automation system and a Level 3 high automation system [7]. Furthermore, the study utilized an A/B UI test in order to compare the influence of different UI elements on the driver's understanding and interaction with each of the systems.

Equipment

Vehicle. The test vehicle was a modified Volvo XC90 that utilised a WOz approach to simulate the two DAS, as well as to control the UI via controls in the back seat that were concealed from the driver in the front. In this configuration, a driving wizard was in charge of operating the simulated Level 3 DAS, while an HMI wizard was responsible for controlling the prompts and feedback to the driver received from the UI. In addition, cameras facing the UI and interaction were installed behind the driver to capture observational data regarding the driver's interaction with the vehicle's systems.

The vehicle was modified in accordance with all applicable road permission standards and audited and approved for road testing by the local authorities, allowing for the investigation of drivers' experiences with automated driving in a real-world driving context. Refer to **Figure 4.5** for an illustration of the configuration.



Figure 4.5: The Wizard-of-Oz vehicle, with video cameras facing the UI, and (1) driving wizard, (2) HMI wizard, (3) test leader and (4) participant.

Automated Driving Systems. During the driving sessions, participants had the option of engaging a supervised (Level 2 Partial Driving Automation) or unsupervised (Level 3 Conditional Driving Automation) DAS.

The Level 2 DAS was the Volvo Cars Pilot Assist system that was already present in the vehicle and was described to the participants as having the ability to maintain a set speed, modify the distance between them and the vehicle in front, and provide steering assistance. Nevertheless, the driver remained in charge and accountable for the driving task. This system was always accessible, and drivers were encouraged to utilise it as frequently as they desired.

The Level 3 DAS was designed to operate autonomously within an operational design domain (ODD). The system was described as one that alerts the driver when available. Upon activation, it would assume full control of the driving task until the conditions were no longer met. The availability conditions depended on the external traffic environment and were defined as follows: (i) partially controlled access roads with up to 80kph speed limits; (ii) freeflowing traffic (LOS A-B), meaning that the system would prompt takeover when merging traffic was approaching; and (iii) clear view of lane markings. When these conditions were met, the driver could activate the system and the driving wizard would take control of the vehicle. When the conditions were no longer met, the driver would be prompted to resume control of the vehicle.

Human Machine Interface (HMI). This study presented two different HMI concepts. Both concepts were developed to support the driver in the interaction with a vehicle offering several driving modes: (i) manual driving, (ii) Level 2 Partial Automation, and (iii) Level 3 Conditional Driving Automation. The two concepts differed mainly in their graphical representation, while interaction schemes and technical solutions remained the same. Thus, in the context of this study, the term HMI encompasses a multi-modal user interface (i.e., employing visual, auditory and haptic feedback), while GUI specifically pertains to the visual concept and associated auditory elements of the various concepts.

The intention of **GUI A** (developed for and used in Study III) was to provide the driver with only the pertinent information in each driving mode. The designers' intention was to be clear about the vehicle's intentions and the driver's responsibilities at all times by providing only the necessary information at any given time and removing other information that could confuse the driver about the driving modes they encountered.

The **GUI B** was developed during a series of workshops where the Design for Perception toolkit was utilised as a design tool (see Paper H). This GUI's underlying concept was derived from a conceptual model describing the factors
that influence the driver's perception and, as a result, affect their understanding of a DAS and their interaction with it. This was addressed primarily by attempting to answer the following three questions throughout all design phases, and for each driving mode the driver would encounter: When can I use the system for? What can the vehicle do? What should I do?

During the driving sessions, the HMI wizard controlled the HMI from the rear seat via a tablet, and various graphics were displayed based on the vehicle's status. When the availability conditions were met, an offer to be driven by the system was displayed in the visual displays of the vehicle (i.e., instrument cluster and centre display) along with an auditory alert. To activate or deactivate the system, the driver was required to press two buttons on the left and right sides of the steering wheel for a prolonged time until the system activated or deactivated. During automated driving, the instrument cluster displayed graphics depicting the vehicle's current intent. When the system's availability conditions were no longer met, participants were instructed to resume manual driving, by first receiving a tug from the seatbelt, and then auditory and visual cues about what they were supposed to do. The different sequences and states of each of the developed GUI are presented in comparison in **Figure 4.6**.



Figure 4.6: Visual and auditory description of GUI sequence and states for each of the DAS and comparison of concepts GUI A and GUI B.

Procedure

The data collection strategy for this study involved two stages: the first was an on-road driving session, and the second was a semi-structured interview.

Introduction. Before the driving session, the participants were given a quick introduction to the structure of the study, after which they were led to the test vehicle, where they were given a brief introduction to the two DAS and how to interact with them.

The information that drivers were provided with in advance of the driving session is outlined in **Table 4.2**, which also gives an overview of the two systems. The introduction to the systems was carried out within the vehicle itself in a manner that was analogous to the orientation that one would receive at a car dealership upon picking up a brand-new vehicle. In order to create a driving experience that was as realistic as possible, the decision was made to give the drivers only a relatively small amount of information. Following the introduction, the participants were given some time to familiarise themselves with the environment and get comfortable in it. It is important to note that the drivers were unable to observe the configuration of the driving or GUI wizard in the back seat because all of the necessary equipment was hidden from view.

System	Description	ODD	Limitations	Interaction
Level 2	-Supervised driving	-Always available	-Clear view of lane	- Activation/deactivation
	automation		markings	via steering wheel
	-Maintains speed		- Driver responsible at	button
	-Adjusts speed distance to		all times	
	vehicle in front			
	-Lane keeping assistance			
	- Unsupervised driving	- Partially	- Clear view of lane	- Activation/deactivation
	automation	controlled access	markings	via a long press of the
Level 3	- Maintains speed	- Speeds up to		steering wheel button
		80kph		

 Table 4.2: Description of DAS capabilities and limitations and information participants received prior to the driving session in Study IV.

On-Road Observations. The driving session took approximately 60 minutes and was conducted on a partially controlled access city highway with speeds up to 80 kph. The route led from the Volvo Cars Torslanda office to a southern section of Gothenburg (Slingan South) where a brief stop was made to swap interfaces. At this juncture, drivers were asked to exit the vehicle with the test leader and fill out a questionnaire based on the same framework used to create GUI B.

This questionnaire was designed to assess the driver's understanding of the two systems experienced. After the switch, the driver returned to the car and drove the same route back to the beginning of the driving session (Slingan North), where they were asked to fill out the same questionnaire regarding the second drive and GUI B.

Figure 4.7 depicts the route taken, including the beginning, and ending points and the point at which the GUI was changed. In addition, it emphasises the predefined stretches where Level 3 driving automation was available, as well as the duration of their availability, which resulted in approximately eight minutes of automated driving in each direction.

Post-Driving Interviews. After returning to the starting location, the test leader and participant would enter the Volvo Cars office and retreat to a separate room for the interview. The purpose of the roughly 30-minute semi-structured interview was primarily to gain insight into how the driver perceived the two distinct DAS, their capabilities, and their limitations. In addition, a comparison of the two GUIs experienced was made by presenting screenshots of the sequences and states (as seen in **Table 4.6**), and the driver was asked to elaborate on what aided them in understanding the systems and what aspects were unclear.

Participants

A total of 16 people, seven of whom were female and nine of whom were male, ranging in age from 23 to 70 years old (mean = 44, and SD = 13.48), were recruited and compensated by means of a recruitment agency. The recruiter was provided with a screener as well as criteria for excluding candidates, and all of the participants met the following criteria: (i) have a valid driver's licence, (ii) drive a vehicle that has an automatic gearbox, and (iii) possess a car with adaptive cruise control (which is considered Level 1 Driver Assistance



Figure 4.7: Route for driving session highlighting stretches for Level 3 DAS availability and exposure time.

by SAE). Every single participant was a seasoned driver who routinely drove to and from their job in their own vehicle. Nobody who took part in the study worked in the automotive industry or for a company that was involved in vehicle research and development.

Findings

The primary focus of this study is to examine the driver's perceptions and understanding of DAS, particularly in relation to their engagement with a multi-level automation system.

Overall, participants agreed that the implementation of automated features provides enhanced assistance and facilitates a more seamless driving experience, hence contributing to increased safety. For example, the inclusion of indicators, such as blind spot warnings, is regarded as useful. Still, participants agreed it would be beneficial to incorporate information about the surroundings and other road users inside the visual displays. However, the participants believed that the DAS possesses a level of intelligence and can be effectively employed in instances of long-distance driving, but also saw potential benefits for drivers experiencing stress, particularly in congested traffic scenarios, due to the belief that the system has situational awareness. Nevertheless, while the DAS was perceived to be smart and aware of its surroundings, participants were divided about the vehicle's ability to consistently maintain its performance, e.g., speed control or the detection of red lights. Thus, the statements conveyed a degree of uncertainty about the vehicle's ability to perceive its surroundings, underlining the need for contextual information. It appears that drivers sought contextual information in order to assess the vehicle's capabilities and limitations, and thus in order to decide if it was trustworthy.

However, the drivers also believed that the more they utilize the systems, the more comfortable they will become with them, and referred to their experiences learning to use Adaptive Cruise Control (Level 1 DAS) in their personal vehicles. This was underlined by different situations experienced during the driving session, where participants explained that observing how the vehicle would handle a traffic merge, for example, would give them confidence in the vehicle's abilities and an indication of what traffic scenarios it could handle.

Notably, while participants only experienced the two driving modes in freeflowing traffic, they believed that the DAS was capable of managing slow traffic scenarios. While its applicability in city traffic with a high pedestrian presence remains questionable for most participants, a few believed the system was able to adapt to complex traffic scenarios and city traffic.

Upon further inquiry, it was revealed that this idea was based on preconceptions obtained from media coverage and automobile magazine reporting on DAS that were already on the market.

This indicates that the participants did not seem to distinguish between the different brands and even though they were experiencing a prototype that had little in common with the systems on the market, the technology was perceived to be the same.

This confusion about the technology was also observed in their lack of mode awareness within the driving modes they experienced. The drivers frequently experienced confusion regarding the active driving mode, expressing ambiguity about the distinction between the two driving modes, despite receiving an introduction to the differences prior to the driving session. This also revealed that, no matter the active driving mode, the participants always felt responsible for the driving task. Further inquiry showed that they also felt responsible when the Level 3 DAS was active even though, according to the taxonomies and the introduction by the test leader, they were allowed to engage in other activities. The prevailing viewpoint was that the driver is always accountable for the actions of the vehicle, and the lack of clarity surrounding legal and insurance issues further reinforced the sense of this.

Nevertheless, the participants enjoyed the experience of handing over the driving task to the vehicle, even though they remained vigilant, throughout all the driving sessions, in observing the driving behaviour. Notably, in this study too the vehicle's driving behaviour was used as an information source in order to understand the vehicle's capabilities. However, a curious effect could be observed. Although the driving behaviour remained consistent, it was observed that the UIs had an impact on the driver's perception of the DAS. The findings from the post-interviews indicate that when GUI B was activated, the DAS was perceived as more adept and smoother in its driving performance than when GUI A was activated. Furthermore, there was an overall preference for GUI B because it offered more information throughout the various driving modes, which indicates that an interplay between the UI and the perceived performance of the DAS influenced the driver's perception of the system. However, a consensus existed about the general user-friendliness of both systems, mostly attributed to their high level of intuitiveness. It is worth noting, however, that some initial struggles were observed, particularly in relation to the activation and deactivation processes of the Level 3 DAS. The participants further appreciated the auditory and haptic feedback of both UIs by announcing their availability or the tug of the seatbelt when their attention was required. However, the use of symbols and text was seen as more helpful in their understanding of the system's intention, what was expected of them, and how they could interact with the system, even though at times the drivers would have liked more input. The participants recognised that the absence of essential feedback from the vehicle hinders their sense of complete control in evaluating situations. Additionally, they emphasized the importance

of explicit communication between the driver and the vehicle in order to understand the vehicle's intentions, e.g., the availability status of the DAS and reasons why. This was seen as another indicator of the participants' reluctance to relinquish control over the driving task to the vehicle and engage in other tasks.

Summary of Insights

- ◊ Also in Study IV, the driving context and thus the question "When can I use the system?" appears essential for the driver's understanding.
 - The driving context includes aspects that are discussed, especially road types and traffic conditions, even though there were no differing conditions experienced.
- ◇ The vehicle's abilities were prominent in understanding many aspects of the interaction, thus the question "What does the vehicle do?" emerged here too as another central guide for drivers.
 - Vehicle behaviour and how drivers perceived each system's performance influenced how capable they perceived the systems to be but were also strong indicators of the system's limitations.
 - ♦ The less human involvement was required, the smarter and more capable the system was perceived to be.
 - ♦ Capable systems were ascribed situational and predictive capabilities and assigned greater trust.
- Trust was closely connected to the learning experience and was calibrated by observing how the systems handled different driving situations but was also connected to the information received from the vehicle.
- ◇ Previous experiences with other or similar systems were frequently mentioned in order to explain expectations of the DAS capabilities and possible interactions with it.
- ◊ The question "What do I do (now)?" emerged as well, especially when trying to make sense of the different driving modes encountered.
 - ♦ Many drivers struggled to understand the purpose of the two systems, as they were perceived to have the same capabilities.
 - ◊ Drivers overall saw themselves as always responsible for supervising and monitoring the system, no matter the automation level.
- ♦ Preconceptions were relevant with regard to using an unknown system.
 - Information received through other sources than actual usage, including the media, social circles, and legislation influenced the driver's perception of the systems and their willingness to use them.

◊ Legislative status and current social discourse surrounding automated vehicle technologies seemed to inspire greater distrust in the vehicle's capabilities.

4.5 Study V: Co-Creation of a Design-Tool

The fifth study was a participatory design study. The study was structured around a series of workshops in which experts in the field utilised the Design for Perception Toolkit to design a user interface for a DAS. The goal of the study was to utilize an initial version of the toolkit and continue developing it further throughout the participatory design study by means of co-creation and other collaborative feedback methods. The detailed process and steps taken during the workshops and co-creation activities are described in Paper H.

Aim

The aim of this study was to apply the knowledge gathered and represented in the Design for Perception Toolkit in a use case study with the goal to gather insights for the development of the toolkit, which was subsequently done through a co-creation approach with the participants in the study.

Method

This study was structured around a series of workshops using the Design for Perception Toolkit to facilitate a participatory design methodology. Throughout the duration of the study, a co-design approach was utilised to further develop the toolkit in collaboration with the practitioners who were the target demographic. At the conclusion of the study, interviews were conducted to gather information regarding the utilization and the strengths and weaknesses of the toolkit. The subsequent sections describe the methodology employed in this study and conclude with a summary of the interview findings.

Co-Design Activity

Using an initial version of the Design for Perception Toolkit, the purpose of the workshops was to develop an interface for driving automation systems. When designing the interface, the toolkit provided a systematic framework that prompted participants to consider crucial factors and aspects regarding the driver's understanding of such systems.

In order to contribute to the design of the toolkit, the workshops provided a setting of collaboration in which participants could share their experiences with and requirements for the design tool. Subsequently, the project utilised a co-design approach to achieve the stated aim.

Post-Interviews

The participants in the use case study were interviewed using a semi-structured approach to gain insight into the design tool's efficacy, limitations, and future enhancements. After the workshops, the participants had complete access to the toolkit and all of its versions and were free to employ it as they saw appropriate. Thus, the interviews determined if the practitioners liked the toolkit and where it could be improved.

The interview topic guide consisted of three sections: (i) the designer's routine responsibilities and obstacles; (ii) tool utilisation and perceived utility; and (iii) the toolkit's strengths and weaknesses. The interview guide was not exhaustive and allowed for additional questions. A colleague of the author conducted the interviews as a neutral third party to reduce bias and ensure that the author's team affiliation did not influence the responses of the experts. The interviews discussed the pros and cons of integrating the toolkit into the existing workflows and other possibilities of usage.

Setup and Participants

Practitioners participated in an iterative use case study during the workshops. Among the practitioners were four interaction designers, one design researcher and a software developer, as well as the author facilitating the workshops, and one user researcher supporting and conducting the subsequent interviews. All participants worked in the field of driving automation, and as such, they are representative of the knowledgeable users who are the design toolkit's target audience. The purpose of the use case was to create a user interface for a vehicle with multiple driving modes and increase the driver's mode awareness. However, the designers had to refrain from using any currently available concepts under development and were instructed to construct a new user interface (UI) from scratch. The purpose of the toolkit was to encourage exploration and innovation, thereby facilitating the inclusion of features and functionalities that could potentially improve the user experience. In addition, they were given the latitude to experiment with various design elements, such as colour schemes and layout options, to create a visually appealing user interface – without any design guidelines other than those provided by the Design for Perception Toolkit. While the author facilitated the use case, the workshops and subsequent feedback sessions and co-design activities concerning the toolkit, she did not involve herself in the design of the two GUI concepts, but merely observed the working sessions.

Process

This project's application of participatory design research can be broken down into several phases, with each phase delineating specific activities that occurred during that phase. This project's phases were guided by Spinuzzi's [129] participatory design methodology and are referred to as (i) work exploration, (ii) the discovery process, (iii) prototyping, and (iv) evaluation, respectively. These phases can occur sequentially, concurrently, or iteratively, making them flexible and adaptable to the requirements of the project.

Figure 4.8 provides an illustration of the process that was applied during the participatory design study, outlining all activities along with the participants. It also demonstrates the iterative and simultaneous character of the process's activities.



Figure 4.8: Participatory design research process indicating activities and participants during the project.

The design of the use case and the review workshops were used to gain insight into the concept development, with the workshops also serving as a forum for the team to discuss the positive and negative aspects of the toolkit with the researchers. Notably, design review sessions are a common way for a practitioner to receive feedback from team members and stakeholders during a project, which is why this configuration was chosen. This process enabled the team to revise the toolkit based on the received feedback. Involving the team in closed-loop review sessions ensured that the toolkit was effectively utilized and that any issues or areas for improvement were promptly addressed.

The workshops provided a forum for team members and stakeholders to discuss their ideas and suggestions in an open manner, fostering a collaborative environment for progress. In addition, the co-creation activities allowed for hands-on exploration of the toolkit, ensuring that it met the precise needs and requirements of the project, which were later assessed through interviews with the participants.

Findings

The experts, who participated in the use case study and subsequent interviews, concurred that the toolkit is a useful instrument for validating interfaces and comprehending user perspectives. It can be used individually or in a group setting, streamlining the process of generating ideas and evaluating subsequent proposals. It was agreed that the tool may elicit questions that would not ordinarily be considered and could be useful in the early design phases, whereas when used in a group it can generate more conversations and a variety of perspectives. It facilitates the screening and interpretation of the initial round of ideation, the determination of the results of the initial conceptualization phase, and a structured approach to specific tasks and design solutions. However, it was also noted that implementing a structured group discussion requires time and preparation. In addition, the tool addresses the fundamentals of what must be considered, making it effective for presenting concepts during a final review. In such settings, the tool was regarded as a useful backup for decision support and can demonstrate traceability, because it provides the possibility to trace design decisions and, when necessary, substantiate them using the underlying data.

Thus, using the Excel variant of the toolkit as a checklist at the conclusion of

a design iteration can ensure thoroughness and identify overlooked opportunities, according to the experts. Furthermore, the Excel variant was compared to heuristic checklist aiming to reduce errors and omissions during the design phase. It was mentioned that using heuristics can also foster creativity and promote alternative thinking, thereby facilitating the identification of overlooked opportunities or mistakes. The Card Deck variant on the other hand, was seen as more of a tool that offers support during phases of ideation and helps to review design solutions in groups, leaving more room to discuss and engage, in a structured and guided manner through the supporting questions.

Based on the responses to the interview questions, it can be inferred that the representation variant of the tool appears to influence the stage of the design process where each variant is deemed useful. Furthermore, the application area was closely linked to the tool and its variants, as different methods were employed to make use of it in diverse tasks. The discussion among the experts centred on the function of the toolkit in providing a structured framework for evaluating and reviewing design concepts and ideas. Due to the nature of the applied processes, the significance of being able to relate the contents of the toolkit to each other at different stages of the design process and how these activities are interconnected was emphasised. Thus, the tool provided a methodical approach to comprehending and addressing complex design challenges, as well as tracking these activities throughout the duration of a project. In summary, designers saw the toolkit's as being useful especially in three application areas: (i) Explore: this area addresses the initial phases of a project when trying to scope solutions and identify gaps; (ii) Design: this area focuses on utilising the toolkit to ideate new ideas, and at the conclusion of design iterations and during design reviews; and (iii) Test: this area of application focuses on using the toolkit as a validation and risk assessment tool.



Figure 4.9: Illustration of the toolkit's application during different activities during the design process.

Figure 4.9 depicts the implementation of the toolkit's variants during a design process based on the triple diamond method [130], according to the findings of the interview. It demonstrates that there were three distinct application areas (Explore, Design, and Test) and that two toolkit variants were utilised at various phases of the process.

It is notable that, despite the fact that the toolkit's content remained unchanged irrespective of the representation, the designer's utilisation preferences varied depending on whether the data was presented as an Excel Sheet or a Card Deck. **Table 4.3** summarises the various scenarios and tasks for which each variant was deemed useful.

	· ·						
	Explore		Design		Test		
	Exploration of Solutions	Identification of Gaps	Ideation	Concept Development	Decision Tracking	Review Activities	Validation and Evaluation
Excel Sheet	х	х			х	х	х
Card Deck		х	х	х		х	

Table 4.3: Overview of application areas for the different variants (Excel Sheet vs Card
Deck) of the tool.

In conclusion, the toolkit was seen as valuable for designers in the automotive industry, as it provides insights into driver's needs when interacting with a DAS. Further, the tools foundation in empirical data was considered to give it credibility. In addition, it provided a structured framework for collaboration and communication among team members, thereby potentially saving time and aligning different perspectives. During the design process, the concise layout of the toolkit facilitated easy reference and swift decision-making. However, the large amount of information that the toolkit provides was seen as an obstacle for new users, since thorough reviews necessitate a considerable investment of time and effort.

Summary of Insights

- ♦ The toolkit can be used as ideation and design support (Card Deck variant), but also as a heuristic's checklist (Excel variant).
- ◊ The toolkit offers a structured approach for design reviews and decision making based on empirical data.
- ♦ The toolkit helps establish a common baseline between stakeholders and enables designers to understand the driver's needs and factors influencing their use of DAS.
- ◇ A limitation of the toolkit is the large amount of information that is provided to the user, which creates a steep learning curve for new users of the toolkit, and the application requires a commitment of time by the teams.
- ◊ Despite this shortcoming, the toolkit's concise format enables improved communication among stakeholders and facilitates quick decision-making and the traceability of design decisions.

Chapter 5 Synthesis



CHAPTER 5

Synthesis

This chapter presents the synthesised results from the conducted research studies. It presents the findings from the cross-study analysis, answering RQ1 by presenting a unified descriptive model illustrating the process of how perception shapes understanding, and discussing the implications for the design of driving automation systems. It concludes with answering RQ2 by presenting the developed Design for Perception toolkit and its two variants.

5.1 Factors Impacting the Driver's Perception and Consequent Understanding

The results of the cross-study analysis suggest interdependencies between the driver's understanding and their perception of the driving automation in their vehicle. This process is presented as a descriptive model of how perception shapes the understanding of a DAS (Figure 5.1). In the context of the presented work, the term 'perception' refers to the cognitive process by which individuals assess and interpret information they receive from their surround-ings; and the term 'understanding' refers to the capacity to construct a mental

representation that facilitates the interaction with a DAS.

The presented model illustrates that, irrespective of the degree of automation, users of DAS understand the systems by reference to three distinct components: the Context, the Vehicle, and the Driver. Further, the cross-study analysis revealed eleven recurring characteristics that constitute the driver's understanding of an automated system. The various aspects and connected sub-aspects have been found to collectively form the elements that contribute to a driver's understanding, or mental model, of a DAS. Moreover, a total of six factors that exert an influence on the driver's perception were found, ultimately impacting and shaping the driver's understanding. These factors were further categorised as perceptual sets and sensory information, which subsequently modify and/or impact the driver's mental representation of the DAS. The illustrated process cycle in **Figure 5.1** can be characterised as



Figure 5.1: Descriptive model of the process of how the driver's perception shapes their understanding of DAS.

continuous – a feedback loop that integrates the information received through a top-down and bottom-up process, as well as a feedback loop updating the driver's mental representation of the DAS. The process can be conceptually divided into three distinct parts: (i) the mental representation, (ii) the perception during engagement with a DAS, and (iii) the dynamic process through which perception shapes understanding.

Mental Representation. This element of the process entails the driver's understanding of the system's characteristics and interaction strategies. It consists of all the aspects and components that comprise the driver's understanding and can be considered the starting point for all interactions. This part of the model represents a static image of the driver's mental representation of the driving automation. This image remains unchanged unless new information is received about the system.

Perception During Use. This block is a mirror of the driver's mental representation of the system. This part of the model demonstrates how the interaction is affected by the perceptual sets and sensory information they receive while driving, operating the vehicle in real-time, and during interaction with a DAS, or when the driver is presented with information about a DAS (e.g., reading about it, or talking to someone about a DAS). Here, the driver's existing knowledge and received information about the DAS is synthesised into a mental representation during use, enabling the driver to assess an interaction or plan and execute an action.

Shaping Understanding. This part of the process is a feedback loop that connects the driver's perception of the system during use with their mental representation of the system. The perception of the system during use is continuously evaluated and the results of that interaction and evaluation have the capacity to change the driver's mental representation based on what they perceive while using the DAS. This can lead to the reinforcement of existing knowledge about a DAS, or when encountering new information, the evaluation of such information and subsequent revision of the driver's understanding of the DAS.

Furthermore, the results from the cross-study analysis suggest that the driver's understanding of a DAS is structured in layers (Context, Vehicle, Driver) that are in continuous interaction (illustrated by the arrows in the model). This means that, for example, the driver's understanding of when they can use the system, and what the vehicle does, will affect their assessment of what they are supposed to do. Likewise, if the driver believes they are allowed to let go of control over the driving task, this will influence their expectations of what the vehicle will do. Each aspect contributes to the overall understanding of how the system operates and how it affects the driver's role. Thus, when a driver is asked to explain how the DAS in their vehicle functions, they will recall the knowledge organised in their mental representation. Similarly, in the moment of using the system, the driver will ask themselves: "When can I use the system?", "What does the vehicle do?" and "What do I do (now)?". During use however, their mental representation of the DAS is accessed and facilitates their interaction with the encountered system. As long as the system works as expected by the driver, their mental representation will be confirmed by what is perceived during use. However, should the driver encounter a new scenario, this may prompt them to reassess their assumptions about the system and update their mental representation and all the connected aspects.

The subsequent sections will elaborate on the aspects of the driver's understanding and perception of DAS, as depicted in **Figure 5.1**.

The Driver's Understanding

The findings indicate that regardless of the level of automation, drivers of such systems discussed driving modes by reference to three distinct elements: the Context, the Vehicle, and the Driver. In addition, the thematic analysis identified eleven recurring aspects: Driving Context, Personal Condition, Vehicle Operations, Comfort, Safety, Abilities, Limitations, Driver Tasks, Attentional Demand, Engagement in Other Tasks, and Authority. According to the findings, the identified aspects, along with their sub-aspects, constitute the driver's knowledge of a DAS. The proposed classification suggests that the driver's understanding consists of a layered structure in which the different elements (Context, Vehicle and Driver) interact (see also Paper E).

Context

On the highest level of the structure is the Context, which describes when and where the DAS can be used, including aspects traffic conditions and road types, but also the driver's personal constitution. In order to make sense of the automation's availability, the driver's understanding is guided by the question "When can I use the system?" and will typically be answered by for example: "Since we didn't use it in city traffic or so, I can't speak about it, but it works on bigger roads and [...] free-flowing traffic."; (P08¹, S4²), or when referring to their personal condition: "[...] if I drive early in the morning, so maybe not fully alert or there is some...you have a meeting you have to fall into or I mean if they are like, you know that there are distractions around you, then I put in the pilot assist functionality as sort of an extra safety." (P02, SII). Descriptions of each aspect in the Context layer and its associated sub-aspects are presented in **Table 5.1**.

	14510 0.11 001	Description	i or appeers and sub-appeers.	
Element	Aspect	Sub-aspect	Description	Identified in Study
Context	Driving Context Different driving situations when the driver can use the system	Traffic Conditions	The traffic conditions needed for the system to be operational, e.g., density or speed of traffic	I, II, III, IV
		Road Types	The road types that the system can operate on, e.g., freeways or urban streets	-
		Weather Conditions	The weather conditions under which the system is operational, e.g., sunny, dry, snow, rain, slippery surface	_
When can I use the system?		Time of Day	Time of day the system is operational, e.g., daylight, night-time	-
39310111		Trip Type	Trip types on which the system is used, e.g., long or short trip, commute to work, leisure activities, traveling	
	Personal Condition	Tired	The physical and mental shape the	I, II, III
	The driver's physical or mental state at a given time	Bored	driver is in, e.g., tired, less attentive, bored, or in a state of stress	
		Stressed		

Table 5.1: Context. Description of aspects and sub-aspects.

 $^2\mathrm{S}{=}$ Study; in this case Study IV

 $^{^{1}}P = Participant;$ in this case 08

Vehicle

The next layer of the structure is represented by the Vehicle or the DAS. The driver's understanding here is guided by the underlying question "What does the vehicle do?" or "What is the vehicle capable of?" and represents the vehicle operations like performing the driving task to different degrees, the comfort and safety it provides, the limitations of the system, as well as any underlying abilities that the driver assumes the vehicle has, such as situational awareness.

Typical impressions that drivers have when describing vehicle operations and abilities are, for example: "[...] it does all the functions for acceleration, deceleration and including steering"; (P05, SIII) or "[...] it would actually notice the car and slow down" (P04, SIV). When talking about comfort, statements like the following were identified: "It's relaxing, because I don't need to take care of certain rather annoying parts, like keeping a safe distance to the car in front of me, and so on. It definitely helps, it takes away certain...maybe no responsibility, but a certain burden." (P12, SII). Table 5.2 provides descriptions of each aspect in the Vehicle layer and its associated sub-aspects.

Element	Aspect	Sub-aspect	Description	Identified in Study	
	Vehicle Operations	Maintaining the Speed	Driving tasks performed by the systems, e.g., accelerate, brake, or	II, III, IV	
	The parts of the driving task the vehicle	Keeping a safe distance from other road users	steer		
	performs	Keeping within the lane			
	Comfort	Physical and	The operations the vehicle performs	II, III, IV	
	The ways in	Mental Relief	that support the driver, e.g.,		
	which the vehicle supports the driver	Stress Relief			
	Safety	Extra Set of Eyes	The enhanced safety the vehicle	II, III, IV	
VEHICLE What does the vehicle do?	The ways in which the vehicle	Smoother Driving Style	offers, e.g., seeing when I am distracted, less aggressive driving by following the traffic flow		
	contributes to a safer driving experience				
	Abilities	Situational	The perceived capabilities of the	II, III, IV	
	The underlying capabilities the driver assumes	Awareness	vehicle to perform the driving task, e.g., understanding traffic situations, seeing other road users		
	that the vehicle has	Predictive Capabilities	The ability to predict traffic development and the actions of other road users, e.g., other road users might pull in/out in front of the vehicle		
		Environment	Reading traffic and road signs, lanes	-	
		Awareness			
	Limitations		The functional limitations of the	II, III, IV	
	The activities a vehicle is not able to perform		system, e.g., not being able to switch lanes, drive in city traffic, or read traffic signs		

Table 5.2: Vehicle. Description of aspects and sub-aspects.

Driver

The last layer of the structure represents the Driver and their responsibilities, thus the underlying guiding question is "What can I do?" or "What should I do?". This layer represents aspects that the driver associates with their responsibilities during driving, e.g., who is in charge of the driving task and which ones, how much attention they need to pay to the driving as well as more fundamental questions regarding the interaction with the displays and controls.

Element	Aspect	Sub-aspect	Description	Identified in Study
	Driver Tasks The tasks the	Interaction with Displays and Controls	The interaction needed from the driver to operate the system, e.g., how to activate the system or manipulate the	II, III, IV
	driver can or needs to perform		interface	_
		Operation of Vehicle	What the driver needs to do to operate the vehicle, e.g., accelerate, brake, and steer	
	Attentional Demand	Supervise the System	The attention needed from the driver for different driving modes, and the	II, III, IV
DRIVER	The amount of attention the driver must pay to the driving activities	Take-Over Ready	 Information required from the vehicle 	
What can I do?		No Attention	-	
	Engagement with	Relaxation	The possibility to engage with other	II, III, IV
	other Tasks What the driver can do when not driving	Productivity	movies, emails, chatting with	
		Socialising	passengers	
	Authority	Responsibility	The allocation of control and	II, III, IV
	The power and	Control	 awareness of driving modes, i.e., who is in charge of the driving task, e.g., 	
	responsibility over the driving task	Mode Awareness	shared control or vehicle taking over control	

 Table 5.3: Driver. Description of aspects and sub-aspects.

Typically, statements regarding the interaction with the displays and controls included: "[...] you put it in self-driving mode with these buttons, and you see the blue line" (P01, SIII), describing the steps they take in the interaction

and feedback they receive. Further, aspects regarding their involvement in the driving task are described as follows in the case of partial automation: "I don't release the steering wheel fully. I still have my hand there, so I still drive the car." (P03, SII), where who has the authority over the driving task is described as "[...] it's kind of ambiguous to me exactly how much responsibility it's going to take" (P17, SIII). Detailed descriptions of each aspect in the Driver layer and its associated sub-aspects are presented in **Table 5.3**.

The Driver's Perception

The cross-study analysis of the data from the four studies (I-IV) identified six factors that influence how drivers perceive driving automation during usage. The six factors are Preconceptions, Previous Experiences, Perceived Safety, Trust, Vehicle Behaviour and Information Sources, which have been further split into different aspects. These factors together with their respective aspects describe how a driver perceives driving automation in the moment of use, but also aspects influencing the driver's perception prior to use, which in turn influences their understanding and therefore usage strategies. The driver develops from this a mental representation of the system, describing when it can be used, what tasks it will take over and what responsibilities the driver has when engaging with the system. The identified factors have been categorised further into two groups: Perceptual Sets and Sensory Information.

Perceptual Sets

Preconceptions, Previous Experiences, Perceived Safety and Trust belong to the category of top-down processing factors, i.e., Perceptual Sets, as they comprise contextual information. The perceptual set is the tendency to perceive objects or situations from a specific frame of reference. Existing schemas, mental representations, and concepts frequently serve as a guide for perceptual sets. Top-down processing begins with the most general perceptions and progresses to the more specific ones. Such perceptions are significantly influenced by prior knowledge and expectations, such as schemas and mental models. In the case of a driver interacting with an automated driving system, the driver's perception will be influenced by preconceived notions, previous interactions with other DAS (not exclusive to a specific level of automation), perceived safety, and trust in the system's capabilities. Thus, the driver's perception of the system is influenced by what they expect to perceive.

Preconceptions

Preconceived ideas or expectations about the DAS are organised under Preconceptions. This factor influences the driver's understanding significantly, as it typically harbours the driver's ideas about the system and their anticipations of it, be they current ones or those concerning its unrealised potential. This includes the driver's notion of the system's purpose and their relationship to it: "It does some driving for you, [...] makes sure you don't get a ticket and go above speed limit. This is more assisting you." (P07, SIII), as well as anticipations about the system and the benefits of using it. This can include anticipated usefulness and/or expected gains prior to experiencing the system, as described by one participant: "I think every car should have it. If you could install it in an older car that doesn't have the system, that would be something for traffic security, for safety. Because I do think it will help you to drive smoother, keep you a little bit more comfortable." (P11, SIV). Expectations will influence how drivers interact with the system; however, these expectations cannot always be met, which consequently led to one participant not using the system: "I had higher expectations than I could receive from the system, so to say. But on the other hand, I also knew about it, kind of. But I still wanted to test it. I just thought 'OK, I will not use it'." (P08, SII). Table 5.4 describes the aspects categorised under Preconceptions in detail.

Previous Experiences

Previous experiences include the driver's experienced situations and learning process with the current system, but also those involving previously encountered DAS. This factor also incorporates knowledge obtained through social discourse, e.g., topics debated in the news and on social media, marketing campaigns, or discussions in social circles. Their experience of usage with any automated system will influence the driver's usage of the currently experienced DAS, and this experience is used as a means to understand encountered systems, like a participant comparing the system in their personal car to their experience of the Level 2 system in Study III: "The adaptive cruise control was very, very different from regular cruise control, a little bit more...it's a more complicated version of cruise control" (P15, SIII). Besides comparison to other experiences, the learning experience and the way the driver learned

Factor	Aspect	Sub-aspect	Description	ldentified in Study
	Purpose of the System	Assisting the Driver,	The driver's understanding of	II, III, IV
	The extent to which the system is supposed to support the driver with executing the driving task	Collaboration	to assist them, e.g., by	
		Take over the Driving Task	assisting or taking over parts of the driving task	
	Capabilities The capabilities the drivers expect the system to be able to execute	Situational Awareness	The impression that the system is capable of	II, III, IV
		Predictive Capabilities	executing complex driving tasks, e.g., seeing other road	
		Environmental	predicting the behaviour of other road users, reading traffic signs and lights	
Preconceptions Mental		Awareness, e.g., reading traffic signs		
representation of the vehicle's capabilities, based on	Anticipation Excitement and aspirations about an	Social	Technology development and future values are discussed within social circles and also - based on media information or information from regulatory authorities, as well as the impact of sudden AD-related accidents in the media and/or hopes for development.	II, III, IV
expectations	e.g., capabilities and future development	Technology		
	Anticipated Usefulness	Driving Support	The ability of the vehicle to	II, III, IV
	The benefits the driver expects to gain from using the system	Enhanced Safety	take over parts or all of the driving task, thereby increasing safety and the possibility to perform secondary tasks, and	
		Free Time		
		Stress Relief	and the effect on their personal condition	

Table 5.4: Perceptual Sets (Top-Down Processing Factors): Preconceptions.

to use the system have also been shown to make a difference to learners from drivers trying on their own to drivers who learned under supervision: "I had a colleague who knew a lot. So, it was learning by doing with some support." (P08, SII). This enabled the driver to feel confident in using the system. Further, discussions in the media and social circles influence how drivers perceive and talk about the systems, even if they have no prior experience with them. These information sources tend to inform the drivers in a way that might misinform them. For example, a participant in Study IV believed they knew details about a system which was, in fact, a prototype built solely for the study: "I'm really interested to drive the second system in the city because I've read about it, and I think it works good even in the city." (P02, SIV). Further details for each aspect are found in **Table 5.5**.

Factor	Aspect	Sub-aspect	Description	Identified in Study
	Experience of Usage The skill or knowledge	Prior Usage of Similar	Driver's comparison to a	II, III, IV
		Systems	system they have used before, and key events that	
Previous	prior systems or the current one, resulting in the overall experience	Positive and/or Negative Experiences	influence the driver's perception of the system	
Experiences	Learning Experience How the driver learned to use the system	Trial and Error	The ways that drivers learn about or use the system, e.g., by reading manuals or tutorials, or through supervision and guidance	II, III, IV
Experienced				
learning processes and		Reading the Manual		
their influence		Under Supervision/		
model		Guided Learning		
	Social Discourse Everything said or written in society about the topic	Media	Written, verbal, or other representative communication about automated vehicles in the media, or social circles, e.g., marketing campaigns, news reports, discussions with other people	III, IV
		Social Circles		

Table 5.5: Perceptual Sets (Top-Down Processing Factors): Previous Experiences.

Perceived Safety

The factor Perceived Safety incorporates a range of aspects describing the driver's subjective assessment of how safe it is to use the system. Aspects like the system's performance and its consistent and predictable behaviour influence whether the driver assesses the system to be safe, but also their awareness of the system's technological and legislative situation influences the driver's perception, and their willingness to risk usage. In many cases, participants' awareness of the legislative status influenced their perception of who would be liable when driving with a fully automated system, which in turn influenced their perception of safety and the usefulness of the system: "I need to stay focused. I think it's a false sense of security. And when it comes to if

it becomes a court case and I run over someone and the car was steering, there have been many such cases. Then, of course, it's my car. I've chosen to buy it with that system, and I chose to activate it. [...] So, I don't see this system as helpful or yeah, legally speaking." (P06, SIV). Further, the impression of the system being intelligent and able to execute tasks without human intervention seemed to be associated with the safety perception, as in this example, where the driver stated that the car had "[...] good predictive capacity on how to engage with other vehicles" (P03, SIII). Generally, it seemed that the smarter the system appeared to the driver, the higher the perceived safety was. A detailed description of Perceived Safety and its assigned aspects can be seen in **Table 5.6**.

Factor	Aspect	Sub-aspect	Description	Identified in Study
	Awareness The awareness a driver has about the technology, the years before it was approved by the authorities, and its current status	Awareness of Technology	Knowledge about the system, its capabilities, and limitations, as well as historical knowledge about	II, III, IV
Perceived		Awareness of Legislative Status	its development and current level of safety and the standpoint of the authorities	
Safety	System Performance The execution of system functionality and how the driver perceives the vehicle's reliability Anthropomorphism Humanisation of vehicles by drivers through ascribing human characteristics to the system and empathising	Predictability	The possibility for the driver to successfully foresee the result of an interaction with or action of the system, and consistent system behaviour, leading to the driver perceiving the vehicle as reliable The technological capabilities to act on behalf of humans without direct human intervention and control. Agency to execute tasks, and perception of the vehicle being intelligent enough to execute complex tasks	II, III, IV
risk assessment the driver		Consistency		
vehicle's behaviour and canabilities		Transparency		
		Intelligence		II, III, IV
		Autonomy		

Trust

Under Trust, aspects that describe trust at different levels of abstraction are found. On a basic level, a driver's appreciation of a brand will influence

whether, even before engaging with it, the driver will trust the system. Although, drivers will also choose when to trust the system and to what extent. Hence, while a brand's reputation can help drivers overcome initial hesitations about trying DAS, "I trust the car. [...] I mean, I know [brand] is very good like it's a very reliable car. So, yeah, it has a very strong reputation." (P13, SIII), the context of use will guide users to build trust in the system in different situations. For example, participants make distinctions between traffic and weather conditions when deciding in which situations to trust the system: "Uh, the clearer the road is, the better the road is, the less traffic it is, the more you can trust the system. But, as soon as you get something in front of you, something where you need to pass a car or whatever it is, the less you can trust the system." (P4, SII). These assessments help drivers to calibrate their trust level and therefore, hopefully, not over-trust and misuse the systems, like a participant who did not see any limitations for the use of the system: "As long as you trust the car, I think you can use it almost anywhere." (P09, SIV), which could potentially end fatally. A detailed list of all factors and connected aspects of Trust can be found in Table 5.7.

Aspect	Sub-aspect	Description	Identified in Study
Level of Trust	No Trust	Trust calibration towards the	II, III, IV
The amount of trust, the	Appropriate Trust	 system, ranging from no trust, resulting in disuse, and 	
driver has in the system	Over-Trust	over-trust, resulting in misuse	
Situational Trust	Driving Context	The driving contexts that the	II, III, IV
The contexts in which the driver trusts the system's capabilities. Distinctions between situations when the driver trusts the vehicle	Driving Context	driver deems the system capable of handling, e.g.,	
	Personal Condition traffic conditions, and physical or mental star which the driver is comfortable using the	traffic conditions, and the	
		which the driver is	
		comfortable using the system	
Brand Perception		The sum of the driver's	III, IV
The culmination of all a customer's thoughts and feelings about the brand, and its products and services		experiences and anticipations expectations about what a brand represents	
	Aspect Level of Trust The amount of trust, the driver has in the system Situational Trust The contexts in which the driver trusts the system's capabilities. Distinctions between situations when the driver trusts the vehicle Brand Perception The culmination of all a customer's thoughts and feelings about the brand, and its products and services	AspectSub-aspectLevel of TrustNo TrustThe amount of trust, the driver has in the systemAppropriate TrustSituational TrustOver-TrustThe contexts in which the driver trusts the system's capabilities. Distinctions between situations when the driver trusts the vehicleDriving ContextBrand PerceptionPersonal ConditionThe culmination of all a customer's thoughts and feelings about the brand, and its products and servicesImage: Sub-aspect	AspectSub-aspectDescriptionLevel of TrustNo TrustTrust calibration towards the system, ranging from no trust, resulting in disuse, and over-trust, resulting in misuseTrust calibration towards the system, ranging from no trust, resulting in disuse, and over-trust, resulting in misuseSituational TrustDriving ContextThe driving contexts that the driver deems the systemSituations I TrustDriving ContextThe driving contexts that the driver deems the system capable of handling, e.g., traffic conditions, and the physical or mental state in which the driver is comfortable using the systemBrand PerceptionThe sum of the driver's experiences and anticipations expectations about what a brand represents

Table 5.7: Perceptual Sets (Top-Down Processing Factors)	: Ti	rust
--	------	------

Sensory Information

The factors Vehicle Behaviour and Information Sources are classified as bottomup processing factors because they pertain to sensory data in terms of environmental stimuli and occur in real-time. The vehicle's behaviour is perceived in real-time via several sensory channels such as visual, auditory and haptic feedback. This means that the driver considers the vehicle's driving behaviour as information from the vehicle to the driver, which will inform their evaluation of the vehicle's capabilities, for example in terms of comfort, reliability, and even the vehicle's ability to communicate.

Vehicle Behaviour

The driver's understanding of the vehicle's capabilities and their inclination to trust it are closely related to the vehicle's driving style and behaviour. If the vehicle's driving style is not in accordance with the expectations of the driver, it can lead to negative experiences, as was the case for one participant who was bothered by the vehicle's placement in the lane when driving through curves: "I think I feel a bit unsafe when I'm in a curve. I don't know if it can handle this curve or not." (P01, SII).

Factor	Aspect	Sub-aspect	Description	ldentified in Study	
Vehicle	Driving Style	Longitudinal Movement	The vehicle's ability to communicate with the driver	II, III, IV	
Behaviour	the driver on a scale ranged from aggressive to cautious, e.g., "drives like me", aggressive,	Placement in Lane	 through its behaviour, capabilities, efficiency, and consideration for other road users through performance feateral live acclustical 		
How the driver perceives the		Distance to Objects			
and driving performance	defensive, etc.	Consideration of Driving Context	frequency of speed shifts, lateral placement, and frequency of change in distance to objects		

Table 5.8: Sensory Information	(Bottom-Up Processing F	Factors): Vehicle Behaviour
--------------------------------	-------------------------	-----------------------------

However, the driving style can also be perceived as considerate and aware by the drivers, for example, in a case of merging traffic: "I think the system was much, much better because when a car came out on the highway, it saw the car $[\ldots]$ it was slowing the speed to let them pass and it also took notice of

this car behind me, because it wasn't braking it just slowed down and made it so smooth. It was like a school example. It was a very good experience." (P05, SIV). This shows that the vehicle's driving style influences the driver's perception of its capabilities but can also foster positive experiences and encourage use. A further description of the included aspects is found in **Table 5.8**.

Information Sources

The factor Information Sources concerns all input that the driver perceives through their senses. This includes multimodal feedback stemming from visual, auditory and tactile sources, but also kinaesthetic aspects like the perceived motion of the vehicle.

Factor	Aspect	Sub-aspect	Description	Identified in Study
Information File Information Sources The information Auc sources The The information Auc sources the The driver uses to and make sense of their own responsibilities The abo vibr term skin Kina The like mov	Visual System	Graphical Elements	Visual feedback that the driver receives from the vehicle, for example, from displays in the form of text, - icons, and graphics, but also information drawn from the environment outside the car	II, III, IV
	The information perceived through the eyes, including elements such as colour, light, proximity, patterns, similarity, and so on	Text Environment		
	Auditory System The loudness, frequency, and meaning of auditory information	Non-Speech Sound	Auditory feedback that the driver receives from the - vehicle, such as warning sounds (ping or beep) or verbal instructions	
		Speech Sound		
	Tactile System The input of messages about pressure, vibration, texture, and temperature through the skin	Steering Wheel	Haptic feedback that the driver receives through their tactile sensory system, like vibrations through the seat or steering wheel, and tug of the seat belt	
		Seat		
		Seat Belt		
	Kinaesthetic System	Seat	The information the driver infers from the vehicle's motion, such as acceleration and deceleration or braking behaviour	
	The feeling of motion, like position, force and movement			
		Vehicle Motion		

Table 5.9: Sensory Information (Bottom-Up Processing Factors): Information Sources.

For example, the driver deduces information about the car's actions through what they perceive from its acceleration and braking behaviour: "I can feel it braking, I can see that it is going faster and slowing down when it needs to." (P02, SIII). However, traditional sources of information such as in-vehicle displays and auditory interfaces are still a primary source of information with regard to understanding a system's status, "When I'm in my car, I can choose the distance, and then if there's a car in front of me, it even displays that car. [...] Green kind of tells me that everything is fine. And then I have the acoustic signal and the hand symbol. And I think there it is even ... that I should put my hands on the steering wheel."; (P11, SII), or what they as a driver are supposed to do. "It was super clear because it was both saying that the system is ready and the text. Two things that make you understand, all right, it's ready." (P09, SIV). Especially in the case of the Information Sources, all aspects work in a multi-source fashion when it comes to helping the driver understand the system's behaviour and capabilities, and it is not possible to consider them in isolation. Table 5.9 provides detailed information about the factor and its specific sub-aspects.
5.2 The Process of how Perception Shapes Understanding

To provide further clarification of how the driver's perception influences their understanding of a DAS, a hypothetical scenario will be introduced, exemplifying how the different aspects shaping the driver's understanding and factors influencing their perception are affected through the use of a DAS and their encounters with different driving situations.

If a person has a system in their personal car that supports the dynamic driving task (DDT) by (i) keeping the vehicle at a set speed, (ii) at a set distance from other moving vehicles, as well as (iii) taking over the steering, as long as there are (iv) visible lane markings. The driver would make sense of the automated system's use by searching for answers to the aforementioned questions that correspond to the aspects of each layer in their understanding. They would do this in order to make sense of the information received from the vehicle or to understand the interaction required. In this scenario, a driver is travelling to work from a residential location, passing through countryside and some highway segments into an urbanised area. When they are driving, they will wonder: "When can I use the system?". In this instance, that would be on well-established roads with good lane markings (Driving Context). The driver might not use the system in the suburbs, but he or she may attempt to use it in the countryside and on the highway. However, the driver could also deduce the wrong specifics, leading them to create a faulty understanding of the DAS and its capabilities and limitations. Considering the same system has certain inherent technical limitations, such as the system performance being unstable when driving on roads with high curvature (Limitations), as are often experienced in the countryside. If the driver further assumes that all that is needed are clear lane markings, they might assume that they can use the DAS on any type of road, while the manufacturer has primarily intended it to be used on highways, where road infrastructure is more controlled and streamlined. This could lead to a situation where the driver experiences the DAS as acting in an unstable way, and may even leave the road, impacting what the driver assumes the benefits of using the system are (Comfort and Safety).

Next, they will ask the question "What does the vehicle do?". The answer is that it maintains the predetermined speed and distance from other moving vehicles, as well as assisting with the steering (Vehicle Operations). Finally, when utilising the system, they might ask "What do I do (now)?" which, depending on the level of the steering assistance, may imply that the driver does not need to steer or control the speed (Driver Tasks), but must supervise the system (Authority). In such cases, the driver must remain attentive and ready to take over control of the vehicle if necessary (Attentional Demand). Assuming that the vehicle automation has another inherent limitation in its technology. While the DAS is able to detect and keep a safe distance to other moving objects, it is not able to do this with objects that are stationary (Limitations). If the driver does not understand this important limitation (which is often only mentioned in a manual), they might wrongly assume that the vehicle will come to a halt when approaching a red light, and that it does not need their intervention. However, since the system is not capable of processing that an object is stationary, it would require the driver in those moments to act by braking and resuming full control over the driving task (Driver Tasks and Authority). Otherwise, this situation could well result in a crash.

This demonstrates that the answers to these questions are reflected through the different aspects and are interrelated with and inform each other. In this way, a comprehensive understanding of the automation while in use is created by the driver.

However, the driver's understanding and consequent use of such systems might not always be the one the designers and developers of such systems intended, which can be explained by introducing the driver's perception into the interplay. For example, the driver's preconceptions about a DAS could influence their perception and understanding while using the DAS in a way that might obscure the intended use of the system. In the case of our hypothetical system, the driver does not need to steer, but still has to pay attention and supervise the system.

To illustrate, the system performs the driving task to a very high degree of satisfaction according to the driver's perception (System Performance). This might lead the driver to perceive the system as more capable than it is and assume that they do not need to pay attention (Purpose of the System). Their willingness to relinquish control and even take up other activities would indicate a high Level of Trust. While previous experiences can influence the driver's expectations as to what the systems are capable of, also other sources like news articles or a chat among friends (Social Discourse) can influence their expectations and consequent use of these systems. These expectations can sometimes be misinformed and lead to negative experiences and unexpected interactions (Learning Experience). For example, the driver might use the DAS in a way that the designers did not intend. These experiences can result in confusion about what the system is doing, which can cause frustration and mistrust, leading the driver to reject the systems (Perceived Safety) – or in the worst case to fatal incidents.

To illustrate further, the hypothetical system is able to keep a set speed and safe distance to other road users. While the technical specification in itself only describes the task the vehicle takes over, the vehicle's behaviour when the driver is using the systems will influence the driver's judgement about its performance. For example, if a system keeps a set distance to other road users and the driver perceives it to accelerate and decelerate too fast in relation to the other vehicles, it may be judged as too aggressive in its Driving Style and thus uncomfortable to use.

Other communication from the system to the driver falls under information sources that are perceived through the in-vehicle user interface(s). This can include visual, auditory, and haptic cues and feedback. For instance, the system might run into limitations and not be able to steer for the driver any longer. It might then send a take-over request (TOR) through a visual and auditory prompt in the in-vehicle interfaces, e.g., displays and speakers. If the driver does not act on the prompt, the vehicle might send a warning by intensifying the signal and adding vibration in the seat, in order to get the driver's attention. All these are ways that the driver can receive information about what the vehicle is doing or what is expected of the driver, and this in turn, will influence their perception of the system's capabilities and limitations.

5.3 Implications for the Design of Driving Automation

Finding an adequate framework that explains the problem area and guides design solutions is a significant challenge for designers, especially in the context of driving automation. In order to enhance system design and the user experience, it is crucial that designers have tools to support their design decisions when developing solutions for DAS. The absence of such support or frameworks can result in solutions that are not understood by users or do not serve their needs. The model presented here seeks to support designers by providing theory; however, in answering the second research question, it became essential to transform the model into a design toolkit to make it applicable to practitioners and their daily challenges, thus addressing a practical need.

The process depicted in the model (Figure 5.1) illustrates that the driver's perception and understanding of DAS are influenced by various factors. For example, the driver's preconceptions about the purpose and capabilities of the system can shape their perception and understanding during usage. Additionally, more abstract ideas, such as previous experiences and social discourse, play a significant role in shaping the driver's expectations and subsequent use of such systems. This has important implications for the design of DAS. Although the levels of automation may be well-defined within the industry, the drivers of vehicles with such capabilities lack an understanding of the associated expectations of them as drivers and their responsibilities (cf. [15][33]).

However, the inherent intricacies underlying the allocation of responsibility for the driving task at all times, and the limitations of vehicles offering such capabilities, fall under the expertise of professionals. Another challenge that drivers face is that, while the taxonomies provide general guidance on technical specifications surrounding the vehicle's capabilities for each level, the implementation of these systems is not standardised, and manufacturers provide different solutions with different feedback and interaction strategies. Further, a widely recognised paradox in the realm of automation pertains to the phenomenon wherein the increasing proficiency of a system in being automated leads to a decrease in the motivation for the human operator to sustain their attention, especially with increased vehicle performance [131]. Therefore, it is crucial for designers to consider the influence of the driver's perception on their understanding of the DAS when designing the system's functionalities and capabilities. Clear communication and education about the system's purpose and limitations can help align the driver's understanding with the intended use of the system, and possibly bridge the gaps between different manufacturers' solutions by taking the driver's perspective.

However, as illustrated by the hypothetical scenario, while there are factors that designers and developers can and must consider during the design of a DAS, there is a range of factors that one cannot directly influence, but nonetheless has to account for. These external factors include unpredictable weather conditions, road infrastructure, and the behaviours of other road users, as well as information that is not published by the manufacturers themselves.

Despite not being directly controllable, designers and developers must anticipate that these variables will influence the driver's interaction with the system – often not in the intended ways.

Hence, for designers and developers of such systems, it is imperative to comprehend the impact of these factors on the driver's perception in order to develop DAS that are in line with the users' mental models and facilitate safe and effective interactions. Thus, the Design for Perception toolkit, which incorporates the knowledge established in the model (**Figure 5.1**), is an important contribution towards a user-centric approach in the design of DAS.

5.4 The Design for Perception Toolkit

The Design for Perception toolkit aims to offer a framework that facilitates the systematic review of potential design solutions and the identification of areas needing improvement. In this context, one can identify requirements for a technical solution from the driver's perspective, as well as specify technical requirements and scenarios that must be addressed through design solutions like the graphical user interface, the vehicle's behaviour, and specific functionalities of the systems. Further, the toolkit addresses a range of usage scenarios that can aid in the assessment and examination of potential solutions. By taking a driver-centric approach to the design of DAS, it enables a thorough examination of user requirements and preferences, enabling the development of a solution that addresses the driver's needs.

The Design for Perception toolkit has two distinct variants, each presenting the same information in different formats, and tailored to correspond to various aims within a project. The contents of the toolkit represent the different components of how the driver's perception shapes their understanding of DAS (**Figure 5.1**). The two components represented in the model are: (i) the aspects of the driver's understanding, including the context, the vehicle, and the driver; and (ii) the driver's perception, including perceptual sets and sensory information. A third part of the toolkit provides guiding questions corresponding to the different aspects and factors and aims to trigger a discussion and critical review of proposed design solutions. The two developed variants can be utilised either independently or in conjunction with one another throughout the entirety of the design phases.



Figure 5.2: QR Code to the Design for Perception repository

The subsequent sections will describe the different variants and components of the toolkit, as well as their respective applications. A repository where the toolkit materials can be downloaded is found under the following link https://doi.org/10.5281/zenodo.10116149 or via the QR code in (Figure 5.2).

The Toolkit's Variants

The Design for Perception toolkit consists of two variants: a Heuristic Checklist in Excel format and a Card Deck. The two variants are presented in the following sections.

Variant 1: Excel - Heuristics Checklist

The Excel variant of the toolkit was identified as specifically useful as a checklist and structured approach to identifying solutions for given technical specifications. The checklist consists of two tables: *Table 01 Understanding*, which supports the development process and provides aspects to consider when designing and identing around solutions; and *Table 02 Perception*, which supports the review of design solutions and aids decision-making and backtracking.

Moreover, the Table allows the designer the option to incorporate the technical specifications of the system that is being developed, that align with the various aspects (e.g., at what speeds the DAS operates, which tasks it will take over), alongside usage scenarios (e.g., describing a driver's activity and context when using the specified functionality) that elucidate the functioning of the system. Additionally, it provides the opportunity to describe design solutions and ways for effectively conveying the associated elements to the driver. **Figure 5.3** shows a section of *Table 01 Understanding* for illustration purposes for how it could be utlized.

The structure of *Table 01 Understanding* is determined by the different elements and aspects that comprise a driver's understanding of a DAS. The goal of this table is to support the exploration and ideation of design solutions by using the driver's mental representation of a DAS as a basis. Thus, the table consists of three parts and their guiding questions: 1. Context – "When can I use the system?"; 2. Vehicle – "What does the car do?"; and 3. Driver – "What do I do?". Each of the sections contains the different aspects and sub-aspects, with descriptions.



Figure 5.3: Table 01 Understanding. Section illustrating the structure of the checklist with example contents for a Level 2 DAS.

The structure of Table 02 – Perception is similar to that of Table 01, and is determined by the different factors that affect the driver's perception of a DAS. The goal of this Table is to review and validate the design solutions in focus. As a result of this, the Table is further divided into the top-down processing factors, i.e., Perceptual Sets, and the bottom-up processing factors, i.e., Sensory Information. Each of the factors and its sub-aspects are described in detail and supported by guiding questions which aim to facilitate a discussion around their corresponding aspects. **Figure 5.4** gives an overview of a section of Table 02 Perception for illustration of how it could be utilized.



Figure 5.4: Table 02 Perception. Section illustrating the structure of the checklist with example contents for a Level 2 DAS.

Additionally, Table 02 introduces a third component called 'Impact', which addresses the design solutions through critical discussion. It does so by introducing, on the one hand, questions that aim to identify the failures and effects of the discussed solution, and on the other hand, by challenging the designer to think of ways to improve the user experience.

Variant 2: Card Deck - Ideation and Validation

The Card Deck variant of the toolkit has been recognised as particularly valuable for facilitating workshops, as well as for individual use in scenarios that require a creative approach, but also when aiming to facilitate a structured review of design solutions within a team. Similar to the Excel variant, the Card Deck is split into three parts: *Deck 01* Understanding, *Deck 02* Perception, and *Deck 03* Impact.

Deck 01 Understanding – Explore and Ideate

The first Card Deck covers the driver's understanding of DAS and facilitates a guided exploration and ideation around possible solutions. **Figure 5.5** shows a selection of cards from Deck 01.



Figure 5.5: Deck 01 Understanding. Selection of cards from the deck.

The cards can be played in any order, and the designer is free to use all of them, or just the cards representing the areas they want to focus on. With this deck, the designer can incorporate the technical specifications of a system that align with the various aspects and ideate around design solutions corresponding to the driver's mental representation or identify areas in need of improvement and create strategies to address these.

Deck 02 Perception – Review and Validate]

This deck supports the review and assessment of design solutions through a guided approach. **Figure 5.6** shows a selection of cards from Deck 02.



Figure 5.6: Deck 02 Perception. Selection of cards from the deck.

With this deck, one can critically review the developed design solutions and discuss the effect of the driver's perception on their understanding and consequent usage of the driving automation. Further, it enables the validation of existing or created solutions from a driver-centric standpoint by challenging the designers and developers to discuss guiding questions corresponding to the factors influencing the driver's perception. The cards provide the flexibility to be utilised in any order, granting the freedom to use all of them or selectively choose those that pertain to the areas of focus.

Deck 03 Impact – Mitigate and Delight

This deck supports the in-depth review of design solutions. With the help of the included cards, questions are introduced which aim to identify the possible failures of a design solution and their effects on the driver in order to mitigate them. In addition, the deck includes cards which aim to challenge design solutions in order to create moments of delight that can have a positive impact on the user experience. **Figure 5.3** shows the cards included in Deck 03.



Figure 5.7: Deck 03 Impact. Selection of cards from the deck.

This deck is special in the sense that its cards can be played on any other card and at any point during the design process. While Deck 02 already offers guiding questions to discuss design solutions critically, this Deck 03 aims to prompt a deeper discussion about the impact of the provided solutions. The impact can be seen as any positive or negative effect on the driver's perception of the system that consequently leads to acceptance and willingness to use a DAS or not.

Chapter 6 Discussion



CHAPTER 6

Discussion

The aim of the present research was to investigate the factors that impact the driver's perception and consequent understanding of driving automation systems (RQ1) and to investigate how this knowledge can be applied to guide design decisions for practitioners involved in the development of driving automation systems (RQ2). This thesis provides both theoretical and practical contributions by addressing these research questions. This chapter presents a comparison between the contribution made through this work and the current body of research, followed by a discussion of the implications for design. The chapter concludes with reflections on the research approach.

6.1 Contributions

The primary objective of the thesis was to address two research questions: (RQ1) What are the factors that impact the driver's perception and subsequent understanding of DAS? and subsequently (RQ2) How can this knowledge be applied to offer design recommendations for practitioners involved in the development of DAS? In order to address the research questions, a mixed-methods methodology was utilised, employing a range of methods, such as surveys, field observations, and interviews. This work makes contributions to both the theoretical and practical aspects of the research area. The subsequent sections will discuss and compare the contributions with previous research, followed by a discussion on the implications for the design of DAS.

Theoretical Contributions

The theoretical contribution of this thesis lies in the identification of the aspects that shape the driver's understanding of DAS, as well as the factors that influence the driver's perception of such systems, which also answers the first research question (Papers A - G). Additionally, the work integrates the discovered aspects constituting understanding, and factors influencing perception, into a unified conceptual model that elucidates the process by which the driver's understanding of a DAS is shaped by their perception.

Consequently, by providing a comprehensive and unified overview, this work addresses a gap in the existing literature: the lack of a holistic understanding of how the driver's perception influences their mental model of DAS.

Earlier research efforts have investigated a range of variables that have been deemed important for the driver's interaction with an automated driving system. These studies have most frequently investigated the topic of safety and take-over requests, trust and complacency, acceptance of and attitude towards automated vehicles, situation awareness, workload and stress, and drowsiness and fatigue, among other factors. For a comprehensive summary, please refer to the literature review conducted by Frison and colleagues [132]. The review emphasises that the existing research on the driver's interaction with DAS often concentrates on a limited range of variables and methodologies, without considering the interplay between variables in the complex environment of the dynamic driving task, or a triangulation of data for a deeper understanding of the driver-automation interaction [132].

However, in order for drivers to understand and interact with DAS in a safe manner, it is crucial for designers to consider the driver's mental model and how their interaction with a DAS influences it. Consequently, numerous authors call for a more human-centric classification of automation levels (cf. [41][133]) than provided by the currently prominent Levels of Driving Automation provided by the Society of Automotive Engineers [7]. While there are advantages to utilising established taxonomies that categorise levels of automation (cf. [19][77]), studies show that the driver's mental models do not align with the technically driven taxonomy (cf. [40][39]). Further, studies show that the SAE taxonomy is not only ambiguous to drivers, but also to practitioners and researchers, who struggle to find a unanimous interpretation of the provided LoA (for an overview, see [134]). Thus, over the years, different approaches have been sought to identify variables relevant to the driver's interaction with DAS.

For example, various works have attempted to describe the driver's interaction with a vehicle through behavioural models. Michon [135] conducted a review of driver behaviour models and identified four different types of models along two dimensions: (i) behavioural models, representing behaviour, vs. (ii) psychological models, representing cognitive processes, vs. (iii) taxonomic models, representing an inventory of facts and their relationships, vs. (iv) functional models, containing components which interact dynamically. Based on his analysis, he remarks on the absence of driver-related factors such as cognitive functions, beliefs and emotions in most models, as they are behavioural-functional and concentrate on specific characteristics of the driving task and driver behaviours. Thus, they do not answer the question as to why the driver behaves in a certain way. He emphasises the need for additional research in the area of cognitive processes to gain a better understanding of driver behaviour and driver motivation [135]. To date, this circumstance has not changed.

Building on Stanton and Young's [136] psychological model of driving automation, Heikoop and colleagues [137] conducted a literature review in order to propose a consensus-based psychological model. Their model aimed to describe the interrelations between identified psychological constructs from the research body. However, their model is solely based on a limited literature search, and only includes nine different variables (i.e., mental model, situation awareness, attention, trust, mental workload, stress, feedback, task demands and fatigue) which represent the consensus in the literature and are critiqued as being highly biased through construct proliferation [137]. Notably, the authors acknowledged the need to extend the model with a range of psychological constructs (e.g., ability, authority, responsibility, amongst others, as well as the identification of further variables), and the empirical investigation of the identified variables and interrelations.

The aforementioned approaches collectively indicate an underlying issue: current research efforts lack a cognitive and holistic approach to the driver's understanding of DAS. In an effort to address this, several studies have examined the difficulties associated with designing DAS and have reached the consensus that numerous difficulties may emerge in the initial phases of development. These challenges are primarily attributed to the adoption of technology-centric taxonomies, which tend to overlook the human driver and prioritise task allocation strategies (cf. [39][40][98]), as well as the lack of variety in methodologies, and no triangulation of data. Moreover, it has been established through additional research that the user's understanding of automation levels does not align with the existing taxonomies [97].

Instead of a single automation level and task-allocation perspective, the research presented in Chapter 5 shows that the driver makes sense of the interaction with a driving automation system by asking themselves the questions 1. "When can I use it?" 2. "What does the vehicle do?", and 3. "What do I do?", each of which additionally contain numerous aspects that they seek answers to, and factors influencing their perception of every interaction with the automated system.

To the best of the author's knowledge, the present work represents the first attempt to thoroughly investigate and develop a model elucidating the process shaping the driver's understanding of DAS through their perception of the DAS. Several key aspects set the proposed model (**Figure 5.1**) apart.

In contrast to prior studies undertaken in the field of driving automation, the present model, and its associated aspects and factors presented in this thesis, provide a holistic examination of variables pertinent to the driver's interaction with DAS. In contrast to theoretical frameworks suggested previously, the present model is firmly grounded in empirical evidence derived from drivers in real world driving environments. In order to enhance the model's robustness, data from four distinct empirical studies (Study I-IV) was triangulated through a structured analysis approach. Further, the conducted studies utilized multiple methodological approaches (e.g., surveys, naturalistic driving studies, in-depth interviews, field observations) capturing a wide array of driver behaviours and impressions, from first time drivers and long term drivers, over prolonged periods. As a result, these studies provided valuable insights into how the driver's understanding of DAS is shaped through their perception.

Through the triangulation of various data points, a comprehensive list of aspects shaping the driver's understanding and factors influencing their perception was identified, allowing for a nuanced approach to the driver's interaction with DAS. Finally, the variables that were identified were systematically classified, enabling the author to analyse patterns and relationships. The comprehensive categorization presented illuminates the intricate interplay between several variables, providing useful insights into the perception and understanding of DAS by drivers.

The proposed model is notable for its attention to the driver's perception and consequent understanding of DAS, as it thoroughly addresses the criticisms put forward by other researchers. These criticisms revolve around the lack of varied and empirical methodologies and data triangulation, as well as the neglect of a wider range of factors and their interrelation (cf. [136][72][137][132][138], in the attempt to identify a holistic driver-centric perspective.

In conclusion, the model illustrates the cognitive process by which perception influences the driver's understanding and subsequent interaction with driving automation, through which the model aims to support a holistic perspective on the driver's understanding and subsequently, on the design of DAS.

Practical Contributions

The developed and presented Design for Perception Toolkit is a significant practical contribution because it is based on broad and deep empirical evidence describing the driver's perception and consequent understanding of DAS, and thus addresses one of the most important needs of the designers and developers of such systems – understanding and applying the driver's perspective. Therefore, this practical contribution aims to address the second research question (Paper H), since the toolkit, being specialised for the automotive community, addresses a gap in the methodological toolbox for designers and developers of DAS, namely the lack of a specific and structured, user-centric tool that enhances the process of designing DAS through an empirically based and guided approach.

With the toolkit, designers can utilize the specified questions: When can I use the systems? 2. What does the vehicle do? and 3. What do I do? and ask themselves questions that drivers will ask when interacting with the systems. In addition, the model defines which factors influence the driver's perception of the system, its capabilities and limitations and, consequently, how they understand its utilization. The provided information is further equipped with guiding questions, which aim to facilitate a critical review and discussions. Thus, the model can be used as a design aid to include the relevant aspects that drivers identify during their interaction with a DAS, as well as the factors affecting them.

This approach can be compared to a Cognitive Walkthrough (CW), which is an inspection method linking an user interface walkthrough to a cognitive model of a user [139]. However, since its first version, the CW has evolved and been refined to imagine and address specific scenarios for each action that a user has to take with the help of guiding questions [140]. One of the main benefits of CW is considered to be its ease of use for developers without specialised knowledge of cognitive theory [141]. However, due to its limited focus on identifying the usability problems of a particular solution, it has been criticised for lacking a high-level perspective, prioritisation of failure-and-success effects, and a complex overview of identified results [142]. While attempts have been made to modify and improve the CW [143], the improved CW has been criticised for remaining limited in its analysis and being tedious, complex and time-consuming to implement [142].

The Design for Perception Toolkit addresses these concerns by providing a comprehensive and structured analysis approach to the effects of the design on perception. Further, it provides a high-level perspective by analysing the overall effect of design decisions on the perceptions and experiences of users. In addition, it streamlines the analysis process, making it more effective and user-friendly, thereby addressing the limitations of previously mentioned method.

The tool does, however, require an initial investment of time, due to the vast quantity of information it provides. However, the designers using the toolkit concurred that this initial threshold diminished as they gained familiarity with it.

The additional critical review support that the toolkit offers, which is loosely based on Design Failure Modes and Effect Analysis (DFMEA), is another important component. Historically, an FMEA seeks to prevent the possibility that a new design, process or system fails to meet the proposed requirements in whole or in part under certain conditions. The purpose of the DFMEA is to identify and prevent design-related failure modes of products in order to validate the established design parameters for a specific functional performance level. The most important function of this type of FMEA is the identification of potential failure modes in the early stages of design development in order to eliminate their effects, select the optimal design variant and develop a documentation base to support future designs [144]. Consequently, by employing a DFMEA-based strategy for the toolkit, designers and developers evaluate the severity, frequency, and detectability of each potential failure mode for the proposed design solutions. An additional component of the toolkit is the identification of prospective areas of delight. This not only enables the designers of DAS to prioritise and mitigate the most significant risks but also aids in ensuring that the final product meets or exceeds driver expectations. In addition, this process encourages cross-functional collaboration between teams in order to collectively mitigate potential effects and improve product reliability overall.

Further, the flexibility of the toolkit to use different variants at different points of the design process, as well as in different settings, i.e., in a group or individually, supports easy implementation into everyday activities. As also stated by the designers during the evaluation of the toolkit, they do not wish to add a new step to their current processes but would prefer it to be a seamless addition.

This notion was also discussed by Gericke and colleagues [145], who concluded that for a method to be successful in the industry, it needs to add value to the operation through effectiveness and efficiency, as well as have the ability to be combined with existing processes within the company, without distracting from the work that needs to be done. Consequently, the toolkit facilitates the identification of design, evaluation, and training approaches to promote appropriate usage strategies and the development of the driver's understanding of DAS that aligns with the intended use promoted by the manufacturers.

Design for Perception

Numerous automotive manufacturers have signalled their intent to introduce highly automated vehicles to the market in the near future, indicating a massive push for automated driving functionality. The motivation is manifold and ranges from heightening comfort and traffic safety to introducing the possibility of engaging in non-driving-related tasks while travelling in the vehicle, thus offering a distinct user experience [75][18].

Nevertheless, it is anticipated that in the near future, there will continue to be a range of driving automation systems available, each with different degrees of automation, that will continue to be widely used on roadways. This implies that the human driver will assume the responsibilities of monitoring, supervising and serving as a backup to the automation in instances where the automated driving system is unable to execute its tasks. However, as discussed widely ever since the introduction of automation, humans are not cut out for these types of tasks (cf. [47][47]). Different work has shown that drivers become bored, fatigued and unalert [146], lose track of what the automation is doing, or even what the surrounding circumstances are [89][147]. Further, they may not understand what their tasks and responsibilities are, or even simply forget which driving mode the automation is engaged in [33]. Therefore, the successful implementation of DAS relies on effective cooperation between the driver and the vehicle. This requires designers to view them as a joint cognitive system, wherein both must work in cooperation to guarantee a safe and enjoyable driving experience, and thus they must take a more human-centric approach to the design of DAS.

Hence, during the development of DAS, designers must take into account that the driver's perception extends beyond the system's functionalities and their own tasks. And while the HMI can play an important role in communicating crucial information about needed actions from the driver [22][23], the driver's understanding of the DAS is not just impacted during its utilization or solely by the HMI. Rather, their knowledge is shaped by various factors and influenced by variables that extend outside the realm of user interface design. This implies that designers' understanding of the system, including its limitations, capabilities, and feedback, may not align with the driver's perception, rendering the use of technology-centred taxonomies problematic as they do not account for a comprehensive view of the complex interaction between the driver and the vehicle. Therefore, it is imperative to adopt a human factors engineering approach in addressing this issue. The human factors approach encompasses the cognitive, physical, and social dimensions of the driver's engagement with the vehicle. This approach recognises that variables such as the driver's experience, emotions and context influence their perception and behaviour. By integrating these variables into the design methodology, a human-centric and effective interface can be developed that more closely corresponds to the requirements and expectations of the driver.

However, the transfer of academic knowledge and methods into industry has proven challenging [148][149][150]. Many variables within the industrial sector, including time constraints, stringent quality criteria, and the intrinsic characteristics of existing processes, present challenges to the effective implementation and incorporation of novel methods, strategies, and tools [151][152]. It is argued that developed methods need to be developed to a degree that industry can use them in conjunction with existing methods, and that efforts in this regard should strive to amplify existing knowledge and extend existing methods instead of creating new methods to replace the existing ones [153][145]. Thus, for a method to be adapted for use, one must take into account that the users of the method need to understand and accept the method, which depends on the use contexts, their needs, and the opportunity to improve established practices meaningfully [154].

The Design for Perception toolkit prioritizes humans and their perception as the focal point of technological advancement, enabling a human-centred approach to determining the requirements for the design of a DAS. At the core of this lies the empirical knowledge, presented in the conceptual model (**Figure 5.1**), describing the process of how perception shapes understanding. However, while the model describes crucial aspects and relevant processes about the driver's perception and consequent understanding of the DAS, a theoretical model retains a high threshold when it comes to the applicability in industrial processes and design activities.

To bridge this gap, a practical framework was needed that could translate the theoretical knowledge into applicable insights [155][152]. This framework had to not only consider the driver's needs and identify their potential impact, but also its application in the industrial context as a design tool. This was achieved by developing and evaluating the toolkit within the desired context and with its target users. The involvement of the designers in the development of the tool permitted two crucial contributions to its success: (i) adoption of the knowledge gained into applicable representation variants that could be used with great flexibility (cf. [156]); and (ii) the transfer of knowledge and a mindset, empowering the adaptation of the tool (cf. [157][158].

As described in the results, the Design for Perception toolkit has the potential to be employed in conjunction with already established methods and processes. It can be utilised to investigate and broaden the scope of the problem area, as well as act as a catalyst for generating design solutions. Further, it can be employed as an assessment tool at different phases of the design process, fulfilling both evaluative and generative purposes. Therefore, it can serve as a checklist to systematically evaluate and define design solutions. It can also be utilized in workshops to facilitate collaboration and generate ideas among various stakeholders. Additionally, it can function as a comprehensive analysis tool to identify weaknesses and areas for improvement in existing solutions.

In summary, the driver's perspective on driving automation differs fundamentally from the technological perspective that currently guides the design and development of these systems. In order to improve the user experience, we must reframe our perspective in order to design products that align with the driver's mental models and facilitate their development of a sufficient understanding of the DAS. The Design for Perception mindset applies a systematic approach based on a human-centric perspective that accounts for the driver's perception. As a result, it can help practitioners to: (i) explore possible solutions using a systematic approach; (ii) identify areas for improvement from the user's perspective; and (iii) ideate and critically evaluate design decisions using a structured process. Thus, the toolkit has the potential to act as common ground, aligning the objectives and motivations of developers, designers, and strategists with those of regulators and most importantly, the drivers.

6.2 Reflections on the Approach

The technique discussion is categorised into distinct sections, each dedicated to a certain approach and concluding observations on the chosen embedded mixed-methods research design for this project. It is crucial to acknowledge that the results of the studies presented in **Chapter 4** are intrinsically linked to the selected methodology. The methodological decisions, encompassing the selection of data collection methods, sampling strategies, and analysis methods, significantly influence the findings and interpretations. Acknowledging this interdependency is crucial for a comprehensive understanding of the research findings.

Study I – International Online Survey

Due to the lack of information regarding when drivers prefer to use DAS. it was decided to conduct a global online survey to collect information from end-users. Online surveys are a quick and inexpensive method for collecting information from a large population, particularly when targeting a specific subset of respondents, regardless of their geographical location. The aim of Study I was to gain insights into when drivers use existing (Level 1 and Level 2) systems in various driving contexts, and to understand if there are any correlations between the use of one system in a situation and the use of the other system in the same situation. The use of an online survey made it possible to reach a targeted group in larger numbers, and to include diverse countries. While the use of online surveys allowed access to a unique population, it was difficult to more precisely target individuals whose vehicles were equipped with both types of systems. As a consequence, only a small sample size of the original group of respondents remained for analysis. This highlights a second issue, which is self-reported data (cf. [159]), thus being the respondents' own evaluations of their utilization strategies. In addition, there is no assurance that respondents provided accurate demographic or other information about the systems available in their cars, which is problematic in the case of DAS given that previous research indicates that many drivers are unaware of the systems installed in their vehicles [35]. In addition, interpretation of the data can be difficult because there is no way to follow up on responses or obtain additional insights beyond the survey questions, which is why the exploration phase was continued with a mixed-methods approach as a next step.

Study II – Naturalistic Driving Study with Subsequent In-Depth Interviews

Naturalistic Driving Studies permit the accumulation of sensory-based vehicle data (such as GPS data, data indicating traffic and road conditions, and data about the use of apps and DAS) over an extended period of time and in the context of natural driving. Since vehicle data is typically collected and processed through unobtrusive technologies, it is possible to monitor driver usage patterns at all times without interfering with their daily lives or the natural environment [121].

Using a longitudinal mixed-methods design, Study II was able to collect a large amount of data from 132 vehicles, allowing for the identification of distinct user groups who employed various strategies when using the DAS available in their vehicles (Level 1 and Level 2). The subsequent interviews with drivers from the identified user groups enabled validation and an in-depth understanding of the collected sensory data, resulting in greater comprehension of the situational usage of the systems and the motivations behind those choices. In addition, initially, emergent usage patterns from Study I could be analysed in depth, and points of interest could be pursued in the interviews, permitted a focused investigation of the driver's motivation for usage and the factors influencing their understanding of the systems. Lastly, the subsequent interviews with the various user groups permitted an in-depth investigation of the drivers' prior experiences and learning processes with the systems, which supported the development of a number of DAS-related aspects pertinent to driver's understanding.

However, one of the limitations of the research was that only Volvo vehicles and Volvo Cars employees were used in the tracked vehicle fleet. Even though only employees who were not engaged in the development of DAS were included in the study and invited to the interviews, one cannot exclude a bias towards the vehicle brand and their equipped systems. In addition, an ND study does not account for possible car-sharing scenarios, and the lack of a driver recognition unit on board could contribute to the problem of driving patterns from one user being indistinguishable from those of another, obscuring data based on the amount of sharing. Even though this was screened for, the self-reported car-sharing habits of the participants should be viewed with caution. A further limitation of the study is the participants' varying degrees of familiarity with the two systems. While all participants had access to the NDS vehicle for approximately three weeks, some participants had prior experience with Level 1 and/or Level 2 systems. However, prior experience was not taken into consideration in the NDS study and could only be assessed through in-depth interviews to determine learning experiences and levels of knowledge. In addition, six distinct vehicle models were included in the study, which may have affected the system performance and driving behaviour of the vehicles, such as sedan versus SUV. Despite indications to the contrary, there are no definitive data regarding the impact of vehicle type on perceived system performance.

Study III and Study IV – Woz Driving Study with Observations and In-Depth Interviews

As the objective of the third and fourth studies was to examine the driver's understanding of a vehicle with multiple levels of automation, a quasi-experimental study design employing a Wizard-of-Oz vehicle was implemented. The semicontrolled study design made it possible to account for a number of variables, including the participants, the driving route, and the levels of driving automation that were evaluated. Despite the meticulous selection of the route and session hours based on collected traffic data, the quasi-experimental design does not permit control of the traffic conditions. This variable could therefore vary in density and exposure periods for the different participants and is regarded as a possible limitation. Nonetheless, all participants had comparable exposure periods and encountered the necessary traffic conditions for the DAS to function. An alternative to conducting on-road studies would have been the use of driving simulators to shed light on the driver's interaction with the DAS (as in studies by [26][160][161][162]. However, the use of simulators was abandoned, as a simulator study does not necessarily provide an accurate depiction of how drivers react to real-world traffic and the encountered scenarios when using driving automation. Due to the incorporation of real-world interventions, a quasi-experimental design has a higher level of external validity [163].

Regarding the sample, it was advantageous for the study to conduct the experiment with novice users because they were likely to engage in more conscious reflections during the think-aloud procedures than more experienced drivers and thus provide greater insight into how they constructed their understanding of the systems. However, a disadvantage of this is that novice users may be overly enthusiastic about using a self-driving vehicle, which may lead to a bias in their behaviour [164]. Nevertheless, the adoption of a similar study design in different locations and time periods, as well as inviting novice users to participate, has proven beneficial, as participants' comments in Study IV were significantly more concerned with safety and legal issues than in Study III, indicating a correlation between mood shifts and political and social events at the time, and providing consistency and validity in the findings. Hence, the implementation of a quasi-experimental mixed-methods study design can be deemed successful because the results of the studies were comparable and also provided new insights into user motivation for using DAS and their perception of such systems.

Nonetheless, certain factors, such as vehicle behaviour, were acknowledged as being crucial to the driver's perception of the system, although the nature of this phenomenon is poorly understood. Thus, to acquire a deeper understanding, it may be necessary to examine this phenomenon in a different, and possibly more controlled setting. However, a deep dive into specific factors during the course of this research was not possible due to time restrictions.

Study V – Co-Creation

For the development and evaluation of the toolkit, the author applied a participatory research approach utilising co-design activities with practitioners in an industry setting. Participatory Research (PR) is an approach that aims to bridge research and action by directly involving stakeholders in the process, taking in their feedback, and feeding back to them [165][166]. PR encompasses a broad range of research designs and methods, employing systematic data collection and analysis, in collaboration with the goal to instigate action [167][168]. Therefore, participatory research approaches hold a key role in facilitating an exchange between researchers and users, or individuals possessing expert knowledge in the field [169]. They actively engage individuals outside the research community as close collaborators instead of as 'subjects', to improve innovation, or the quality or speed of the design process, and ultimately user satisfaction [170]. As iterated by Kujala [170], the direct involvement of product stakeholders, i.e., designers and developers of DAS, through co-design techniques during the development process has manifold benefits.

However, the most striking benefits when designing with users are that users are invited and empowered to actively shape the outcome of the design process [123][171], as well as the contribution of specific expertise regarding the usage and use context through the involved stakeholders [172]; with regard to Study V, these were designers, and developers of driving automation systems. Subsequently, the project applied the framework of User-Centred Research (UCR) in order to address the aforementioned aim. UCR is a form of co-design that is characterized by an iterative design process that involves users in the design of products that are intended for them [173][174][175].

Thus, the co-creation workshops and activities offered a cooperative environment in which the practitioners could share their experiences with and requirements for the design toolkit, with the purpose of contributing to the design process and shaping the toolkit according to their needs. The subsequent interviews supported this validation of the toolkit and an assessment of the outcome regarding its utility, strengths, limitations, and potential improvements.

However, the approach was constrained by the time and resources available for this use case study. Since all participants were primarily engaged in the development efforts of the company, carving out time for consecutive workshops and activities alongside the workshops was at times challenging, even though the participants were participating voluntarily and with great engagement. However, their pressing workload made it necessary to make concessions with regard to the study's duration and iterations. Nevertheless, despite the fact that there are details of the toolkit that could be improved or further developed, the toolkit has already proved useful to the participants, resulting in them becoming independent advocates for the toolkit.

Concluding Remarks on the Research Approach

The facilitation of empirical research is of utmost importance in order to gain a comprehensive understanding of complex phenomena [176]. Further, recognising the end-users of DAS as valuable sources of information, innovation and adaptability while striving to create and enhance DAS, is a choice that is critical to the success of the designed solution [54][114][177], and thus should be seen as the focus of attention. Numerous research endeavours pertaining to the assessment of DAS primarily rely on simulator studies. While these studies provide valuable insights into the utilisation of such systems [178][179], their limited realism undermines the ecological validation of the results. Consequently, these findings fail to provide genuine insights into the user's strategies and perception of the driving automation, as the presence of a safety net in simulated environments obscures the true experiences and challenges encountered during real-world driving scenarios [180][181][182][183].

The decision to employ a mixed-methods study approach enabled an examination of the driver's utilization, perception, and understanding of driving automation technologies, hence providing valuable insights. The integration of quantitative and qualitative data yielded a more comprehensive awareness of the subject matter compared to relying just on one method, as the different approaches typically possess distinct characteristics and explore different facets of the same issue. The utilisation of both qualitative and quantitative methodologies was additionally advantageous due to the exploratory and inductive nature of the chosen research methodology, since it facilitated the gathering of primary quantitative data from both the survey and the ND study that pertained to when and how drivers make use of DAS in their personal vehicles. Subsequently, this prompted a more comprehensive examination employing a hybrid approach of quantitative and qualitative methodologies by utilising observational techniques paired with in-depth interviews and triangulating the qualitative data with quantitative data points. This was done to acquire a broader and deeper understanding of the driver's perception and consequent understanding of DAS.

Chapter 7 Conclusions



CHAPTER 7

Conclusions

In conclusion, this thesis has made significant contributions to the understanding of the driver's perception and consequent understanding of driving automation systems and its implications for design. The research addressed two main research questions: (RQ1) What are the factors that impact the driver's perception and subsequent understanding of DAS? and (RQ2) How can this knowledge be applied to offer design recommendations for practitioners involved in the development of Driving Automation Systems? The first research question was answered by identifying a large number of aspects constituting the driver's understanding and factors influencing their perception of a driving automation system and describing the process shaping the mental model of a driver. The second research question led to the exploration and development of a method aiming to turn the gathered theoretical knowledge into an applicable tool for industry professionals.

The contributions of this thesis can be succinctly summarized as follows:

(i) The theoretical contributions of this thesis identified critical aspects shaping the driver's understanding of Driving Automation Systems and factors influ-
encing their perception.

By integrating these aspects into a unified conceptual model, this work addressed a gap in the existing literature—a lack of a holistic understanding of how the driver's perception influences their mental model of DAS. Existing research efforts often focused on limited variables and methodologies, neglecting the complexity of the dynamic driving task and the interplay between various factors. This research bridged this gap by providing a comprehensive overview grounded in empirical evidence derived from real drivers. The proposed model, developed through rigorous empirical studies and data triangulation, presents a holistic examination of variables relevant to the driver's interaction with driving automation.

(ii) The practical contributions of this thesis are manifested in the development and evaluation of the Design for Perception Toolkit.

Based on deep empirical evidence, the toolkit provides a structured, usercentric approach to designing DAS. It addresses the limitations of existing methods by offering a high-level perspective, comprehensive analysis, and a flexible application. The toolkit not only enhances the design process but also aids designers in critically reviewing and discussing design decisions. By incorporating critical reviews, it enables the ideation and evaluation of design solutions and the identification of areas for improvement. Finally, it addresses the existing gap in the automotive industry: A systematic tool or approach to effectively address the development of driving automation from a humancentric perspective which clarifies cognitive processes and drivers' needs in the interaction with a DAS.

(iii) This thesis advocates for a shift in perspective—a Design for Perception mindset. It emphasizes the human-centric approach necessary for the successful implementation of Driving Automation Systems.

By understanding the driver as part of a joint cognitive system with the vehicle, this approach aligns the objectives of developers, designers, strategists, regulators, and, most importantly, drivers. The Design for Perception mindset enables a JCS perspective by understanding the cognitive demands placed on the driver, when interacting with a DAS, and recognizing that cognition is not confined to the individual, but distributed between the driver, the vehicle, and the environment in which the collaboration takes place. Thus, the Design for Perception toolkit provides a systematic approach to exploring solutions, identifying user perspectives, and evaluating design decisions driven by a human-centric perspective of driving automation. It acts as common ground, fostering collaboration and facilitating the development of DAS that aims at facilitating a safe and enjoyable driving experience to users of DAS.

In summary, this research has significantly advanced the understanding of driver perception and interaction with DAS. By providing a comprehensive theoretical model and a practical toolkit, this work equips practitioners with a valuable approach to the design of driving automation systems that reflects the driver's mental model of such systems from a human-centric perspective.

References

- [1] P. Gao, R. Hensley, and A. Zielke. "A road map to the future for the auto industry," McKinsey & Company. (2014), [Online]. Available: https://www.mckinsey.com/ industries/automotive-and-assembly/our-insights/a-road-map-to-thefuture-for-the-auto-industry.
- [2] M. Kyriakidis, C. van de Weijer, B. van Arem, and R. Happee, "The deployment of advanced driver assistance systems in europe," SSRN Electronic Journal, 2015. DOI: 10.2139/ssrn.2559034.
- [3] National Highway Traffic Safety Administration, "Automated driving systems 2.0: A vision for safety," US Department of Transportation, DOT HS, Washington, DC, Tech. Rep. 812, 442, 2017.
- [4] J. B. Cicchino, "Effectiveness of forward collision warning and autonomous emergency braking systems in reducing front-to-rear crash rates," Accident Analysis & Prevention, vol. 99, pp. 142–152, 2017. DOI: 10.1016/j.aap.2016.11.009.
- J. B. Cicchino, "Effects of lane departure warning on police-reported crash rates," Journal of Safety Research, vol. 66, pp. 61–70, 2018. DOI: 10.1016/j.jsr.2018.05. 006.
- J. B. Cicchino, "Effects of blind spot monitoring systems on police-reported lanechange crashes," *Traffic Injury Prevention*, vol. 19, no. 6, pp. 615–622, 2018. DOI: 10.1080/15389588.2018.1476973.
- [7] "Taxonomy and definitions for terms related to driving automation systems for onroad motor vehicles," SAE International. (2021).
- [8] M. Marcano, S. Diaz, J. Perez, and E. Irigoyen, "A review of shared control for automated vehicles: Theory and applications," *IEEE Transactions on Human-Machine Systems*, vol. 50, no. 6, pp. 475–491, 2020. DOI: 10.1109/thms.2020.3017748.
- [9] C. Ahlström, A. Anund, C. Fors, and T. Åkerstedt, "Effects of the road environment on the development of driver sleepiness in young male drivers," *Accident Analysis* & *Prevention*, vol. 112, pp. 127–134, 2018. DOI: 10.1016/j.aap.2018.01.012.

- [10] M. M. Ahmed and A. Ghasemzadeh, "The impacts of heavy rain on speed and headway behaviors: An investigation using the shrp2 naturalistic driving study data," *Transportation Research Part C: Emerging Technologies*, vol. 91, pp. 371–384, 2018. DOI: 10.1016/j.trc.2018.04.012.
- [11] E. Papazikou, M. A. Quddus, and P. Thomas, "Detecting deviation from normal driving using shrp2 nds data," in *Presented at the Transportation Research Board* (TRB) 96th Annual Meeting, Washington D.C., US, 8th-12th January 2017, 2017.
- [12] Y. Zhai, T. Wo, X. Lin, Z. Huang, and J. Chen, "A context-aware evaluation method of driving behavior," in Advances in Knowledge Discovery and Data Mining: 22nd Pacific-Asia Conference, PAKDD 2018, Melbourne, VIC, Australia, June 3-6, 2018, Proceedings, Part I, Springer International Publishing, 2018, pp. 462–474.
- [13] J. W. Jenness, N. D. Lerner, S. Mazor, J. S. Osberg, and B. C. Tefft, "Use of advanced in-vehicle technology by young and older early adopters. survey results on adaptive cruise control systems," Tech. Rep. DOT HS, 810, 917, 2008.
- [14] A. F. Larsson, "Driver usage and understanding of adaptive cruise control," Applied Ergonomics, vol. 43, no. 3, pp. 501–506, 2012. DOI: 10.1016/j.apergo.2011.08.005.
- [15] A. McDonald, C. Carney, and D. McGehee, "Vehicle owners' experiences with and reactions to advanced driver assistance systems," AAA Foundation for Traffic Safety, Washington, D.C., Tech. Rep., 2018.
- [16] A. Boelhouwer, A. van den Beukel, M. van der Voort, C. Hottentot, R. de Wit, and M. Martens, "How are car buyers and car sellers currently informed about adas? an investigation among drivers and car sellers in the netherlands," *Transportation Research Interdisciplinary Perspectives*, vol. 4, p. 100103, 2020. DOI: 10.1016/j.trip.2020.100103.
- [17] M. Regan, P. Prabhakharan, P. Wallace, M. Cunningham, and J. Bennett, "Educating and training drivers of vehicles equipped with assisted and automated driving features," Austroads, AP-R616-20, 2020.
- [18] L. Vasile, N. Dinkha, B. Seitz, C. Däsch, and D. Schramm, "Comfort and safety in conditional automated driving in dependence on personal driving behavior," *IEEE Open Journal of Intelligent Transportation Systems*, pp. 1–1, 2023. DOI: 10.1109/ ojits.2023.3323431.
- [19] M. R. Endsley, "Level of automation forms a key aspect of autonomy design," Journal of Cognitive Engineering and Decision Making, vol. 12, no. 1, pp. 29–34, 2017. DOI: 10.1177/1555343417723432.
- [20] N. B. Sarter and D. D. Woods, "How in the world did we ever get into that mode? mode error and awareness in supervisory control," *Human Factors: The Journal of* the Human Factors and Ergonomics Society, vol. 37, no. 1, pp. 5–19, 1995. DOI: 10.1518/001872095779049516.
- [21] A. Feldhütter, N. Härtwig, C. Kurpiers, J. Hernandez, and K. Bengler, "Effect on mode awareness when changing from conditionally to partially automated driving," in *Proceedings of the 20th Congress of the International Ergonomics Association* (*IEA 2018*), ser. Advances in Intelligent Systems and Computing, vol. 823, Springer, 2019, pp. 240–244. DOI: 10.1007/978-3-319-96074-6_34.

- [22] V. A. Banks, A. Eriksson, J. O'Donoghue, and N. A. Stanton, "Is partially automated driving a bad idea? observations from an on-road study," *Applied Ergonomics*, vol. 68, pp. 138–145, 2018. DOI: 10.1016/j.apergo.2017.11.010.
- [23] O. Carsten and M. H. Martens, "How can humans understand their automated cars? hmi principles, problems and solutions," *Cognition, Technology & Work*, vol. 21, no. 1, pp. 3–20, 2018. DOI: 10.1007/s10111-018-0484-0.
- [24] B. Zhang, J. De Winter, S. Varotto, R. Happee, and M. Martens, "Determinants of take-over time from automated driving: A meta-analysis of 129 studies," *Transport Research Part F: Traffic Psychology & Behaviour*, vol. 64, pp. 285–307, 2019. DOI: 10.1016/j.trf.2019.04.020.
- [25] S. Kim, R. van Egmond, and R. Happee, "Effects of user interfaces on take-over performance: A review of the empirical evidence," *Information*, vol. 12, no. 4, p. 162, 2021. DOI: 10.3390/info12040162.
- [26] M. Beggiato and J. F. Krems, "The evolution of mental model, trust and acceptance of adaptive cruise control in relation to initial information," *Transportation Research Part F: Traffic Psychology and Behaviour*, vol. 18, pp. 47–57, 2013. DOI: 10.1016/ j.trf.2012.12.006.
- [27] J. Jenness et al., "Spaceship, guardian, coach: Drivers' mental models of advanced vehicle technology," in *HCI International 2019 Posters*, C. Stephanidis, Ed., ser. Communications in Computer and Information Science, vol. 1034, Cham: Springer, 2019. DOI: 10.1007/978-3-030-23525-3_46.
- [28] C. Carney, J. G. Gaspar, and W. J. Horrey, "Longer-term exposure vs training: Their effect on drivers' mental models of adas technology," *Transportation Research Part F: Traffic Psychology and Behaviour*, vol. 91, pp. 329–345, 2022. DOI: 10.1016/ j.trf.2022.09.017.
- [29] F. Novakazi, J. Orlovska, L. O. Bligård, and C. Wickman, "Stepping over the threshold linking understanding and usage of automated driver assistance systems (adas)," *Transportation Research Interdisciplinary Perspectives*, vol. 8, p. 100 252, 2020. DOI: 10.1016/j.trip.2020.100252.
- [30] R. E. Llaneras, "Exploratory study of early adopters, safety-related driving with advanced technologies. draft final task 2 report: In-vehicle systems inventory, recruitment methods & approaches, and owner interview results," Tech. Rep. HS-809 972, 2006.
- [31] S. Choi, F. Thalmayr, D. Wee, and F. Weig. "Advanced driver-assistance systems: Challenges and opportunities ahead." (2016).
- B. D. Seppelt and J. D. Lee, "Making adaptive cruise control (acc) limits visible," International Journal of Human-Computer Studies, vol. 65, no. 3, pp. 192–205, 2007. DOI: 10.1016/j.ijhcs.2006.10.001.
- [33] K. M. Wilson, S. Yang, T. Roady, J. Kuo, and M. G. Lenné, "Driver trust & mode confusion in an on-road study of level-2 automated vehicle technology," *Safety Sci*ence, vol. 130, p. 104 845, 2020. DOI: 10.1016/j.ssci.2020.104845.

- [34] S. D. Leonard and E. W. Karnes, "Compatibility of safety and comfort in vehicles," Proceedings of the Human Factors and Ergonomics Society Annual Meeting, vol. 44, no. 20, pp. 3–357, 2000. DOI: 10.1177/154193120004402036.
- [35] L. Viktorová and M. Sucha, "Drivers' acceptance of advanced driver assistance systems. what to consider?" *International Journal for Traffic and Transport Engineering*, vol. 8, no. 3, pp. 320–333, 2018. DOI: 10.7708/ijtte.2018.8(3).06.
- [36] I. M. Harms, L. Bingen, and J. Steffens, "Addressing the awareness gap: A combined survey and vehicle registration analysis to assess car owners' usage of adas in fleets," *Transportation Research Part A: Policy and Practice*, vol. 134, pp. 65–77, 2020. DOI: 10.1016/j.tra.2020.01.018.
- [37] O. Oviedo-Trespalacios, J. Tichon, and O. Briant, "Is a flick-through enough? a content analysis of advanced driver assistance systems (adas) user manuals," *PLOS ONE*, vol. 16, no. 6, e0252688, 2021. DOI: 10.1371/journal.pone.0252688.
- [38] H. Abraham, B. Seppelt, B. Mehler, and B. Reimer, "What's in a name: Vehicle technology branding & consumer expectations for automation," in *Proceedings of the* 9th international conference on automotive user interfaces and interactive vehicular applications, 2017, pp. 226–234.
- [39] J. H. Yang, J. Han, and J. M. Park, "Toward defining driving automation from a human-centered perspective," in *Proceedings of the 2017 CHI Conference Extended Abstracts on Human Factors in Computing Systems*, 2017, pp. 2248–2254.
- [40] B. Seppelt, B. Reimer, L. Russo, B. Mehler, J. Fisher, and D. Friedman. "Towards a human-centric taxonomy of automation types." White Paper. (2018).
- [41] G. A. Jamieson and G. Skraaning, "Levels of automation in human factors models for automation design: Why we might consider throwing the baby out with the bathwater," *Journal of Cognitive Engineering and Decision Making*, vol. 12, no. 1, pp. 42–49, 2017. DOI: 10.1177/1555343417732856.
- [42] B. Kirwan, A Guide To Practical Human Reliability Assessment. CRC Press, 2017.
- [43] J. F. Schønheyder and K. Nordby, "The use and evolution of design methods in professional design practice," *Design Studies*, vol. 58, pp. 36–62, 2018. DOI: 10. 1016/j.destud.2018.04.001.
- [44] Merriam-Webster. "Automation." Accessed 10 Jun. 2023. (), [Online]. Available: https://www.merriam-webster.com/dictionary/automation.
- [45] P. M. Fitts, "The information capacity of the human motor system in controlling the amplitude of movement," *Journal of Experimental Psychology*, vol. 47, no. 6, pp. 381–391, 1954. DOI: 10.1037/h0055392.
- [46] R. Parasuraman, T. Sheridan, and C. Wickens, "A model for types and levels of human interaction with automation," *IEEE Transactions on Systems, Man, and Cybernetics - Part A: Systems and Humans*, vol. 30, no. 3, pp. 286–297, 2000. DOI: 10.1109/3468.844354.
- [47] L. Bainbridge, "Ironies of automation," Automatica, vol. 19, no. 6, pp. 775–779, 1983. DOI: 10.1016/0005-1098(83)90046-8.

- [48] G. Baxter, J. Rooksby, Y. Wang, and A. Khajeh-Hosseini, "The ironies of automation...still going strong at 30?" In *Proceedings of the 30th European Conference on Cognitive Ergonomics*, ACM, 2012, pp. 65–71.
- [49] S. W. A. Dekker and D. D. Woods, "Maba-maba or abracadabra? progress on human-automation co-ordination," *Cognition, Technology & Work*, vol. 4, no. 4, pp. 240–244, 2002. DOI: 10.1007/s101110200022.
- [50] D. B. Kaber and M. R. Endsley, "The effects of level of automation and adaptive automation on human performance, situation awareness and workload in a dynamic control task," *Theoretical Issues in Ergonomics Science*, vol. 5, no. 2, pp. 113–153, 2004. DOI: 10.1080/1463922021000054335.
- [51] M. R. Endsley and D. J. Garland, Situation Awareness Analysis and Measurement. CRC Press, 2000.
- [52] M. R. Endsley, "Automation and situation awareness," in Automation and Human Performance: Theory and Applications, R. Parasuraman and M. Mouloua, Eds., Mahwah, NJ: Lawrence Erlbaum Associates, 1996, pp. 163–181.
- [53] C. E. Billings, Aviation automation: The search for a human-centered approach. CRC Press, 2018.
- [54] E. Hollnagel and D. D. Woods, Joint Cognitive Systems: Foundations of Cognitive Systems Engineering. Boca Raton, FL, USA: Taylor & Francis Books, Inc., 2005.
- [55] J. V. Draper, "Teleoperators for advanced manufacturing: Applications and human factors challenges," *International Journal of Human Factors in Manufacturing*, vol. 5, no. 1, pp. 53–85, 1995. DOI: 10.1002/hfm.4530050105.
- [56] D. B. Kaber, "The effect of level of automation and adaptive automation on performance in dynamic control environments," Ph.D. dissertation, Texas Tech University, 1996.
- [57] M. R. Endsley, "The application of human factors to the development of expert systems for advanced cockpits," *Proceedings of the Human Factors Society Annual Meeting*, vol. 31, no. 12, pp. 1388–1392, 1987. DOI: 10.1177/154193128703101219.
- [58] M. R. Endsley and D. B. Kaber, "Level of automation effects on performance, situation awareness and workload in a dynamic control task," *Ergonomics*, vol. 42, no. 3, pp. 462–492, 1999. DOI: 10.1080/001401399185595.
- [59] T. B. Sheridan, "Human supervisory control," in Handbook of human factors and ergonomics, 2012, pp. 990–1015.
- [60] T. B. Sheridan and W. Verplank, "Human and computer control of undersea teleoperators," MIT Man-Machine Systems Laboratory, Cambridge, MA, Technical Report, 1978.
- [61] D. Spath, M. Braun, and W. Bauer, "Integrated human and automation systems," in Springer Handbook of Automation, ser. Springer Handbooks, Berlin, Heidelberg: Springer, 2009, pp. 436–459. DOI: 10.1007/978-3-540-78831-7_34.
- [62] R. Onken and A. Schulte, System-Ergonomic Design of Cognitive Automation. Springer Science & Business Media, 2010.

- [63] M. R. Endsley, "From here to autonomy," Human Factors: The Journal of the Human Factors and Ergonomics Society, vol. 59, no. 1, pp. 5–27, 2016. DOI: 10. 1177/0018720816681350.
- [64] Aircraft Accident Investigation Bureau, "Investigation report on accident to the b737-max8 reg. et-avj operated by ethiopian airlines 10 march, 2019," Tech. Rep. AI-01/19, 2022.
- [65] N. B. Sarter and D. D. Woods, "Team play with a powerful and independent agent: Operational experiences and automation surprises on the airbus a-320," *Human Factors: The Journal of the Human Factors and Ergonomics Society*, vol. 39, no. 4, pp. 553–569, 1997. DOI: 10.1518/001872097778667997.
- [66] B. G. Kanki, J. Anca, and R. L. Helmreich, Crew Resource Management. Cambridge, MA: Academic Press, 2010.
- [67] S. M. Casner, The Pilot's Guide to the Airline Cockpit. Aviation Supplies & Academics, 2013.
- [68] S. M. Casner and E. L. Hutchins, "What do we tell the drivers? toward minimum driver training standards for partially automated cars," *Journal of Cognitive En*gineering and Decision Making, vol. 13, no. 2, pp. 55–66, 2019. DOI: 10.1177/ 1555343419830901.
- [69] M. Vagia, A. A. Transeth, and S. A. Fjerdingen, "A literature review on the levels of automation during the years. what are the different taxonomies that have been proposed?" *Applied Ergonomics*, vol. 53, pp. 190–202, 2016. DOI: 10.1016/j.apergo. 2015.09.013.
- [70] S. E. Merriman, K. L. Plant, K. M. Revell, and N. A. Stanton, "Challenges for automated vehicle driver training: A thematic analysis from manual and automated driving," *Transportation Research Part F: Traffic Psychology and Behaviour*, vol. 76, pp. 238–268, 2021. DOI: 10.1016/j.trf.2020.10.011.
- [71] J. Naranjo, C. Gonzalez, J. Reviejo, R. Garcia, and T. de Pedro, "Adaptive fuzzy control for inter-vehicle gap keeping," *IEEE Transactions on Intelligent Transportation Systems*, vol. 4, no. 3, pp. 132–142, 2003. DOI: 10.1109/tits.2003.821294.
- [72] M. S. Young and N. A. Stanton, "Taking the load off: Investigations of how adaptive cruise control affects mental workload," *Ergonomics*, vol. 47, no. 9, pp. 1014–1035, 2004. DOI: 10.1080/00140130410001686348.
- [73] K. Bengler, K. Dietmayer, B. Farber, M. Maurer, C. Stiller, and H. Winner, "Three decades of driver assistance systems: Review and future perspectives," *IEEE Intelligent Transportation Systems Magazine*, vol. 6, no. 4, pp. 6–22, 2014. DOI: 10.1109/ mits.2014.2336271.
- [74] M. Brännström, E. Coelingh, and J. Sjöberg, "Decision-making on when to brake and when to steer to avoid a collision," *International Journal of Vehicle Safety*, vol. 7, no. 1, p. 87, 2014. DOI: 10.1504/ijvs.2014.058243.

- [75] A. Ziebinski, R. Cupek, D. Grzechca, and L. Chruszczyk, "Review of advanced driver assistance systems (adas)," in *Proceedings of the international conference of computational Methods in Sciences and Engineering (ICCMSE-2017)*, Thessaloniki, Greece, 2017, p. 120002. DOI: 10.1063/1.5012394.
- [76] J. M. Sullivan, M. J. Flannagan, A. K. Pradhan, and S. Bao, "Literature review of behavioral adaptation to advanced driver assistance systems," AAA Foundation of Traffic Safety, Washington, DC, Tech. Rep., 2016.
- [77] D. B. Kaber, "Issues in human-automation interaction modeling: Presumptive aspects of frameworks of types and levels of automation," *Journal of Cognitive En*gineering and Decision Making, vol. 12, no. 1, pp. 7–24, 2017. DOI: 10.1177/ 1555343417737203.
- [78] J. D. Lee, "Perspectives on automotive automation and autonomy," Journal of Cognitive Engineering and Decision Making, vol. 12, no. 1, pp. 53–57, 2017. DOI: 10.1177/1555343417726476.
- [79] B. D. Seppelt and J. D. Lee, "Modeling driver response to imperfect vehicle control automation," *Procedia Manufacturing*, vol. 3, pp. 2621–2628, 2015. DOI: 10.1016/ j.promfg.2015.07.605.
- [80] F. Flemisch, M. Heesen, T. Hesse, J. Kelsch, A. Schieben, and J. Beller, "Towards a dynamic balance between humans and automation: Authority, ability, responsibility and control in shared and cooperative control situations," *Cognition, Technology & Work*, vol. 14, no. 1, pp. 3–18, 2011. DOI: 10.1007/s10111-011-0191-6.
- [81] Organisation for Economic Co-operation and Development, Behavioural adaptations to changes in the road transport system. Paris: Organisation for Economic Co-operation and Development, 1990.
- [82] M. P. Manser, J. Creaser, and L. Boyle, Behavioural Adaptation and Road Safety: Theory, Evidence, and Action. London: CRC Press, 2013.
- [83] F. Saad, "Dealing with behavioural adaptations to advanced driver support systems," in *Modelling Driver Behaviour in Automotive Environments*, P. Cacciabue, Ed., London: Springer, 2007. DOI: 10.1007/978-1-84628-618-6_9.
- [84] B. Metz, J. Wörle, M. Hanig, M. Schmitt, A. Lutz, and A. Neukum, "Repeated usage of a motorway automated driving function: Automation level and behavioural adaption," *Transportation Research Part F: Traffic Psychology and Behaviour*, vol. 81, pp. 82–100, 2021. DOI: 10.1016/j.trf.2021.05.017.
- [85] G. B. Grayson, T. R. L. Crowthorne, and T. R. Laboratory, "Behavioural adaptation," 1996.
- [86] R. Parasuraman and V. Riley, "Humans and automation: Use, misuse, disuse, abuse," Human Factors: The Journal of the Human Factors and Ergonomics Society, vol. 39, no. 2, pp. 230–253, 1997. DOI: 10.1518/001872097778543886.
- [87] H. Xiong, L. N. Boyle, J. Moeckli, B. R. Dow, and T. L. Brown, "Use patterns among early adopters of adaptive cruise control," *Human Factors: The Journal of the Human Factors and Ergonomics Society*, vol. 54, no. 5, pp. 722–733, 2012. DOI: 10.1177/0018720811434512.

- [88] F. Naujoks, A. Kiesel, and A. Neukum, "Cooperative warning systems: The impact of false and unnecessary alarms on drivers' compliance," Accident Analysis & Prevention, vol. 97, pp. 162–175, 2016. DOI: 10.1016/j.aap.2016.09.009.
- [89] J. C. de Winter, R. Happee, M. H. Martens, and N. A. Stanton, "Effects of adaptive cruise control and highly automated driving on workload and situation awareness: A review of the empirical evidence," *Transportation Research Part F: Traffic Psychol*ogy and Behaviour, vol. 27, pp. 196–217, 2014. DOI: 10.1016/j.trf.2014.06.016.
- [90] R. D. Ledesma, S. A. Montes, F. M. Poó, and M. F. López-Ramón, "Measuring individual differences in driver inattention," *Human Factors: The Journal of the Human Factors and Ergonomics Society*, vol. 57, no. 2, pp. 193–207, 2014. DOI: 10.1177/0018720814546530.
- [91] S. Hergeth, L. Lorenz, R. Vilimek, and J. F. Krems, "Keep your scanners peeled," *Human Factors: The Journal of the Human Factors and Ergonomics Society*, vol. 58, no. 3, pp. 509–519, 2016. DOI: 10.1177/0018720815625744.
- [92] E. E. Miller and L. N. Boyle, "Driver adaptation to lane keeping assistance systems: Do drivers become less vigilant?" *Proceedings of the Human Factors and Er*gonomics Society Annual Meeting, vol. 61, no. 1, pp. 1934–1938, 2017. DOI: 10.1177/ 1541931213601963.
- [93] N. Merat, B. Seppelt, T. Louw, et al., "The "out-of-the-loop" concept in automated driving: Proposed definition, measures and implications," Cognition, Technology & Work, vol. 21, no. 1, pp. 87–98, 2018. DOI: 10.1007/s10111-018-0525-8.
- [94] T. Vogelpohl, M. Kühn, T. Hummel, and M. Vollrath, "Asleep at the automated wheel—sleepiness and fatigue during highly automated driving," Accident Analysis & Prevention, vol. 126, pp. 70–84, 2019. DOI: 10.1016/j.aap.2018.03.013.
- [95] H. Kim, M. Song, and Z. Doerzaph, "Is driving automation used as intended? realworld use of partially automated driving systems and their safety consequences," *Transportation Research Record*, vol. 2676, pp. 30–37, 2021.
- [96] N. Monsaingeon, L. Caroux, A. Mougine, S. Langlois, and C. Lemercier, "Impact of interface design on drivers' behavior in partially automated cars: An on-road study," *Transportation Research Part F-Traffic Psychology and Behaviour*, vol. 81, pp. 508– 521, 2021. DOI: 10.1016/j.trf.2021.06.019.
- [97] H. Homans, J. Radlmayr, and K. Bengler, "Levels of driving automation from a user's perspective: How are the levels represented in the user's mental model?" In Advances in Intelligent Systems and Computing, vol. 1018, 2020, pp. 21–27.
- [98] P. J. Smith, "Conceptual frameworks to guide design," Journal of Cognitive Engineering and Decision Making, vol. 12, no. 1, pp. 50–52, 2017. DOI: 10.1177/ 1555343417732239.
- [99] J. Hochberg, "Perception: Toward the recovery of a definition," *Psychological Review*, vol. 63, no. 6, p. 400, 1956.
- [100] J. J. Gibson, "A theory of direct visual perception," in Vision and Mind: Selected Readings in the Philosophy of Perception, A. Noe and E. Thompson, Eds., MIT Press, 1972, pp. 77–89.

- [101] R. L. Gregory, Concepts and Mechanisms of Perception. London: Duckworth, 1974.
- [102] J. S. Bruner, L. Postman, and J. Rodrigues, "Expectations and the perception of color," *American Journal of Psychology*, vol. 64, pp. 216–27, 1951.
- [103] F. H. Allport, "Theories of perception and the concept of structure: A review and critical analysis with an introduction to a dynamic-structural theory of behavior," 1955.
- [104] R. L. Oliver, "Effect of expectation and disconfirmation on postexposure product evaluations: An alternative interpretation," *Journal of Applied Psychology*, vol. 62, no. 4, pp. 480–486, 1977. DOI: 10.1037/0021-9010.62.4.480.
- [105] APA Dictionary of Psychology. "Perception." (2023), [Online]. Available: https: //dictionary.apa.org/perception.
- [106] W. M. Wundt and W. Wirth, Grundzüge der Physiologischen Psychologie. 3 Bde. [And] Gesamtregister, Bearb. Von W. Wirth. 1902.
- [107] V. Gijsbers, "Understanding, explanation, and unification," Studies in History and Philosophy of Science Part A, vol. 44, no. 3, pp. 516–522, 2013. DOI: 10.1016/j. shpsa.2012.12.003.
- [108] P. N. Johnson-Laird, Mental Models: Towards a Cognitive Science of Language, Inference, and Consciousness. Harvard University Press, 1983.
- [109] J. M. Carroll and J. R. Olson, "Mental models in human-computer interaction," in Handbook of Human-Computer Interaction, 1988, pp. 45–65.
- [110] W. B. Rouse and N. M. Morris, "On looking into the black box: Prospects and limits in the search for mental models," *Psychological Bulletin*, vol. 100, no. 3, p. 349, 1986.
- [111] D. A. Dickie and L. N. Boyle, "Drivers' understanding of adaptive cruise control limitations," in *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, vol. 53, Los Angeles, CA: SAGE Publications, 2009, pp. 1806–1810.
- [112] B. D. Seppelt and J. D. Lee, "Keeping the driver in the loop: Dynamic feedback to support appropriate use of imperfect vehicle control automation," *International Journal of Human-Computer Studies*, vol. 125, pp. 66–80, 2019. DOI: 10.1016/j. ijhcs.2018.12.009.
- [113] K. Blömacher, G. Nöcker, and M. Huff, "The evolution of mental models in relation to initial information while driving automated," *Transportation Research Part F: Traffic Psychology and Behaviour*, vol. 68, pp. 198–217, 2020. DOI: 10.1016/j.trf. 2019.11.003.
- [114] M. Beggiato, F. Hartwich, K. Schleinitz, J. Krems, I. Othersen, and I. Petermann-Stock, "What would drivers like to know during automated driving? information needs at different levels of automation," in 7. Tagung Fahrerassistenzsysteme, 2015. DOI: 10.13140/RG.2.1.2462.6007.
- [115] J. G. Gaspar, C. Carney, E. Shull, and W. J. Horrey, "The impact of driver's mental models of advanced vehicle technologies on safety and performance," 2020.

- [116] J. G. Gaspar, C. Carney, E. Shull, and W. J. Horrey, "Mapping drivers' mental models of adaptive cruise control to performance," *Transportation Research Part F: Traffic Psychology and Behaviour*, vol. 81, pp. 622–638, 2021.
- [117] R. B. Johnson and A. J. Onwuegbuzie, "Mixed methods research: A research paradigm whose time has come," *Educational Researcher*, vol. 33, no. 7, pp. 14–26, 2004. DOI: 10.3102/0013189x033007014.
- [118] K. Moon and D. Blackman, "A guide to understanding social science research for natural scientists," *Conservation Biology*, vol. 28, no. 5, pp. 1167–1177, 2014. DOI: 10.1111/cobi.12326.
- [119] J. W. Creswell and V. L. P. Clark, Designing and Conducting Mixed Methods Research. SAGE Publications, 2017.
- [120] J. W. Creswell, Research design: Qualitative, quantitative, and mixed methods approaches, 4th. Sage Publications, 2014.
- [121] L. Fridman, D. E. Brown, M. Glazer, et al., "Mit advanced vehicle technology study: Large-scale naturalistic driving study of driver behavior and interaction with automation," *IEEE Access*, vol. 7, pp. 102021–102038, 2019. DOI: 10.1109/access. 2019.2926040.
- [122] J. F. Kelley, "An iterative design methodology for user-friendly natural language office information applications," ACM Transactions on Information Systems, vol. 2, no. 1, pp. 26–41, 1984. DOI: 10.1145/357417.357420.
- [123] E. B. N. Sanders and P. J. Stappers, "Co-creation and the new landscapes of design," *CoDesign*, vol. 4, no. 1, pp. 5–18, 2008. DOI: 10.1080/15710880701875068.
- [124] Swedish Research Council. "Good research practice." Retrieved 18th of October 2023. (2017).
- [125] J. Ritchie and L. Spencer, "Qualitative data analysis for applied policy research," *Analyzing qualitative data*, pp. 173–194, 2002.
- [126] L. Spencer, J. Ritchie, R. Ormston, W. O'Connor, and M. Barnard, Analysis: Principles and processes, 2nd. Sage, 2014, pp. 269–290.
- [127] V. Braun and V. Clarke, "Using thematic analysis in psychology," Qualitative Research in Psychology, vol. 3, no. 2, pp. 77–101, 2006.
- [128] R. Likert, "A technique for the measurement of attitudes," Archives of Psychology, 1932.
- [129] C. Spinuzzi, "The methodology of participatory design," *Technical Communication*, vol. 52, no. 2, pp. 163–174, 2005.
- [130] J. Schleith and D. Tsar, "Triple diamond design process," in HCI International 2022 - Late Breaking Papers. Design, User Experience and Interaction. HCII 2022, ser. Lecture Notes in Computer Science, vol. 13516, Springer, 2022, pp. 1–13. DOI: 10.1007/978-3-031-17615-9_9.
- [131] B. Strauch, "Ironies of automation: Still unresolved after all these years," *IEEE Transactions on Human-Machine Systems*, vol. 48, no. 5, pp. 419–433, 2017.

- [132] A. K. Frison, Y. Forster, P. Wintersberger, V. Geisel, and A. Riener, "Where we come from and where we are going: A systematic review of human factors research in driving automation," *Applied Sciences*, vol. 10, no. 24, p. 8914, 2020. DOI: 10. 3390/app10248914.
- [133] E. Roth, B. Depass, J. Harter, R. Scott, and J. Wampler, "Beyond levels of automation: Developing more detailed guidance for human automation interaction design," *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, vol. 62, no. 1, pp. 150–154, 2018. DOI: 10.1177/1541931218621035.
- [134] J. Parker, J. D. Lee, A. McDonald, P. Rau, and B. Philips, "Do researchers use the levels of automation appropriately? how automation is discussed in research," in *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, 2023. DOI: 10.1177/21695067231195913.
- [135] J. A. Michon, "A critical view of driver behavior models: What do we know, what should we do?" In *Human behavior and traffic safety*, Boston, MA: Springer US, 1985, pp. 485–524.
- [136] N. A. Stanton and M. S. Young, "A proposed psychological model of driving automation," *Theoretical Issues in Ergonomics Science*, vol. 1, no. 4, pp. 315–331, 2000. DOI: 10.1080/14639220052399131.
- [137] D. D. Heikoop, J. C. de Winter, B. van Arem, and N. A. Stanton, "Psychological constructs in driving automation: A consensus model and critical comment on construct proliferation," *Theoretical Issues in Ergonomics Science*, vol. 17, no. 3, pp. 284–303, 2015. DOI: 10.1080/1463922x.2015.1101507.
- [138] J. de Winter, N. Stanton, and Y. B. Eisma, "Is the take-over paradigm a mere convenience?" *Transportation Research Interdisciplinary Perspectives*, vol. 10, p. 100370, 2021. DOI: 10.1016/j.trip.2021.100370.
- [139] C. Lewis, P. G. Polson, C. Wharton, and J. Rieman, "Testing a walkthrough methodology for theory-based design of walk-up-and-use interfaces," in *Proceedings of the* SIGCHI conference on Human factors in computing systems, 1990, pp. 235–242.
- [140] C. Wharton, J. Rieman, C. Lewis, and P. G. Polson, "The cognitive walkthrough method: A practitioner's guide," in *Usability Inspection Methods*, J. Nielsen and R. L. Mack, Eds., USA: John Wiley & Sons, Inc., 1994, pp. 105–140.
- [141] J. May and P. Barnard, "The case for supportive evaluation during design," Interacting With Computers, vol. 7, no. 2, pp. 115–143, 1995. DOI: 10.1016/0953-5438(95)93504-x.
- [142] L.-O. Bligård and A.-L. Osvalder, "Enhanced cognitive walkthrough: Development of the cognitive walkthrough method to better predict, identify, and present usability problems," *Advances in Human-Computer Interaction*, vol. 2013, Article ID 931698, 2013. DOI: 10.1155/2013/931698.
- [143] C. Lewis and C. Wharton, "Cognitive walkthrough," in *Handbook of Human-Computer Interaction*, M. Helander, T. K. Landauer, and P. Prabhu, Eds., Elsevier Science BV, 1997, pp. 717–732.

- [144] K. D. Sharma and S. Srivastava, "Failure mode and effect analysis (fmea) implementation: A literature review," J Adv Res Aeronaut Space Sci, vol. 5, no. 1-2, pp. 1–17, 2018.
- [145] K. Gericke, C. Eckert, F. Campean, et al., "Supporting designers: Moving from method menagerie to method ecosystem," *Design Science*, vol. 6, 2020. DOI: 10. 1017/dsj.2020.21.
- [146] M. Körber, A. Cingel, M. Zimmermann, and K. Bengler, "Vigilance decrement and passive fatigue caused by monotony in automated driving," *Procedia Manufacturing*, vol. 3, pp. 2403–2409, 2015. DOI: 10.1016/j.promfg.2015.07.499.
- [147] T. Louw and N. Merat, "Are you in the loop? using gaze dispersion to understand driver visual attention during vehicle automation," *Transportation Research Part C: Emerging Technologies*, vol. 76, pp. 35–50, 2017. DOI: 10.1016/j.trc.2017.01.001.
- [148] T. Tomiyama, P. Gu, Y. Jin, D. Lutters, C. Kind, and F. Kimura, "Design methodologies: Industrial and educational applications," *CIRP Annals*, vol. 58, no. 2, pp. 543– 565, 2009. DOI: 10.1016/j.cirp.2009.09.003.
- [149] S. Jagtap, A. Warell, V. Hiort, D. Motte, and A. Larsson, "Design methods and factors influencing their uptake in product development companies: A review," in *Proceedings of International Design Conference - Design 2014*, The Design Society, 2014, pp. 231–240.
- [150] K. Gericke, J. Kramer, and C. Roschuni, "An exploratory study of the discovery and selection of design methods in practice," *Journal of Mechanical Design*, vol. 138, no. 10, 2016. DOI: 10.1115/1.4034088.
- [151] R. Stetter and U. Lindemann, "The transfer of methods into industry," in Design process improvement: A review of current practice, 2005, pp. 436–459.
- J. Daalhuizen, O. Person, and V. Gattol, "A personal matter? an investigation of students' design process experiences when using a heuristic or a systematic method," *Design Studies*, vol. 35, no. 2, pp. 133–159, 2014. DOI: 10.1016/j.destud.2013.10. 004.
- [153] I. Karlsson and M. Kaulio, "Improving capabilities in new product development: Experiences from an action research project," in *Proceedings from the 10th International Product Development Management Conference*, Brussels, 2003, pp. 503– 515.
- [154] K. Gericke, C. Eckert, and M. Stacey, "Elements of a design method a basis for describing and evaluating design methods," *Design Science*, vol. 8, 2022. DOI: 10.1017/dsj.2022.23.
- [155] L. T. Blessing and A. Chakrabarti, "Drm: A design research methodology," in DRM: A Design Research Methodology, Springer London, 2009, pp. 13–42.
- [156] K. Gericke, C. Eckert, and M. Stacey, "What do we need to say about a design method?" In Proceedings of the International Conference on Engineering Design, ICED, The Design Society, 2017.
- [157] M. M. Andreasen, "Improving design methods' usability by a mindset approach," in *Human Behaviour in Design*, U. Lindemann, Ed., Springer, 2003, pp. 209–218.

- [158] K. Vaajakallio and T. Mattelmäki, "Design games in codesign: As a tool, a mindset and a structure," *CoDesign*, vol. 10, no. 1, pp. 63–77, 2014. DOI: 10.1080/15710882. 2014.881886.
- [159] K. B. Wright, "Researching internet-based populations: Advantages and disadvantages of online survey research, online questionnaire authoring software packages, and web survey services," *Journal of Computer-Mediated Communication*, vol. 10, no. 3, pp. 00–00, 2006. DOI: 10.1111/j.1083-6101.2005.tb00259.x.
- [160] A. B. Nylen, M. L. Reyes, C. A. Roe, and D. V. McGehee, "Impacts on driver perceptions in initial exposure to adas technologies," *Transportation Research Record: Journal of the Transportation Research Board*, vol. 2673, no. 10, pp. 354–360, 2019. DOI: 10.1177/0361198119847975.
- [161] A. D. McDonald, H. Alambeigi, J. Engström, et al., "Toward computational simulations of behavior during automated driving takeovers: A review of the empirical and modeling literatures," Human Factors: The Journal of the Human Factors and Ergonomics Society, vol. 61, no. 4, pp. 642–688, 2019. DOI: 10.1177/0018720819829572.
- [162] R. Rossi, M. Gastaldi, F. Biondi, F. Orsini, G. De Cet, and C. Mulatti, "A driving simulator study exploring the effect of different mental models on adas system effectiveness," in *In: De Paolis, L., Bourdot, P. (eds) Augmented Reality, Virtual Reality, and Computer Graphics. AVR 2020. Lecture Notes in Computer Science()*, vol. 12242, Springer, 2020, pp. 71–87. DOI: 10.1007/978-3-030-58465-8_7.
- [163] J. Grabbe, "Implications of experimental versus quasi-experimental designs," in *The Palgrave Handbook of Research Design in Business and Management*, K. Strang, Ed., Palgrave Macmillan, New York, 2015. DOI: 10.1057/9781137484956_10.
- [164] R. McCarney, J. Warner, S. Iliffe, R. van Haselen, M. Griffin, and P. Fisher, "The hawthorne effect: A randomised, controlled trial," *BMC Medical Research Method*ology, vol. 7, no. 1, 2007. DOI: 10.1186/1471-2288-7-30.
- [165] A. Clement and P. Van den Besselaar, "A retrospective look at pd projects," Communications of the ACM, vol. 36, no. 6, pp. 29–37, 1993. DOI: 10.1145/153571.163264.
- [166] A. Cornwall and R. Jewkes, "What is participatory research?" Social Science & Medicine, vol. 41, no. 12, pp. 1667–1676, 1995. DOI: 10.1016/0277-9536(95)00127-s.
- [167] O. W. Bertelsen, "Design artefacts: Towards a design-oriented epistemology," Scandinavian Journal of Information Systems, vol. 12, no. 1, p. 2, 2000.
- [168] M. Cargo and S. L. Mercer, "The value and challenges of participatory research: Strengthening its practice," Annual Review of Public Health, vol. 29, no. 1, pp. 325– 350, 2008. DOI: 10.1146/annurev.publhealth.29.091307.083824.
- [169] J. Jagosh, A. C. Macaulay, P. Pluye, et al., "Uncovering the benefits of participatory research: Implications of a realist review for health research and practice," Milbank Quarterly, vol. 90, no. 2, pp. 311–346, 2012. DOI: 10.1111/j.1468-0009.2012. 00665.x.
- [170] S. Kujala, "User involvement: A review of the benefits and challenges," Behaviour & Information Technology, vol. 22, no. 1, pp. 1–16, 2003. DOI: 10.1080/01449290301782.

- [171] K. Halskov and N. B. Hansen, "The diversity of participatory design research practice at pdc 2002–2012," *International Journal of Human-Computer Studies*, vol. 74, pp. 81–92, 2015. DOI: 10.1016/j.ijhcs.2014.09.003.
- [172] S. Bødker and O. S. Iversen, "Staging a professional participatory design practice: Moving pd beyond the initial fascination of user involvement," in *Proceedings of the* second Nordic conference on Human-computer interaction, 2002, pp. 11–18.
- [173] E. B.-N. Sanders, "From user-centered to participatory design approaches," in *Design and the social sciences*, J. Frascara, Ed., CRC Press, 2002, pp. 18–25.
- [174] J.-Y. Mao, K. Vredenburg, P. W. Smith, and T. Carey, "The state of user-centered design practice - ucd is gaining industry acceptance - but its current practice needs fine-tuning," *Communications of the ACM*, vol. 48, no. 3, pp. 105–109, 2005. DOI: 10.1145/1047671.1047677.
- [175] L. M. Vaughn and F. Jacquez, "Participatory research methods choice points in the research process," *Journal of Participatory Research Methods*, vol. 1, no. 1, 2020. DOI: 10.35844/001c.13244.
- [176] S. Dawadi, S. Shrestha, and R. A. Giri, "Mixed-methods research: A discussion on its types, challenges, and criticisms," *Journal of Practical Studies in Education*, vol. 2, no. 2, pp. 25–36, 2021. DOI: 10.46809/jpse.v2i2.20.
- [177] L. Buckley, S. A. Kaye, and A. K. Pradhan, "Psychosocial factors associated with intended use of automated vehicles: A simulated driving study," *Accident Analysis* & *Prevention*, vol. 115, pp. 202–208, 2018. DOI: 10.1016/j.aap.2018.03.021.
- [178] L. Nilsson, "Behavioural research in an advanced driving simulator experiences of the vti system -," *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, vol. 37, no. 9, pp. 612–616, 1993. DOI: 10.1177/154193129303700921.
- [179] S. L. Jamson and A. H. Jamson, "The validity of a low-cost simulator for the assessment of the effects of in-vehicle information systems," *Safety Science*, vol. 48, no. 10, pp. 1477–1483, 2010. DOI: 10.1016/j.ssci.2010.07.008.
- [180] G. J. Blaauw, "Driving experience and task demands in simulator and instrumented car: A validation study," *Human Factors: The Journal of the Human Factors and Ergonomics Society*, vol. 24, no. 4, pp. 473–486, 1982.
- [181] S. T. Godley, T. J. Triggs, and B. N. Fildes, "Driving simulator validation for speed research," Accident Analysis & Prevention, vol. 34, no. 5, pp. 589–600, 2002. DOI: 10.1016/s0001-4575(01)00056-2.
- [182] D. L. Fisher, M. Rizzo, J. Caird, and J. D. Lee, Handbook of Driving Simulation for Engineering, Medicine, and Psychology. CRC Press, 2011.
- [183] A. F. Larsson, "Automation and the nature of driving: The effects of adaptive cruise control on drivers' tactical driving decisions," Ph.D. dissertation, Lund University, 2013.